ΑΡΙΣΤΟΤΕΛΕΙΟ ΠΑΝΕΠΙΣΤΗΜΙΟ ΘΕΣΣΑΛΟΝΙΚΗΣ ΣΧΟΛΗ ΘΕΤΙΚΩΝ ΕΠΙΣΤΗΜΩΝ ΤΜΗΜΑ ΦΥΣΙΚΗΣ ΕΡΓΑΣΤΗΡΙΟ ΦΥΣΙΚΗΣ ΤΗΣ ΑΤΜΟΣΦΑΙΡΑΣ

Συνδυαστικές επιπτώσεις της ατμοσφαιρικής ρύπανσης και του θερμικού στρες στο περιβάλλον

Διδακτορική Διατριβή της

Δάφνης Π. Παρλιάρη

B.Sc. Φυσικής M.Sc. Φυσικής Περιβάλλοντος

> Επιβλέπων Καθηγητής: Δημήτριος Μελάς

> > Θεσσαλονίκη, 2023

Η υλοποίηση της διδακτορικής διατριβής συγχρηματοδοτήθηκε από την Ελλάδα και την Ευρωπαϊκή Ένωση (Ευρωπαϊκό Κοινωνικό Ταμείο) μέσω του Επιχειρησιακού Προγράμματος «Ανάπτυξη Ανθρώπινου Δυναμικού, Εκπαίδευση και Δια Βίου Μάθηση», 2014-2020, στο πλαίσιο της Πράξης «Ενίσχυση του ανθρώπινου δυναμικού μέσω της υλοποίησης διδακτορικής έρευνας- Υποδράση 2: Πρόγραμμα χορήγησης υποτροφιών ΙΚΥ σε υποψηφίους διδάκτορες των ΑΕΙ της Ελλάδας»







Η άδεια χρήσης του ψηφιακού αρχείου παραχωρείται στο Εθνικό Κέντρο Τεκμηρίωσης και Ηλεκτρονικού Περιεχομένου

Combined effects of atmospheric pollution and thermal stress on the environment

A dissertation by

Daphne Parliari

B.Sc. Physics M.Sc. Environmental Physics

submitted in partial fulfillment of the requirements for the Degree of

DOCTOR OF PHILOSOPHY

in the

Aristotle University of Thessaloniki

Faculty of Sciences

School of Physics

Thessaloniki, 2023

Τριμελής Συμβουλευτική Επιτροπή

Δημήτριος Μελάς	Καθηγητής, Τμήμα Φυσικής, Αριστοτέλειο Πανεπιστήμιο Θεσσαλονίκης
Δημήτριος Μπαλής	Καθηγητής, Τμήμα Φυσικής, Αριστοτέλειο Πανεπιστήμιο Θεσσαλονίκης
Πρόδρομος Ζάνης	Καθηγητής, Τμήμα Γεωλογίας, Αριστοτέλειο Πανεπιστήμιο Θεσσαλονίκης

Λοιπά Μέλη Επταμελούς Επιτροπής (με αλφαβητική σειρά)

Θεόδωρος Γιάνναρος	Εντεταλμένος Ερευνητής, Εθνικό Αστεροσκοπείο Αθηνών
Μιχάλης Λαζαρίδης	Καθηγητής, Σχολή Χημικών Μηχανικών και Μηχανικών Περιβάλλοντος, Πολυτεχνείο Κρήτης
Χαρίκλεια Μελέτη	Επίκουρη Καθηγήτρια, Τμήμα Φυσικής, Αριστοτέλειο Πανεπιστήμιο Θεσσαλονίκης
Αναστασία Πούπκου	Εντεταλμένη Ερευνήτρια, Κέντρον Ερεύνης Φυσικής της Ατμοσφαίρας και Κλιματολογίας, Ακαδημία Αθηνών

Acknowledgment

The work presented here was made possible with the valuable support of my people. First and foremost, I would like to express my utmost appreciation to my supervisor, Professor Dimitris Melas. I am fortunate to have his guidance since my bachelor studies and I am thankful for his unwavering guidance, mentorship, and support throughout the entirety of this long journey. Beyond the academic realm, Professor Melas constantly expresses his genuine interest in my personal and professional development, and for that I am grateful. I am also indebted to the members of my supervising committee, Professors Dimitris Balis and Prodromos Zanis.

My sincerest thanks to Dr. Christos Giannaros and Dr. Theo Economou for helping in various technical and scientific aspects, for reading and commenting this work. Guys, I owe you big time.

My colleague and friend Mrs. Sophia Papadogiannaki was a source of courage and support in times I lacked them. I commit to be there for you when the time comes.

My heartfelt gratitude goes to my parents, Anastasia and Paris, and my brother Dimitris. You are the reason I can grow, and make mistakes, and come back when I need a hug or advice (or a reality check).

Last, I will be forever grateful to Dr. Theodore Giannaros. Words cannot express the ways you have impacted my life. Thank you for putting up with me, for guiding, supporting, and offering "constructive criticism" when necessary. You believed in me when I still hadn't a clue. I hope we keep going for the years to come.

Thank you all for making this possible.

This work could not be accomplished without the data provided by Hellenic Statistical Authority and by the Region of Central Macedonia.

"Με τον καιρό να 'ναι κόντρα,

έχει τιμή σαν πετάς.

Κι ας μένεις μόνος."

Ο ακροβάτης - Χαΐνηδες

Abstract

The scope of the present Doctoral Dissertation is to present a thorough assessment of the compound effects of thermal stress and air pollution on human mortality in the greater area of Thessaloniki, Greece. To meet this goal, daily concentrations of air pollutants (PM10, NO₂ and O₃) and daily values of thermal index (maximum Apparent Temperature, Tapp_{max}) during 2006-2016, were utilized.

The work presented in the following chapters was realized in discrete steps:

At a first stage, a Distributed Lag Non-Linear Model (DLNM) was developed for the association of Tapp_{max} and mortality. The obtained results revealed that heatattributable mortality in Thessaloniki was mainly associated with high temperatures and confirmed that heat has prompt impact on health. As expected, the Elderly (65+) showed increased vulnerability, compared to general population.

At a second stage, the same methodology was followed to examine the effects of short-term exposure to PM10 and O_3 on cause-specific mortality. It was showed that citizens of the city, especially older individuals, were at risk from the current levels of air pollution. O_3 demonstrated more severe impact than PM10, particularly with regards to cardiorespiratory mortality. A secondary aim of this study was to quantify health benefits from two air pollution abatement scenarios that resulted in immediate decreases in mortality risk.

The final objective was to analyze the interactive effects between daily maximum Apparent Temperature and air pollution (NO₂, O₃, PM10) in cause-specific mortality (non-accidental, cardiovascular, respiratory), by developing a Distributed Lag Nonlinear Model using the framework of Generalized Additive Models (GAM). Separate analysis was conducted for the Elderly citizens, as well as evaluation of the lag structure. This study found that deteriorated air quality rendered the local population more susceptible to the effects of temperature variability, and vice versa. More specifically, it was revealed that air pollution enhanced temperature effects on daily mortality, and associations between elevated temperatures and casualties were generally stronger at increased pollution levels. Low temperatures were associated with increased mortality as well, but not to the same extent. Impacts of heat were immediate, whereas the prevalence of cold effects became more pronounced as longer time lags were considered. Based on this analysis, the elderly citizens were susceptible to the adverse combination of thermal conditions and poor air quality.

This dissertation presents for the first time a General Additive Model-based approach to assess the synergistic impact of thermal conditions and air quality for the urban area of Thessaloniki, by using an advanced statistical approach that captures the complex non-linear and lagged dependencies in both the exposure-response and lagresponse associations, and flexible enough to be modified for other synergies. The results presented here hold significant significance for the advancement of public health interventions designed to manage and prevent the health outcomes associated with exposure to unfavorable thermal and air quality conditions.

Περίληψη

Σκοπός της παρούσας Διδακτορικής Διατριβής είναι η παρουσίαση της διεξοδικής αξιολόγησης των συνδυαστικών επιπτώσεων της θερμικής καταπόνησης και της ατμοσφαιρικής ρύπανσης στην ανθρώπινη θνησιμότητα, στην ευρύτερη περιοχή της Θεσσαλονίκης. Για την επύτευξη αυτού του στόχου, χρησιμοποιήθηκαν ημερήσιες συγκεντρώσεις ατμοσφαιρικών ρύπων (PM10, NO2 και O3) και ημερήσιες τιμές θερμικού δείκτη, κατά την περίοδο 2006-2016.

Η έρευνα πραγματοποιήθηκε σε διακριτά βήματα:

Σε πρώτο στάδιο, αναπτύχθηκε ένα μη-γραμμικό μοντέλο κατανεμημένης χρονικής υστέρησης (Distributed Lag Non-Linear Model, DLNM) για τη συσχέτιση του θερμικού στρες και της θνησιμότητας. Τα αποτελέσματα που προέκυψαν αποκάλυψαν ότι η αποδιδόμενη από τη θερμότητα θνησιμότητα στη Θεσσαλονίκη συσχετίστηκε κυρίως με υψηλές θερμοκρασίες και επιβεβαίωσαν ότι η ζέστη έχει άμεσο αντίκτυπο στην υγεία. Όπως ήταν αναμενόμενο, τα ηλικιωμένα άτομα (65+) παρουσίασαν αυξημένη ευαλωτότητα σε σύγκριση με τον γενικό πληθυσμό.

Σε δεύτερο στάδιο, ακολουθήθηκε η ίδια μεθοδολογία για την εξέταση των επιπτώσεων της βραχυπρόθεσμης έκθεσης σε PM10 και O₃ στην ημερήσια θνησιμότητα. Αποδείχθηκε ότι ο πληθυσμός της πόλης, ιδιαίτερα τα άτομα μεγαλύτερης ηλικίας, κινδυνεύουν από τα σημερινά επίπεδα ατμοσφαιρικής ρύπανσης. Το O₃ επέδειξε εντονότερη επίδραση από τα PM10, ιδιαίτερα όσον αφορά τη θνησιμότητα από καρδιοαναπνευστικές αιτίες. Ένας δευτερεύων στόχος αυτής της μελέτης ήταν η ποσοτικοποιήση του όφελους για την υγεία από δύο σενάρια μείωσης της ατμοσφαιρικής ρύπανσης, που είχαν ως αποτέλεσμα την άμεση μείωση του κινδύνου θνησιμότητας.

Τελικός στόχος ήταν να εξεταστούν οι διαδραστικές επιδράσεις μεταξύ του ημερήσιου βιομετεωρολογικού δείκτη και της ατμοσφαιρικής ρύπανσης (NO₂, O₃, PM10) στη θνησιμότητα, αναπτύσσοντας ένα DLNM στο πλαίσιο των Γενικευμένων Προσθετικών Μοντέλων (Generalized Additive Models, GAM). Διαπιστώθηκε ότι η επιδείνωση της ποιότητας του αέρα καθιστά τον τοπικό πληθυσμό πιο ευάλωτο στις επιπτώσεις της θερμοκρασίας, και αντίστροφα. Πιο συγκεκριμένα, η ατμοσφαιρική ρύπανση ενίσχυσε τις επιπτώσεις της ζέστης στην ημερήσια θνησιμότητα και οι συσχετίσεις μεταξύ των αυξημένων θερμοκρασιών και των θανάτων ήταν γενικά ισχυρότερες σε αυξημένα επίπεδα ρύπανσης. Οι χαμηλές θερμοκρασίες συσχετίστηκαν και εκείνες με αυξημένη θνησιμότητα, αλλά όχι στον ίδιο βαθμό. Οι επιπτώσεις της ζέστης ήταν άμεσες, ενώ οι συνέπειες λόγω κρύου απαιτούσαν περισσότερες μέρες για να εκδηλωθούν. Με

βάση αυτή τη μελέτη, οι ηλικιωμένοι πολίτες ήταν ιδιαίτερα επιρρεπείς στο δυσμενή συνδυασμό θερμικής επιβάρυνσης και κακής ποιότητας αέρα.

Η παρούσα διατριβή παρουσιάζει για πρώτη φορά μια προσέγγιση βασισμένη σε Γενικό Προσθετικό Μοντέλο για την αξιολόγηση της συνδυαστικής επίδρασης των θερμικών συνθηκών και της ποιότητας του αέρα στην αστική περιοχή της Θεσσαλονίκης, χρησιμοποιώντας μια προηγμένη στατιστική προσέγγιση που αποτυπώνει τις πολύπλοκες συσχετίσεις, η οποία είναι αρκετά ευέλικτη για να επεκταθεί και σε άλλες μελέτες. Τα αποτελέσματα που παρουσιάζονται εδώ συμβάλουν στην ανάπτυξη μέτρων για την προστασία της δημόσιας υγείας από την έκθεση σε δυσμενείς μετεωρολογικές συνθήκες και συνθήκες ποιότητας του αέρα.

Preface

This dissertation is submitted in partial fulfillment of the requirements for the Degree of Doctor of Philosophy at the Aristotle University of Thessaloniki, Faculty of Sciences, School of Physics, Thessaloniki, Greece. The research presented herein was conducted under the supervision of Professor Dimitrios Melas between January 2018 and September 2023.

Parts of this research have been presented in international scientific journals and proceeding of international peer-reviewed conferences:

Studying the Effects of Urban Heat Island during Hot Days: the Case of Thessaloniki, Greece. Parliari et al (2019). 9th International Symposium on Atmospheric Sciences, October 23-26 2019, Istanbul Turkey.

Short-term effects of temperature and air pollution on human health: the case of Thessaloniki. D. Parliari. (2022). Final Conference of "KASTOM - Innovative System for Air Quality and Forecasting", Thessaloniki, Greece.

Assessment of Heat and Cold Waves Phenomena and Impacts on Environment. Parliari, D., Giannaros, C., Keppas, S. (2022). In: Saxena, P., Shukla, A., Gupta, A.K. (eds) Extremes in Atmospheric Processes and Phenomenon: Assessment, Impacts and Mitigation . Disaster Resilience and Green Growth. Springer, Singapore. <u>https://doi.org/10.1007/978-981-16-7727-4_7</u>

Short-Term Effects of Apparent Temperature on Cause-Specific Mortality in the Urban Area of Thessaloniki, Greece. Parliari, D.; Cheristanidis, S.; Giannaros, C.; Keppas, S.C.; Papadogiannaki, S.; de'Donato, F.; Sarras, C.; Melas, D. Atmosphere 2022, 13, 852. <u>https://doi.org/10.3390/atmos13060852</u>

Short-Term Effects of Air Pollution on Mortality in the Urban Area of Thessaloniki, Greece. Parliari D, Giannaros C, Papadogiannaki S, Melas D. Sustainability. 2023; 15(6):5305. <u>https://doi.org/10.3390/su15065305</u>

Projections of heat-related mortality under the impact of climate change in Thessaloniki, Greece. Parliari D, Keppas S, Papadogiannaki S, Papadopoulos G, Kontos S, Melas D. Environmental Sciences Proceedings. 2023; 26(1):72. https://doi.org/10.3390/environsciproc2023026072

Using statistical approaches to quantify the synergy of heat stress and air pollution on human mortality for a Mediterranean city. Economou T, Parliari, D., and Kushta, J. (2023). EMS Annual Meeting 2023, Bratislava, Slovakia, 4–8 Sep 2023, EMS2023-69, <u>https://doi.org/10.5194/ems2023-69</u>, 2023.

A unifying modelling approach for hierarchical distributed lag models. Economou T., Parliari D., Tobias A., Dawkins L., Stoner O., Steptoe H., Lowe R., Sarran C., Lelieveld J. (2023). [Submitted to Proceedings A].

Synergistic impact of air quality and thermal conditions on human mortality. Parliari D., Economou E., Giannaros C., Kushta J., Melas D., Lelieveld J. (2023). [Submitted to Atmospheric Environment].

Contents

Acknowledgment
Abstract5
Περίληψη7
Preface9
Introduction1
1.1 Problem statement1
1.2 Research objectives4
1.3 Dissertation outline
Health impact of environmental stressors6
2.1 Meteorological conditions
2.1.1 Thermal indices
2.1.1.1 Wet Bulb Globe Temperature (WBGT)9
2.1.1.2 Physiological Equivalent Temperature (PET)9
2.1.1.3 Physiological Equivalent Temperature (DI)9
2.1.1.4 Universal Thermal Climate Index (UTCI)10
2.1.1.5 Apparent Temperature (AT)10
2.2 Air Quality conditions10
2.2.1 PM1012
2.2.2 NO ₂ 13
2.2.3 O ₃
2.2.4 Air Quality indices14
2.2.4.1 AQI system of U.S. Environmental Protection Agency15

2.2.4.2 European Air Quality Index (EAQI)	16
2.2.4.3 Air Quality Health Index (AQHI)	17
2.3 Susceptible population groups	18
2.4 Mortality displacement	19
2.5 Combined effects of heat and air pollution	20
Meteorology and air quality in the area	23
3.1 The Mediterranean region	23
3.2 Greece	24
3.3 Thessaloniki	26
The statistical modeling framework	29
4.1 Distributed Lag Models	29
4.2 Distributed Lag Non-Linear Models	31
4.3 A unifying modelling approach for distributed lag models	32
4.3.1 Interpretation of estimated quantities	33
Short-Term Effects of Apparent Temperature on Cause-Specific Mortality Urban Area of Thessaloniki, Greece	in the
5.1 Data and Methods	38
5.1.1 Study area	38
5.1.2 Meteorological and Mortality Data	39
5.1.3 Data analysis	40
5.2 Results and Discussion	41
5.3 Conclusions	51
Short-Term Effects of Air Pollution on Mortality in the Urban A Thessaloniki, Greece	rea of 53
6.1 Data and Methods	56
6.1.1 Study area	56
6.1.2 Air Quality and Mortality Data	57
6.1.3 Data analysis	57
6.2 Results	59

6.2.1 Mortality Data analysis
6.2.2 Lag Effect analysis60
6.2.3 Total effect analysis62
6.2.4 Suitability of Studied Scenarios in Terms of Health Benefits64
6.3 Discussion65
6.4 Conclusions67
Synergistic impact of air quality and thermal conditions on human mortality68
7.1 Data and Methods70
7.1.1 Study area70
7.1.2 Data description71
7.1.3 Statistical analysis72
7.1.4 Data analysis74
7.2 Results76
7.2.1. Application of GAM for thermal stress76
7.2.2 Synergy between PM10 and Tapp _{max} 77
7.2.2.1 Weight of evidence78
7.2.3 Synergy between NO ₂ and Tapp _{max} 80
7.2.4 Synergy between O ₃ and Tapp _{max} 80
7.2.5 Lag structure analysis81
7.2.6 Heat effect by pollutant levels (all- and specific-cause mortality).83
7.2.7 Attributable Mortality Fraction (all- and specific-cause mortality) for non-optimum Tapp _{max} ranges
7.3. Discussion
7.4. Conclusions90
7.5. Supplementary material91
Conclusions
References
Appendix A: List of Figures

Appendix B: List of Tables	
----------------------------	--

Chapter 1

Introduction

1.1 Problem statement

Rapid urbanization has caused cities to face increasing pressure from growing populations, limited resources and escalating impacts of climate change (World Health Organization, 2017). United Nations estimate that 68% of the world's population will reside in urban areas by 2050 (UN Department of Economic and Social Affairs, 2018) with limited access to nature and increased exposure to environmental hazards, such as air and noise pollution (World Health Organization, 2017). The degradation of air quality has become so prominent in cities that the term urban pollution island (UPI) was recently introduced, to describe the spatial and temporal variations in pollution concentrations between urban and rural areas (Ulpiani, 2021). In modern cities, the synergistic interactions of urban heat islands (UHIs) and UPIs lead to increased pollutant concentrations via various mechanisms, such as accelerated atmospheric chemistry cycles due to high temperatures and increased CO_2 emissions, and ozone-precursors from airconditioning (Ulpiani, 2021).

Regions with poor air quality are home to approximately 90% of the global population (WHO, 2021), making it the most significant environmental risk to human health (European Environmental Agency, 2020). According to World Health Organization (WHO), the most critical air pollutants are sulfur dioxide (SO₂), nitrogen oxides (NO, NO₂), ground level ozone (O₃) and particulate matter (PM), and the exposure to them is linked to adverse health outcomes (WHO, 2021).



Figure 1.1: Health impacts of air pollution (Adapted from (European Environment Agency, 2019))

Exposure to PM₁₀ results in asthma attacks, bronchitis, high blood pressure, heart attack, strokes, immune system reactions, lung irritation, and increased hospitalization due to respiratory and cardiovascular diseases (Larrieu *et al.*, 2007; WHO Regional Office for Europe, 2013; *Health effects of PM10*, 2023). NO₂ is linked to asthma development, inflammation of the airways, reduced lung function and chronic lung disease (*Nitrogen Dioxide / American Lung Association*, no date; D. Jarvis, G. Adamkiewicz, M. Heroux, R. Rapp, 2010; WHO Regional Office for Europe, 2013). Lastly, tropospheric O₃ is associated to human health impacts such as increase in hospital admissions due to cardiovascular and respiratory diseases, shortness of breath, throat irritation, reduced lung function and asthma attacks (WHO Regional Office for Europe, 2013; Yari *et al.*, 2016).

Meanwhile, climate change is one of the most imperative global health threats in the 21st century. Higher average global temperatures are expected to lead to more frequent, persistent, and intense heat waves in the future (Meehl and Tebaldi, 2004) which will undeniably rise climate-sensitive health risks.

As a result, the interactions between ambient temperature and human health have emerged as major issues in the global research community during recent years. Elevated temperatures have profound impact of human health, including increased risks of heat-related illnesses, infectious diseases, malnutrition, and mental health issues (Koppe, C.; Kovats, S.; Jendritzky, G.; Menne, 2004; McGregor *et al.*, 2015). In addition, there is direct link between heat exposure and human mortality from respiratory, cardiovascular and cerebrovascular causes (Brooke Anderson and Bell, 2011; Liu *et al.*, 2011; Gasparrini *et al.*, 2012; Song *et al.*, 2017; World Health Organization, 2022). However, not only extreme temperatures (Basu and

Samet, 2002b; Anderson and Bell, 2009; McGregor *et al.*, 2015), but also changes in moderate temperatures have been shown to present a direct increase in mortality (Analitis *et al.*, 2008).



Figure 2.2: Health impacts of heat (Adapted from (Union of Concerned Scientists, 2019))

While adverse environmental conditions impact the entire population, specific groups such as the elderly, pregnant women and individuals with underlying health conditions are more at risk in terms of health. Evidence shows that senior citizens are particularly susceptible to air pollution due to vasodilation, chronic diseases and reduced heat dissipation (Barnett, 2007; Kenney, Craighead and Alexander, 2014). Research also suggests that individuals with reduced ability for thermo-regulation are the most vulnerable groups of the population to the effect of extreme high temperatures (Basu, Dominici and Samet, 2005; Bunker *et al.*, 2016; Chersich *et al.*, 2020).

The combined effect of air quality and thermal stress, coupled with rapid urbanization, pose a growing health concern for urban populations. These phenomena not only affect the well-being of individuals but also have adverse effects on overall prosperity in metropolitan areas. Therefore, there is a critical need to enhance our understanding of the underlying mechanisms of air pollution and the interaction with elevated temperatures. By doing so, we can effectively address this issue and implement proactive measures to mitigate the impacts of environmental stressors on urban communities.

1.2 Research objectives

The principal goal of this dissertation is to analyze in depth the compound effects of thermal stress and air pollution on human mortality in the greater area of Thessaloniki, Greece.

This study represents the pioneering application of the proposed General Additive Model-based approach to evaluate the health impact of thermal conditions and air quality in Thessaloniki's urban area. This advanced statistical method effectively captures the intricate relationships with optimal use of the data and thus ensures optimal statistical significance, being also adaptable to other synergistic analyses. The present work provides valuable insights to the vulnerable Southern European region, emphasizing the significance of comprehending and mitigating the health burdens associated with ambient air pollution and thermal conditions.

Limitations should be recognized, as well. Generalizing our findings to health impact assessments in other regions with different underlying health conditions and air pollution compositions may be constrained due to the localized nature of similar studies. Additionally, using air quality data from stationary monitoring stations rather than individual exposure data may introduce bias due to potential misclassification of exposure, as individual exposure does not always align precisely with measured atmospheric parameters. The limited availability of air quality and morbidity data also restricted our ability to consider important pollutants such as PM_{2.5} and health aspects such as hospitalization and emergency room visits. Finally, further research is required to establish the causal and etiological mechanisms that explain the results obtained in our study.

On the basis of this general research framework, numerous aspects are investigated:

- ✓ Analysis on mortality causes (all causes, cardiovascular, respiratory) specifically linked with effects on health, caused by heat and air pollution.
- ✓ Examination of the contribution of non-optimum temperature ranges coupled with increasing concentrations of pollutants, to the overall mortality burden.
- ✓ Lag structure analysis across biometeorological index range and pollutants' ranges.
- ✓ Investigation of the behavior of vulnerable groups (Elderly citizens, aged 65+).
- ✓ Interpretation of the produced results considering mortality risk and mortality burden.

1.3 Dissertation outline

The outline of the current dissertation is as follows:

In Chapter 1, a concise overview of the environmental issue tackled in this dissertation is provided, along with a clear delineation of the research goals to be accomplished.

Chapter 2 describes the impact of environmental stressors on health, mainly on thermal and air quality conditions.

In Chapter 3 the area under study is presented, started from the general Mediterranean region and then focusing on Thessaloniki.

Chapter 4 contains the description of the statistical approach that was adopted in this work.

Chapter 5 presents the first stage: evidence on the air pollution-mortality relationship in Thessaloniki urban area from 2006 to 2016, accounting for the cause-specific deaths, lag structure and elderly mortality.

Chapter 6 describes second stage: the effects of thermal stress on daily mortality by examining the associations between thermal indicator and cause-specific mortality, and investigation of the effect of time lag and non-optimum temperatures.

The final stage is presented in Chapter 7: the very first application of an innovative GAM-based approach to assess the synergistic impact of thermal conditions and air quality for the urban area of Thessaloniki, with an advanced statistical approach that captures the complex non-linear and lagged dependencies in both the exposure-response and lag-response associations.

Chapter 8 includes the summary and conclusions of this dissertation.

Chapter 2

Health impact of environmental stressors

2.1 Meteorological conditions

The impact of weather conditions on society has been a major concern for the scientific community in recent decades. In particular, extreme temperatures pose a significant global concern due to their substantial effect on several aspects of life.

A prolonged period of intense heat can profoundly influence society, leading to additional pressure on critical infrastructure, such as power, water, and transportation systems. Sectors like clothing and food retail, tourism, and ecosystem services are not immune to the impact of heat waves, creating potential socioeconomic disparities (McGregor *et al.*, 2015). Elevated temperatures pose several environmental risks, e.g. ecosystem imbalances, habitat loss, water scarcity, disruption of phenological patterns and loss of biodiversity (Guo *et al.*, 2017; Union of Concerned Scientists, 2019; Zinzi and Santamouris, 2019; Galanaki *et al.*, 2023). In certain cases, heat waves can even instigate social disruptions across various levels: hot weather has been linked with higher levels of street violence and attacks, as well as rioting and unrest (Rotton and Cohn, 2000).

The economic effects of high temperatures should not be overlooked. Anthropogenic extreme heat has caused losses of about \$16 trillion to the global economy during 1992-2013 (Callahan and Mankin, 2022). According to the same study, wealthy regions like Europe and North America encountered an average annual decline of 1.5% in GDP per capita due to extreme heat. In contrast, lowerincome regions such as India and Indonesia suffered a more significant reduction, with a yearly GDP per capita loss of 6.7%. European Environment Agency estimates that between 1980 and 2021, weather- and climate-related extremes caused economic losses estimated at EUR 560 billion in the EU Member States, whereas heat waves were responsible for over 13% of the total losses (European Environment Agency (EEA), 2023).

At present, extreme temperatures have become a public health hazard, contributing to a substantial burden of disease. Heat demonstrates profound impact on human health, including increased risks of heat-related illnesses, infectious diseases, malnutrition, and mental health issues (Koppe, C.; Kovats, S.; Jendritzky, G.; Menne, 2004; McGregor *et al.*, 2015). Increased temperature is also associated with preterm birth and abnormal birth weight (Strand, Barnett and Tong, 2011; Carolan-Olah and Frankowska, 2014). Studies have found links between rising temperatures and a range of mental health issues including mental fatigue, aggression and even higher rates of suicide (Schmit *et al.*, 2017; Miles-Novelo and Anderson, 2019). A 1 °C temperature rise was also associated with a significant increase in mood disorders, organic mental disorders, schizophrenia, neurotic, anxiety disorders and hospitalizations for Alzheimer's disease (Xu *et al.*, 2019; Liu *et al.*, 2021).

Thermal conditions significantly impact morbidity rates and the severity of chronic diseases. Heat-related morbidity appears to be associated with a range of pre-existing chronic health conditions, including cardiovascular, cerebrovascular and respiratory disorders (Brooke Anderson and Bell, 2011; Liu *et al.*, 2011; Gasparrini *et al.*, 2012; Bunker *et al.*, 2016; Song *et al.*, 2017; Campbell *et al.*, 2018; World Health Organization, 2022).

Elevated temperatures have a profound influence on mortality rates, encompassing various causes of death. Heat waves are considered to be the deadliest weather disaster in many parts of the world (Brooke Anderson and Bell, 2011). In a research focusing on heat waves in 43 U.S. cities (1987–2005), it was found that mortality increased by 3.74% during heat waves compared with nonheat wave days (Brooke Anderson and Bell, 2011). A systematic review by (Bunker et al., 2016) reports that a 1 °C temperature rise increased cardiovascular, respiratory, and cerebrovascular mortality by 3.44%, 3.60% and 3.18%, respectively. Similarly, a 1 °C increase in mean temperature above 31°C was associated with a 25.18% increase in non-accidental, 34.10% in cardiovascular and 24.27% increase in respiratory mortality, in a study over Wuhan (Zhang et al., 2016). High temperatures were clearly associated with increased risk for those dying from cardiovascular, respiratory, cerebrovascular, and some specific cardiovascular diseases, such as ischemic heart disease, congestive heart failure, and myocardial infarction in the review of epidemiological studies by (Basu, 2009).

Thermal conditions exert a profound influence on human health. Understanding the relationship between meteorology and mortality/morbidity is essential for developing effective public health strategies and interventions aimed at mitigating the adverse effects of extreme temperatures on human well-being.

2.1.1 Thermal indices

Thermal comfort is the subjective sensation that can be defined as the condition of mind which expresses satisfaction with the thermal environment (Epstein and Moran, 2006). Thermal indices serve as tools to quantify the intricate heat exchange between the human body and its thermal environment. These indices aim to assess comfort levels, as well as the stress imposed on the human body. Outdoor environments are influenced by various factors such as air temperature, relative humidity, water vapor pressure, wind velocity, solar radiation, and mean radiant temperature (Deb and Alur, 2010). Furthermore, personal characteristics and behavioral adaptations, including clothing choices, metabolic rate, activity levels, pre-existing health conditions, and exposure duration, significantly impact the individual's thermal sensation (Brake and Bates, 2002).

Simple indices have advantages such as ease of calculation, forecasting, and communication. However, they suffer from significant drawbacks: neglecting essential variables, producing non-comparable results, leading to misrepresentations of the thermal environment, and relying on arbitrary safety limits (Jendritzky, De Dear and Havenith, 2012). Consequently, only indices that incorporate all relevant parameters can be universally utilized across different climatic zones, regions, and seasons (Jendritzky, De Dear and Havenith, 2012).

According to (Epstein and Moran, 2006), there are 4 criteria that a biometeorological index must meet:

- a) Feasibility and accuracy in a wide range of environmental and metabolic conditions.
- b) Consideration of all important factors (environmental, metabolic, clothing etc).
- c) Relevant measurements should reflect the worker's exposure.
- d) Exposure limits should consider physiologic and/or psychological responses reflecting increased risk to safety or health.

Heat stress indices can be categorized into three groups based on their underlying principles (Parsons, 2003): rational indices, empirical indices, and direct indices. Rational indices involve calculations that incorporate the heat balance equation. Empirical indices are based on objective and subjective measures of strain, while direct indices rely on direct measurements of environmental variables. Implementing indices from the first two groups can be challenging in work environments as they involve numerous variables and, in some cases, invasive

measurements. In contrast, the third group of indices is more user-friendly and practical, as it is based on the monitoring of environmental variables.

2.1.1.1 Wet Bulb Globe Temperature (WBGT)

The WBGT index is a widely used direct thermal index, developed in the US Navy as part of a study on heat related injuries during military training. It takes into account the readings from three different temperature sensors: a wet bulb thermometer, a dry bulb thermometer, and a globe thermometer. It is commonly used in occupational settings, sports events, and military operations to assess the risk of heat-related illnesses and to guide appropriate safety measures.

WBGT is described by

WBGT = 0.567 * T + 0.393 * vp + 3.94

where T is air temperature and vp is air vapor pressure (Blazejczyk et al., 2012).

2.1.1.2 Physiological Equivalent Temperature (PET)

PET is a universal index for characterising thermal bioclimate, allowing for the evaluation of thermal conditions in a physiologically significant manner. The index is derived from human energy balance models (Deb and Alur, 2010) and takes into account individual metabolic rates and clothing insulation levels.

PET is defined as "the physiological equivalent temperature at any given place (outdoors or indoors) and is equivalent to the air temperature at which, in a typical indoor setting, the heat balance of the human body is maintained with core and skin temperatures equal to those under the conditions being assessed" (Hoppe, 1999).

By using PET, it is possible to evaluate the thermal comfort or discomfort levels for individuals in a specific environment. The index finds applications in urban planning, outdoor activities, and the assessment of thermal comfort in buildings and it is well suited to the human biometeorological evaluation of the thermal component of different climates (Matzarakis, Mayer and Iziomon, 1999).

2.1.1.3 Physiological Equivalent Temperature (DI)

DI was first introduced by (Thom, 1959) and currently, it is expressed by

DI = 0.5*Tw + 0.5*Ta

where Ta is air temperature and Tw the wet-bulb temperature.

DI is particularly relevant in hot and humid climates where high temperatures combined with high humidity can lead to heat-related discomfort and potential health risks. It helps individuals, especially those who are sensitive to heat or engaged in outdoor activities, to assess the potential impact on their well-being and take appropriate precautions such as staying hydrated, seeking shade, or adjusting clothing choices.

2.1.1.4 Universal Thermal Climate Index (UTCI)

UTCI was conceived as a thermal index covering the whole climate range from heat to cold (Havenith *et al.*, 2012). According to (Jendritzky, De Dear and Havenith, 2012), "UTCI is defined as the isothermal air temperature of the reference condition that would elicit the same dynamic response of the physiological model". It is perceived as one of the most comprehensive indices calculating human thermal stress (Zare *et al.*, 2018).

The formula calculating UTCI index is:

where T is air temperature (°C), Mrt is mean radiant temperature (°C), v is wind speed at 10 m above ground (m/s), and RH is relative humidity (%).

2.1.1.5 Apparent Temperature (AT)

AT is a thermal index that expresses thermal stress and comfort perceived by human, defined as "a measure of relative discomfort due to the combination of heat and high humidity" (Baccini *et al.*, 2008). It is computed from temperature and dew-point data as follows:

$$AT = -2.653 + 0.994 * Ta + 0.0153 * Td^{2}$$

AT is a discomfort index widely met in the literature (de' Donato *et al.*, 2015; Ullah *et al.*, 2021) and it is used operationally as the main thermal index in the Italian National Heat Health Watch Warning Systems (Michelozzi *et al.*, 2010). It is also found that AT is an optimal indicator for predicting all-cause mortality risk and for activating heat alerts and warnings (Lin *et al.*, 2012; Zhang *et al.*, 2014).

2.2 Air Quality conditions

The gradual rise of human activities since the late 18th century, primarily fueled by the burning of fossil fuels and biomass, has led to the deterioration of air quality (Watts, Adger and Agnolucci, 2015; Pozzer *et al.*, 2023). Rapid urbanization rate has caused cities to face increasing pressure from growing populations, limited resources and escalating impacts of climate change (World Health Organization, 2017). UN estimates that 68% of the world's population (2.5 billion people) will reside in urban areas by 2050 (UN Department of Economic and Social Affairs, 2018) with limited access to nature and increased exposure to environmental hazards, such as air and noise pollution (World Health Organization, 2017). The degradation of air quality has become so prominent in cities that the term urban pollution island (UPI) was lately introduced, to describe the spatial and temporal variations in pollution concentrations between urban and rural areas (Ulpiani, 2021). In modern cities, the synergistic interactions of urban heat islands (UHIs) and UPIs lead to increased pollutant concentrations via various mechanisms (Ulpiani, 2021).

Regions with poor air quality are home to approximately 90% of the global population (WHO, 2021), making it the most significant environmental risk to human health and ranked as the second-greatest environmental concern among Europeans (European Environmental Agency, 2020). According to (WHO, 2021), the most critical air pollutants are sulfur dioxide (SO₂), nitrogen oxides (NO, NO₂), ground level ozone (O₃) and particulate matter (PM), and the exposure to them is linked to adverse health outcomes.

Poor air quality, both ambient and indoor, has become a pressing issue, with more frequent and intense episodes of high pollution levels being prevalent in cities across the globe. (Fuller *et al.*, 2022) estimates that air pollution caused 6.7 million deaths in 2019, and this figure is rising. They also state that more than 90% of pollution-related deaths occur in low-income and middle-income countries. Most evidence has been related to natural, cardiovascular and respiratory causes, nervous system causes, stroke, coronary heart disease and chronic obstructive pulmonary disease (Shang *et al.*, 2013; Chen *et al.*, 2019; Gariazzo *et al.*, 2023).

Air pollution not only impacts mortality, but morbidity as well. According to European Environment Agency, exposure can lead to a wide range of diseases, including stroke, chronic obstructive pulmonary diseases, trachea, bronchus and lung cancers, aggravated asthma and lower respiratory infections (European Environment Agency, no date). The most compelling evidence regarding the health consequences of air pollution relates to cardiovascular and respiratory ailments (European Environment Agency, 2019); nevertheless, studies exploring other health impacts are also increasing (WHO, 2022; WHO Regional Office for Europe, 2022).

Older adults are more susceptible to the negative health impacts of air pollution due to their decreased ability to adapt to stressors on their physiological, metabolic, and compensatory processes, as well as their higher likelihood of having cardiovascular and respiratory diseases (Geller and Zenick, 2005; Shumake *et al.*, 2013). Elderly mortality has been found to be particularly affected by PM10 and O_3 , with higher excess risks than other age groups (Cakmak, Dales

and Vidal, 2007; Katsouyanni et al., 2009; Liu et al., 2022; Olstrup, Åström and Orru, 2022).

In literature there is evidence that temperature exacerbates air pollution-related health effects. During the summer season, (Shi *et al.*, 2019) observed that a 1°C increase in temperature corresponds to a rise of 1.05 μ g/m³ in PM2.5 levels and a 1.02% increase in mortality associated with it. In (K. Chen *et al.*, 2018), an increase of 10 μ g/m³ in PM10 was associated with 0.03% and 0.93% increase in total natural mortality at low and high temperatures, respectively. (Jhun *et al.*, 2014) report that high temperatures exacerbate physiological responses to short-term ozone exposure, increasing mortality by 1.35% at a 10-ppb O₃ increase.

It is worth noticing that on days with high air pollution, both heat- and coldrelated mortality increases (K. Chen *et al.*, 2018). Although (Stafoggia *et al.*, 2008) found higher PM10 effects on mortality during warmer days (2.54% increase on death rate), there was also an increase (0.2%) in winter, as well. Similarly, a $10\mu g/m^3$ increase in NO₂ was associated with larger raise in nonaccidental mortality in summer (2.65%) than winter (0.01%). By contrast, in (Qian *et al.*, 2010), the strongest effects of a $10-\mu g/m^3$ increase in PM10 daily concentration on mortality were reported for winter (0.69%) than summer (0.45%).

Pollution prevention, despite its significant impacts on health, societies, and economies, has received inadequate attention and funding within the international development agenda. International organizations and national governments need to continue expanding the focus on pollution as one of the most imperative global environmental issues.

2.2.1 PM10

Particulate matter is reported to pose significant risk in health (Wang *et al.*, 2009). In 2020, the European population experienced approximately 238,000 premature deaths due to PM10 concentrations exceeding the World Health Organization's 2021 guideline of 45 μ gr/m³ (EEA, 2022a).

Exposure to PM10 results to asthma attacks, bronchitis, high blood pressure, heart attack, strokes, immune system reactions, lung irritation, and increased hospitalization due to respiratory and cardiovascular diseases (Larrieu *et al.*, 2007; WHO Regional Office for Europe, 2013; *Health effects of PM10*, 2023).

High levels of PM cause an increase in all-cause, cardiovascular, respiratory and cerebrovascular mortality rates, as well. (Khaniabadi *et al.*, 2017) reports that a 10 μ g/m³ change in PM10 generates 6.6% excess of relative risk in cardiovascular mortality; similar results are obtained by (Christina Adebayo-Ojo *et al.*, 2006), where increase in PM10 concentrations are associated with an increased risk of

2.4% in CVD causes. In the systematic review and meta-analysis by (Orellano *et al.*, 2020), a significant body of evidence shows that an increase in outdoor concentrations of PM10, increases the risk of all-cause and cardiovascular, respiratory, and cerebrovascular mortality in humans. PM10 demonstrated independent effect on mortality in (Olstrup *et al.*, 2019), as an interquartile range increase in pollutant concentration was associated with a 0.8% increase in daily mortality. (Revich and Shaposhnikov, 2010) linked 10 μ g/m³ increases in daily average measures of PM₁₀ with a 0.33% increase in all-cause non-accidental mortality, a 0.66% increase in mortality from ischemic heart disease, and a 0.48% increase in mortality from cerebrovascular diseases.

2.2.2 NO₂

NO₂ is a harmful air pollutant that has detrimental impacts on human health. It is linked to asthma development, increased inflammation of the airways, reduced lung function and chronic lung disease (*Nitrogen Dioxide / American Lung Association*, no date; D. Jarvis, G. Adamkiewicz, M. Heroux, R. Rapp, 2010; WHO Regional Office for Europe, 2013).

As concentration increases, health effects aggravate, too. In 2018, 1.65 million deaths were attributed to ambient NO₂ exposure in China (Zhang *et al.*, 2022). Positive associations to all-cause mortality (0.72%) are reported for 10 μ g/m³ increase in NO₂ in (Orellano *et al.*, 2020). (Christina Adebayo-Ojo *et al.*, 2006) linked 10.7 μ g/m³ increase of NO₂ to 2.2% rise in CVD and 4.5% in respiratory mortality; similar results (1.2% excess of RR in cardiovascular mortality) are demonstrated in (Khaniabadi *et al.*, 2017). Short-term exposure to this pollutant is associated to a 0.75% increase of non-accidental mortality for all age-groups (Corso *et al.*, 2020). Per 10 mg/m³ increase in annual NO₂ exposure was associated with an hazard ratio of 1.127 for all-cause mortality in (Zhang *et al.*, 2022).

According to (Song *et al.*, 2023), more than half a million cases of deaths attributed to NO_2 exposure in urban areas could be prevented worldwide in 2019, if compliance with the latest WHO guideline.

2.2.3 O₃

Tropospheric O_3 is the second most critical pollutant of the atmosphere, playing an important role in climate change and deterioration of air quality (Khaniabadi *et al.*, 2017). It is associated to human health impacts such as increase in hospital admissions due to cardiovascular and respiratory diseases, shortness of breath, throat irritation, reduced lung function and asthma attacks (WHO Regional Office for Europe, 2013; Yari *et al.*, 2016).

Exposure to ozone accounted for 365,000 deaths in 2019 (Health Impacts of Ozone / State of Global Air, no date). O₃ emerges as a risk factor to CVD

mortality related to 2% risk increase, based on (Khaniabadi *et al.*, 2017); the respective rise in (Christina Adebayo-Ojo *et al.*, 2006) is equal to 1.4%. (Orellano *et al.*, 2020) reports strong associations between O3 and all-cause mortality (0.43% increase). O₃ was associated with an increase of 2% in daily mortality two days after the exposure and 1.9% after 3 days (Olstrup *et al.*, 2019). All-cause mortality risk found to be increased by 1.09%; mortality from ischemic heart disease increased by 1.61% and 1.28% mortality increase from cerebrovascular diseases was associated to 10 μ g/m³ O₃ rise, according to (Revich and Shaposhnikov, 2010).

2.2.4 Air Quality indices

Clean air is an essential prerequisite for human health and overall well-being. While air pollution data can be complex and challenging to comprehend, environmental synthetic indices offer a valuable means to condense intricate situations into a single numerical value, enabling comparisons across different time periods and geographical locations (Plaia and Ruggieri, 2011). According to (Shooter and Brimblecombe, 2009), air quality indices (AQIs) serve the purpose of expressing the concentration of individual pollutants on a standardized scale, where the occurrence of effects, particularly health-related effects, is shared among all pollutants. Put simply, air quality indices provide a straightforward and easily understandable method to gauge the quality of air in relation to its potential impact on human health.

(Kanchan, Gorai and Goyal, 2015) set specific criteria that an AQI must fulfill:

- 1. be readily understandable by the public;
- 2. include the major criteria pollutants and their synergisms;
- 3. be expandable for other pollutants and averaging times;
- 4. be related to National Ambient Air Quality standards used in individual provinces;
- 5. avoid "eclipsing" (eclipsing occurs when an air pollution index does not indicate poor air quality despite the fact that concentrations of one or more air pollutants may have reached unacceptably high values);
- 6. avoid "ambiguity" (ambiguity occurs when an air pollution index gives falls alarm despite the fact that concentrations of all the pollutants are within the permissible limit except one);
- 7. be usable as an alert system;
- 8. be based on valid air quality data obtained from monitoring stations that are situated so as to represent the general air quality in the community.

AQIs can serve various aims and purposes. They are utilized to effectively communicate air quality information to the public, evaluate the effectiveness of

pollution reduction strategies, monitor medium and long-term trends, and more. While the fundamental concepts behind AQIs remain similar, their practical implementation can vary significantly (Plaia and Ruggieri, 2011). In general, there is not only one system of air pollution indicators, as several countries have created their own sets of indicators tailored to the needs of each country or region, setting national limits for air pollution.

2.2.4.1 AQI system of U.S. Environmental Protection Agency

U.S. EPA's AQI is defined with respect to the five main common pollutants: carbon monoxide (CO), nitrogen dioxide (NO₂), ozone (O₃), particulate matter (PM10 and PM2.5) and sulphur dioxide (SO₂). AQI shows daily how clean or polluted the ambient air is, highlighting the corresponding effects on public health. The index ranges from 0 to 500. The higher its value, the higher the level of air pollution and the greater is the concern for the health of the citizens (United States Environmental Protection Agency, 2014).

AQI is a piecewise linear function of the pollutant concentration calculated as follows:

$$I = \frac{I_{high} - I_{low}}{C_{high} - C_{low}} * (C - C_{low}) + I_{low}$$

where I is the Index for pollutant, C is the rounded concentration of pollutant, C_{high} is the break point that is greater than or equal to C, C_{low} is the breakpoint that is less than or equal to C, I_{high} is the AQI value corresponding to C_{high} and I_{low} is the AQI value corresponding to C_{low} . The highest individual pollutant index, I, represents the Air Quality Index of the location.

Air Quality Index (AQI) Values	Levels of Health Concern	Colors
When the AQI is in this range:	air quality conditions are:	as symbolized by this color:
0 to 50	Good	Green
51 to 100	Moderate	Yellow
101 to 150	Unhealthy for Sensitive Groups	Orange
151 to 200	Unhealthy	Red
201 to 300	Very Unhealthy	Purple
301 to 500	Hazardous	Maroon

Figure 2.1: AQI scale and health warnings (Adapted from (United States Environmental Protection Agency, 2014))

2.2.4.2 European Air Quality Index (EAQI)

EAQI gives to users the opportunity to gain a better understanding of air quality in their respective living, working, or travel locations. By providing real-time information for Europe, users can access valuable insights into the air quality of specific countries, regions, and cities. This index is built upon concentration values for up to five significant pollutants, namely PM10, PM2.5, O_3 , NO_2 and SO_2 .

EAQI reflects the potential health impact of air quality, focusing on the pollutant with the poorest concentration and its associated health effects. To generate the index, current air quality data reported hourly by member countries of the European Environment Agency are utilized. Additionally, where necessary, the index incorporates modeled air quality data from the European Union's Copernicus Atmosphere Monitoring Service (CAMS).

EAQI is accompanied by health-related messages that provide recommendations for both the general population and sensitive populations. The latter category includes individuals, both adults, and children, who have respiratory problems, as well as adults with heart conditions (European Environment Agency (EEA), 2022).

AQ index	General population	Sensitive populations
Good	The air quality is good. Enjoy your usual outdoor activities.	The air quality is good. Enjoy your usual outdoor activities.
Fair	Enjoy your usual outdoor activities	Enjoy your usual outdoor activities
Moderate	Enjoy your usual outdoor activities	Consider reducing intense outdoor activities, if you experience symptoms.
Poor	Consider reducing intense activities outdoors, if you experience symptoms such as sore eyes, a cough or sore throat	Consider reducing physical activities, particularly outdoors, especially if you experience symptoms.
Very poor	Consider reducing intense activities outdoors, if you experience symptoms such as sore eyes, a cough or sore throat	Reduce physical activities, particularly outdoors, especially if you experience symptoms.
Extremely poor	Reduce physical activities outdoors.	Avoid physical activities outdoors.

Figure 2.2: EAQI health messages (Adapted from (European Environment Agency (EEA), 2022))

2.2.4.3 Air Quality Health Index (AQHI)

In Canada, AQHI was developed as a scale aimed at assessing the effects of air quality on human health, jointly coordinated by the Department of the Government of Canada responsible for National Health Policy and the Department of the Government of Canada responsible for Coordinating Environmental Policies and Programs. This index serves as a tool for safeguarding health and helps individuals make informed decisions to minimize their short-term exposure to air pollution by adjusting their activity levels during periods of heightened pollution. Additionally, AQHI offers guidance on enhancing air quality through recommended behavioral changes that contribute to reducing the environmental impact.

One significant aspect of the index is its focus on individuals who are more susceptible to the impacts of air pollution. It provides specific advice to these sensitive populations on how to safeguard their health during periods of varying air quality, categorized as low, moderate, high, and very high health risks. By tailoring recommendations based on air quality levels, the Air Quality Health Index aims to empower individuals to protect their well-being (Stieb *et al.*, 2008).

AQHI provides a number from 1 to 10+ to indicate the level of health risk associated with local air quality. Occasionally, when the amount of air pollution is abnormally high, the number may exceed 10. The following are the objectives of the AQHI:

- ✓ Create a Canadian communications and planning tool for individuals and caregivers when considering adverse health effects associated with the air pollution mixture.
- ✓ Use "health awareness" as a tool to promote:
 - greater understanding of air quality/health links;
 - physical activity when health risk/air pollutant levels are low; and
 - personal action to reduce air pollution.
- ✓ Create advocates for reducing air pollution.

Health Risk	Air Quality Health Index	Health Messages for At-Risk Population	Health messages for General Population
Low	1-3	Enjoy your usual outdoor activities.	Ideal air quality for outdoor activities
Moderate	4-6	Consider reducing or rescheduling strenuous activities outdoors if you are experiencing symptoms.	No need to modify your usual outdoor activities unless you experience symptoms such as coughing and throat irritation.
High	7-10	Reduce or reschedule strenuous activities outdoors. Children and the elderly should also take it easy.	Consider reducing or rescheduling strenuous activities outdoors if you experience symptoms such as coughing and throat irritation.
Very high	Above 10	Avoid strenuous activities outdoors. Children and the elderly should also avoid outdoor physical exertion.	Reduce or reschedule strenuous activities outdoors, especially if you experience symptoms such as coughing and throat irritation.

Figure 2.3: AQHI scale and health messages (Adapted from (Health Canada, 2010))

2.3 Susceptible population groups

Despite the widespread nature of air quality and thermal stress issues, their impact varies among different groups due to individual physiological sensitivities and varying degrees of exposure influenced by behavioral factors (Makri and Stilianakis, 2008). Extensive evidence demonstrates that the health consequences associated with exposure to air pollution and high temperatures exhibit different degrees of severity based on factors such as life stage (with children and older adults generally being more susceptible), preexisting cardiovascular or respiratory conditions, genetic polymorphisms, and low socioeconomic status (Basu and Samet, 2002a; Sacks *et al.*, 2011). Pregnant women are thought to particularly at risk of heat related effects (Balbus and Malina, 2009), as there is evidence that supports an association between high environmental temperature and preterm birth due to difficulty with thermoregulation and dehydration during heat exposure (Basu, Malig and Ostro, 2010).

Climate change and rapid population ageing are significant public health challenges. Understanding the health risks associated with exposure to adverse conditions on elderly people is vital for preventing related deaths and illnesses in this frail population.

Elderly vulnerability to heat is attributable to physiological and social factors, such as living alone, multiple co-morbidities and high medication use, slow physiological adaptation and behavioral response to thermal stress, vasodilation, limited access to medical care and housing with heating or cooling (Barnett, 2007; Kenney, Craighead and Alexander, 2014; Bunker *et al.*, 2016). In addition, people in the age group 65+ years of age are assumed to be more susceptible to air pollution-induced health effects, due to decreased physiological, metabolic, and compensatory processes, and due to a greater incidence of cardiovascular and respiratory diseases (Geller and Zenick, 2005; Shumake *et al.*, 2013).

(Tsoutsoubi, Ioannou and Flouris, 2021) found high fatality risks due to circulatory causes when older people are exposed to non-neutral environmental

conditions. A 1°C temperature rise increased cardiovascular, respiratory, diabetes mellitus, genitourinary, infectious disease and heat-related morbidity among the elderly (65+ years) in the study by (Bunker *et al.*, 2016). High and very high temperatures increase not only cardiovascular and respiratory mortality of people over 65 years of age by 20% to 35% (Paravantis *et al.*, 2017), but hospital admissions as well (Oudin Åström, Bertil and Joacim, 2011).

On the other hand, the elderly are particularly susceptible to dying from air pollution (Cakmak, Dales and Vidal, 2007). Exposure to PM10 is linked to the development of vascular dementia (Shim, Byun and Lee, 2023) and biomass fuel is associated with 40% risk increase in vision impairment (Zhou *et al.*, 2023). A 10 μ gr/m³ rise of NO₂ concentrations increased the risk of cardiovascular deaths by 0.25% in cold and 3.17% in warm period for people aged > 75 years; the respective values for all-cause mortality were 0.20% and 3.07% (Corso *et al.*, 2020). Last, fine particulate matter is associated to myocardial infarction deaths (Mo *et al.*, 2023) and a variety of air pollutants, such as NO₂, SO₂, O₃ and PM10 results in kidney disease mortality (Cai *et al.*, 2023).

According to (World Health Organization, 2022), public health interventions need to identify and target particularly vulnerable population groups and individuals. This allows for tailored interventions, reduces health disparities, ensures equitable distribution of resources, improves emergency planning, and enhances public awareness and education. These efforts collectively contribute to safeguarding public health, particularly for those who are most susceptible to the impacts of heat and air pollution.

2.4 Mortality displacement

Mortality displacement (or harvesting effect) is identified by a decline in mortality following occurrences of extreme values of environmental stressor under study. The presence of harvesting implies that elevated temperatures and/or pollutant concentrations primarily impact a specific subset of individuals who are already vulnerable to chronic illnesses. These individuals would have succumbed to their conditions within a few days regardless of the weather and air quality conditions.



Figure 2.4: Hypothetical lag structure corresponding to the mortality displacement effect (Adapted from (Zanobetti *et al.*, 2002))

This effect is observed in literature (Schwartz, 2000; Hajat *et al.*, 2005; Baccini *et al.*, 2008; McGregor *et al.*, 2015), and it can be potentially attributed to the socioeconomic and baseline health status of the population and to possible interactions between heat and air pollution (Kouis *et al.*, 2019). The variability in factors influencing individuals' vulnerability to exposure-related mortality may account for the contrasting findings in studies, where some observe a harvesting effect (Baccini *et al.*, 2008), while others do not (D'Ippoliti *et al.*, 2010).

Harvesting effect does not appear in a uniform way among similar studies and there is evidence that patterns of mortality displacement for heat-related deaths depend on the population at risk. More specifically, (Hajat *et al.*, 2005) observed differences in displacement patterns between low-income and high-income countries, with the first affected mainly by infectious diseases of the young individuals, and the latter by chronic diseases of the elderly.

2.5 Combined effects of heat and air pollution

Air pollution is a leading cause of death and disability, contributing to 6.7 million deaths globally in 2019 (*Global Health Impacts of Air Pollution / State of Global Air*, no date). The exposure to pollutants like PM10, O_3 and NO₂ results to direct increase in all-cause, cardiovascular, respiratory and cerebrovascular mortality (Christina Adebayo-Ojo *et al.*, 2006; Khaniabadi *et al.*, 2017; Shi *et al.*, 2019; Meng *et al.*, 2021; Gariazzo *et al.*, 2023). In the meantime, it is established that non-optimal temperatures disrupt the functioning of the body's thermoregulatory system, leading to various physiological responses (Osilla, Marsidi and Sharma, 2023). These responses can contribute to an increased risk of morbidity and

mortality across diverse health conditions, encompassing cardiovascular ailments, respiratory diseases, diabetes, as well as genitourinary and neurological disorders (Psistaki, Dokas and Paschalidou, 2023).

The available evidence concerning the potential interactive effects of heat and ambient air pollution remains inconclusive, by large. Until recently, the vast majority of relative studies were considering the impact of environmental stressors on human health separately (e.g. (Bao et al., 2016; Kriit et al., 2022)). A limited number of epidemiological studies have examined the influence of temperature while considering daily air pollution levels (Stafoggia et al., 2008; Cheng and Kan, 2012; Meng et al., 2012; Jhun et al., 2014) either in the hot (Meng et al., 2012) or the cold tail (Cheng and Kan, 2012) of the temperature distribution, whereas even less have analyzed the impact of air pollutants while adjusting for daily temperature (Turner et al., 2012; Breitner et al., 2014). (Jhun et al., 2014) report that temperature exacerbates ozone impact on mortality, as a 10ppb increase in daily 24-h ozone was associated with a 0.47% increase in mortality. (Meng et al., 2012) connected 10 µg/m³ increment in PM10 with 0.54% increase of total mortality, 0.56% increase of cardiovascular mortality, and 0.80% increase of respiratory mortality, with results being more prominent at high temperatures. Statistically significant interactions between PM10/O₃ and moderate temperatures were reported for all-cause (0.17%/0.66%), cardiovascular (0.23%/0.88%) and respiratory mortality (0.26%/0.79%) (Cheng and Kan, 2012). According to (Stafoggia et al., 2008), seasons strongly affect the relationship between mortality and PM₁₀, and reported 2.54% increase in death risk in summer compared to 0.20% in winter, for a 10-lg/m3 variation in PM10. When confounding with high levels of O_3 and PM10, the associations between high temperatures and mortality were amplified, leading to increases in non-accidental (11.5%, 10.5%) and cardiovascular mortality (9.6%, 9.4%) (Breitner et al., 2014). (Ren and Tong, 2006) revealed that the maximum temperature had a modifying effect on the relationships between PM10 and various health outcomes, including respiratory hospital admissions, emergency visits, cardiovascular emergency visits, and cause-specific mortality at different time lags.

Either confounding the temperature-mortality relationship with pollutants, or the pollution-mortality association with temperature, no uniform conclusions can be drawn. For example, some publications have provided evidence on the synergistic effects (Baccini *et al.*, 2008), others report no combined impact (Samet *et al.*, 1998; Hales *et al.*, 2000; Basu, Feng and Ostro, 2008), while in some instances, effect modifications are reported for only one of the pollutants under study (Breitner *et al.*, 2014).

The diverse perspectives on this matter continue to be a subject of debate within academic literature. Moreover, there has been relatively limited investigation into the simultaneous interactive effects of temperature and air pollution as predictors of mortality.
Studying synergy and not just confounding is a recent development in this research field. (Burkart et al., 2013a) assessed the interactive effects between UTCI and PM10/O₃ on Berlin and Lisbon mortality by developing response surface models. They found strong increases in death numbers at high temperatures and high levels of O₃ for both cities, while Lisbon demonstrated mortality rise during high PM10 pollution (3% mortality change per 1 °C UTCI increase); nevertheless, changes in cold effects with increasing levels of air pollution were not reported. A similar statistical approach was implemented in (Scortichini et al., 2018), in order to study PM10 and O₃ as potential factors influencing the association between temperature and natural mortality during the summer season. The tensor smoother introduced in the analysis showed an overall increase of the heat effects stratified by levels of PM10 and O₃. (Rai et al., 2023) also reported significant modification of heat effect by various pollutants. Summer temperatures increased respiratory mortality by 7.7%, 11.3% and 14.3% at low, medium, and high levels of PM2.5, respectively. Likewise, cardiovascular mortality increased by 1.6%, 5.1% and 8.7% at low, medium and high levels of O₃.

The combined effect of pollution and heat on human health is complex and multifaceted, requiring further investigation to fully comprehend its impacts. It is crucial to conduct comprehensive research in order to develop effective measures that address both pollution and heat, ensuring the protection of human well-being. Taking proactive steps to mitigate these factors is essential for safeguarding public health and fostering a sustainable and resilient environment.

Chapter 3

Meteorology and air quality in the area

3.1 The Mediterranean region

Mediterranean region occupies a unique geographical position in the transition zone between North Africa and central Europe. Minor modifications of the general circulation can lead to substantial changes in the Mediterranean climate, making the area vulnerable to climatic changes (Ulbrich *et al.*, 2006). Moreover, local population is experiencing adverse health impacts from the impaired air quality due to global air pollution crossroads over the Mediterranean (Lelieveld *et al.*, 2002).

For instance, in a research on 10 Mediterranean metropolitan areas a $10-\mu g/m^3$ increase in PM2.5 was associated with increases in all-cause (0.55%) and respiratory mortality (1.91%), and effects were more prominent among those ≥ 75 years of age (Samoli *et al.*, 2013); positive associations between PM10 and several different mortality causes (diabetes, cardiac, cerebrovascular, chronic obstructive pulmonary disease) are reported in (Samoli *et al.*, 2014). The conditions become even more adverse, as increased temperatures in the area are linked to high pollutant production (Pyrgou, Hadjinicolaou and Santamouris, 2018).

Threshold violations in air pollutant concentrations are particularly pronounced in Southern and Eastern Europe (EEA, 2018), especially with respect to particulate matter and ozone concentrations in Greece, Spain, and Italy (Pleijel, 2000; Karanasiou *et al.*, 2014; Sicard *et al.*, 2021). The collective impacts of climate

change and air pollution in these areas should be taken into consideration (Bytnerowicz, Omasa and Paoletti, 2007) under the specific topographical and meteorological conditions of each region (Valjarević *et al.*, 2021). More specifically, the ground-level ozone concentrations in Southern Mediterranean countries are often alarmingly high and are comparable to the highest levels of places that are located in the most contaminated parts of Central Europe (Moussiopoulos *et al.*, 2009).

On the other hand, the Mediterranean area is anticipated to experience more intense, severe, and frequent heat waves under warmer and drier future conditions, making it particularly susceptible to temperature increases (Giannakopoulos *et al.*, 2009; Giannaros, Melas and Giannaros, 2019; Georgoulias *et al.*, 2022). Future climate change projections characterize the Mediterranean as a prominent "hot spot" with the potential for scorching summer conditions to become commonplace by the end of the 21st century (Lelieveld *et al.*, 2012).

Warming rate is estimated between 2°C and 5°C, and precipitation decrease will reach 40%, especially in southern Europe (Lelieveld *et al.*, 2012; Zanis *et al.*, 2015). As a result, more dry days are expected while the northern part of the region will experience the most rapid increase in daytime maximum temperatures (Giorgi and Lionello, 2008; Giannakopoulos *et al.*, 2009; Lelieveld *et al.*, 2012). But both maximum and minimum temperatures will rise, exacerbating thermal discomfort conditions (Diffenbaugh *et al.*, 2007; S. C. Keppas *et al.*, 2021). (Kendrovski *et al.*, 2017) studied the projected impact of heat on population mortality across Europe under RCP8.5 and RCP4.5, and reported that heat impacts will dramatically increase over time in Mediterranean and Eastern European countries. Similar results were demonstrated in (Martínez-Solanas *et al.*, 2021), who found that temperature-attributable mortality in Europe will rapidly start to increase in RCP6.0 by the end of the century and in RCP8.5 already by the middle of the century, especially in the Mediterranean countries.

3.2 Greece

Climate change is perceived as "major threat" for Greece (Watts, Adger and Agnolucci, 2015). Studies estimate that the average temperature is projected to increase between 4.6°C and 2.6 °C by the end of the century (S. C. Keppas *et al.*, 2021; Georgoulias *et al.*, 2022), and the decreased mean precipitation will lead to prolonged dry spell length (Gao, Pal and Giorgi, 2006). According to (Vicedo-Cabrera *et al.*, 2021), the proportion of heat-related mortality attributed to human-induced climate change exceeded 20% during the period 1991–2018; in Athens alone, more than 400 heat-attributable deaths per year are expected for 2030 under RCP8.5 (World Health Organization, 2022).

It is worth noticing that Greece's share of population over 65 is 22.3%, above EU's average (20.6%), whereas the leading causes of death in 2018 were

ischaemic heart disease and stroke (European Observatory on Health Systems and Policies, 2021).

Detrimental air quality also poses a significant risk in Greece. According to (EEA, 2019) in 2019, Greece was found to be in violation of the three most commonly exceeded EU air quality standards for PM10, O₃, and NO₂. This violation has had severe consequences, with an estimated 75 deaths per 100,000 population attributed to air pollution in 2019. The primary causes of these deaths were ischemic heart disease, stroke, and respiratory infections (WHO Regional Office for Europe, 2022), resulting in approximately 1,101 disability-adjusted life years (DALYs) per 100,000 citizens (WHO, no date a) or 104,000 years of life lost (YLLs) for the entire Greek population (WHO, no date b).



Figure 3.1: Premature deaths due to PM2.5, 2019 (Adapted from (OECD/European Union, 2022))

The acute air quality problems are most pronounced in the country's largest cities due to their dense population and the accumulation of air pollutants influenced by topography and adverse meteorological conditions, such as the urban heat island effect (Poupkou *et al.*, 2011). Athens and Thessaloniki, the two major cities in Greece, significantly exceed the EU air quality standards for PM10 concentrations (Kalabokas, Adamopoulos and Viras, 2010). Moreover, the YLLs are primarily affected by exposure to PM10, with O₃ playing a lesser role, as highlighted by (Kassomenos, Dimitriou and Paschalidou, 2013).

It is worth noting that in 2019, air pollution, specifically fine particulate matter and ozone exposure, accounted for 5% of all deaths in Greece, which is 1% higher than the EU average (European Observatory on Health Systems and Policies, 2021). Greece ranks among the countries with the highest absolute impacts regarding O_3 and NO_2 . In 2019, premature deaths attributed to PM2.5, NO_2 , and O_3 reached 10,400, 2,310, and 650 fatalities, respectively (González Ortiz *et al.*, 2021).

Overall, the combined impact of projected elevated temperatures due to climate change and air pollution in Greece presents significant health risks that demand immediate governmental efforts for national adaptation and mitigation.

3.3 Thessaloniki

The urban area of Thessaloniki consists of seven municipalities (Figure 3.2). Accommodating more than 1,000,000 inhabitants (*Population-Housing Census 2021 - ELSTAT*, no date), it is the second largest city in Greece and an important economic and industrial center in the Balkans. Thessaloniki's population has a notable trend of advanced aging with 22.3% of the population aged above 65 years (European Observatory on Health Systems and Policies 2021).

Its Mediterranean climate is significantly affected by the adjacent Thermaikos Gulf in the south; the mean annual temperature of Thessaloniki is ~16 °C and the mean annual relative humidity is 62.4%. The city demonstrates prominent UHI effect with intensity between 1°C and 4°C (Giannaros and Melas, 2012).

The primary causes of air pollution in Greater Thessaloniki Area are road traffic, residential heating (Progiou *et al.*, 2023), biomass burning (Diapouli *et al.*, 2017), and industrial emissions (Moussiopoulos *et al.*, 2009). These activities have led to a decline in air quality (Melas *et al.*, 2017), particularly during the economic crisis (Zyrichidou *et al.*, 2019). In addition, particle pollution in the area is also substantially impacted by dust storms that originate from North Africa (Achilleos *et al.*, 2020; Psistaki *et al.*, 2022; Rizos *et al.*, 2022).



Figure 3.2. The urban area of Thessaloniki. Dashed lines represent municipal borders.

The city experiences exceptionally impaired meteorological and air quality conditions. Despite this fact, current literature focuses almost entirely on Athens, either on heat- (Baccini *et al.*, 2008; Paravantis *et al.*, 2017; Zafeiratou *et al.*, 2019), or pollution-related mortality (Gryparis *et al.*, 2004; Touloumi *et al.*, 2006; Kassomenos, Dimitriou and Paschalidou, 2013). Only recently, a few studies concerning Thessaloniki were published, e.g. (Kouis *et al.*, 2019; Parliari *et al.*, 2022; Psistaki, Dokas and Paschalidou, 2022; Parliari, Giannaros, *et al.*, 2023).

Thessaloniki is one of the most polluted cities in Europe, especially with respect to the PM10 levels (Vlachokostas *et al.*, 2012), but also regarding PM2.5 (Diapouli *et al.*, 2017), O₃, VOCs, and noise pollution levels (Vlachokostas *et al.*, 2013). O₃ limit values are mostly exceeded during the summer months, while winter is the most favorable season for PM10 violations (Moussiopoulos *et al.*, 2009; Kalabokas, Adamopoulos and Viras, 2010). Nevertheless, Thessaloniki's major air quality problem consists of PM10 concentration levels. As a result of consecutive violations of daily PM10 limits established by the EU during 2005-2019, the European Court of Justice recently convicted Greece for the impaired air quality of Thessaloniki (Case C-70/21). Indicatively, 67 PM10 daily exceedances were reported for 2019, instead of 35 permitted by EU rules (Ministry of Environment and Energy, 2020). The increased exposure to PM10 accounted for 530 M€ in 2002, based on the median value-of-a-life-year metric (Vlachokostas *et al.*, 2012).

Exposure to such levels of pollution impacts human health, as expected. (Giannaros et al., 2011) found positive correlation between daily maximum Aggregate Risk Index values and daily hospital admissions for cardiac diseases. Recently, (Parliari, Giannaros, et al., 2023) showed direct increases in all-cause and cardiorespiratory mortality with regard to PM10 (2.3% and 2%) and O₃ (3.9% and 5.3%), which were more prominent between day 0 and 3. Similarly, a 10 µg/m3 increase in PM2.5 resulted in 1.1% rise in cardiovascular mortality, published by (Psistaki et al., 2022). The interaction between air quality and thermal stress is exceptionally important for the local population, as Thessaloniki demonstrates strong Urban Heat Island Effect, ranging from 2°C to 4°C and from 1 °C to 3 °C during summer and winter, accordingly (Giannaros and Melas, 2012). City center, with acute air quality conditions, is characterized by high thermal and strong air quality stress which are extremely detrimental to public health (Zoumakis et al., 2011). (Papanastasiou, Melas and Kambezidis, 2015) revealed combined impact of air pollution and discomfort conditions in Thessaloniki, which was more pronounced during HW days. According to this study, most of the citizens suffered discomfort in 76% of the HW days and in the same period the concentrations of PM10, NO₂ and O₃ increased as well.

Thermal stress conditions, especially elevated temperatures, deteriorate living conditions for the local population. According to (Kouis *et al.*, 2019), heat

depicted immediate increase on cardiovascular (4.4%) and respiratory mortality risk (5.9%) of the city, above 33 °C. Although extreme low temperatures demonstrated larger risk (1.61) to public health than extreme high (1.54), the highest mortality impact was estimated for moderately cold temperatures in the study of (Psistaki, Dokas and Paschalidou, 2022). Based on (Parliari *et al.*, 2022), 2.34% of deaths were attributed to heat and 1.34% to cold; percent changes in mortality per 1°C change above and below 22°C showed larger increases for cause-specific mortalities in heat, in contrast to smaller increases in cold.

Under the influence of climate change, there is urgent need for further study of the area. (S. Keppas *et al.*, 2021) showed that urban heat island intensity during heat waves will exceed 6 °C in Thessaloniki until 2100, whereas number of heat wave days is expected to increase 12 times compared to present. The city is expected to witness an escalation in heat-related cardiorespiratory mortality, with the projected surplus of annual heat-related deaths ranging from 2.4 to 433.7 across different scenarios (2080 – 2099) (Kouis *et al.*, 2021). Last, (Parliari, Keppas, *et al.*, 2023) estimated that mortality burden due to heat will reach 10% during 2096-2100, compared to 6% in current climate conditions (2006-2010), under scenario RCP8.5.

To address these interconnected issues, effective and enduring public action health plans must be identified and put into action (Miranda *et al.*, 2015; Silveira *et al.*, 2016). In general, there is a significant lack of studies that examine the suitability of mitigation measures in terms of health benefits for the area of Thessaloniki. Nevertheless, specific policies to combat particulate air pollution were tested in a recent study (Progiou *et al.*, 2023), which resulted in a more than 20% reduction in the PM10 concentrations in Thessaloniki, Greece. Based on this scenario, (Parliari, Giannaros, *et al.*, 2023) showed respective decrease in total excess mortality by 0.4%. As it is evident, it is necessary to assess the health benefits of the abatement measures by quantifying the impact of air pollution and thermal discomfort on human health for the citizens of Thessaloniki.

Chapter 4

The statistical modeling framework

4.1 Distributed Lag Models

The primary objective of a statistical regression model is to establish the connection between a group of predictors and an outcome variable, and to subsequently estimate the associated impact. However, complications arise when the relationship exhibits temporally delayed effects. For example, in biomedical research it is commonly appreciated that an exposure event produces effects lasting well beyond the exposure period, with an increase in risk occurring from few hours to many years later, depending on the physiological processes linking the exposure and the health outcome.

In such cases, the influence of a predictor, referred to as an "exposure event", extends beyond the immediate period and affects the outcome over a certain duration. This phenomenon has been associated with chronic exposures to environmental stressors, drug intake or occupational exposures to carcinogenic substances (Gasparrini, 2014b). Addressing this temporal dependency necessitates the use of more intricate models that can accurately capture the association and describe the temporal structure of the relationship (Gasparrini, 2011a).

The solution to this issue comes as the incorporation of an extra dimension to the exposure-response relationship, describing the temporal dependence between the exposure and outcome in terms of temporal lag. This term represents the time interval between the exposure event and the outcome when evaluating the delay of the effect and introduces the concept of exposure–lag–response associations (Gasparrini, 2014b).

Several techniques have been proposed to accommodate the temporal variation in risk because of protracted exposures (Thomas, 1988; Vacek, 1997; Hauptmann *et al.*, 2000; Sylvestre and Abrahamowicz, 2009). However, the main limitation of these statistical methods was the assumption of a linear exposure–response relationship.

Distributed lag models (DLMs) are statistical models based on a distributed lag function, analogous to the weighting function described in previous studies (Hauptmann *et al.*, 2000), and analyzing the relationship between an independent variable (predictor) and a dependent variable (the outcome) while accounting for the time-lag effects. It is an extremely useful statistical modeling tool, firstly introduced in econometrics (Almon, 1965) and then in epidemiological studies that capture the links between environmental stresses (such as heat or air pollution) and health outcomes (Lubczyńska, Christophi and Lelieveld, 2015; Gu *et al.*, 2020; Kriit *et al.*, 2022).

This methodology allows the effect of a single exposure event to be distributed over a specific period of time, using several parameters to explain the contributions at different lags, thus providing a comprehensive picture of the timecourse of the exposure-response relationship. The inclusion of lagged values allows for the examination of the dynamic relationship between the variables over time.

Assuming a linear exposure-response relationship between the outcome variable M_t (e.g. mortality counts) and some environmental exposure AT_t (e.g., temperature), DLMs can be generically defined as

$$M_t \sim NegBin(\mu_t, \theta) \quad (1)$$
$$log(\mu_t) = \alpha + \beta_0 A T_t + \beta_1 A T_{t-1} + \dots + \beta_L A T_{t-L} \quad (2)$$

where L days is the maximum temporal lag. The use of the Negative Binomial distribution (NegBin) is a conventional choice in epidemiological analyses as it extends the Poisson distribution to allow for over-dispersion (extra variance). The linear "effects" β_l act multiplicatively on the mean mortality count μ_t (and additively on log(μ_t)).

Although simple and effective, the approach assumes that the effect of the exposure at each lag is linear, which is a substantial constraint, particularly with respect to scientific understanding on how temperature relates non-linearly to mortality (Honda *et al.*, 2014). Some attempts to relax this assumption and explore non-linear lagged effects of exposure variables have been proposed (Braga, Zanobetti and Schwartz, 2001; Roberts and Martin, 2007; Muggeo, 2010). The most commonly used development is the one firstly conceived and applied by (Armstrong, 2006), who modeled non-linear and lagged exposure-response relationships as Distributed Lag Non-Linear Models.

4.2 Distributed Lag Non-Linear Models

DLMs are built on the premise of a linear connection between exposure and outcome. Nevertheless, it is crucial to consider the well-established nonlinear relationships, as emphasized by (Braga, Zanobetti and Schwartz, 2001; Gasparrini and Armstrong, 2010).

Distributed Lag Non-linear Models (DLNMs) offer a versatile modeling approach that can effectively capture both the nonlinear and delayed impacts of environmental factors on the outcome. In contrast to conventional models, DLNMs introduce the temporal lag explicitly as a dimension, allowing for the representation of the association between temperature/air pollution and mortality over time. This bi-dimensional space, encompassing both the exposure and the lag, defines the exposure-lag-response relationship, providing a comprehensive characterization of the dependency (Gasparrini, Armstrong and Kenward, 2010; Gasparrini, 2021).

A statistical development of these models is based on the utilisation of basis functions for defining non-linear functions that describe the association of the independent variable (exposure) with the dependent variable (mortality) across different lags. Specifically, two sets of (marginal) basis functions are chosen independently, one for the lag dimension and one for the exposure dimension. These are then combined using the idea of a tensor product, to produce a bi-dimensional cross-basis function (Gasparrini, 2011b, 2014b). The choice of the two sets of basis functions determines the shape of the relationship in each dimension. In this sense, DLMs are a special case of the more general DLNMs, when the exposure-response is assumed linear.

The framework described in Equation (2) can be extended to DLNMs in order for the effect of the exposure at each lag to be non-linear, by extending the formulation of μ_t via:

$$\log(\mu_t) = \alpha + h(0, AT_t) + h(1, AT_{t-1}) + \dots + h(L, AT_{t-L})$$
(3)

where $h(l, AT_{t-l})$ is a two dimensional function of lag and the stressor/s. Constructing this function using regression splines can be achieved using tensor products (also termed a cross-basis function) (Wood, 2017; Economou *et al.*, 2023).

DLNMs have become the most commonly used method in literature to quantify the health effects of various environmental stressors, concerning thermal stress (e.g. (de' Donato *et al.*, 2015; Martínez-Solanas *et al.*, 2018; Parliari *et al.*, 2022; Psistaki, Dokas and Paschalidou, 2023)), or air pollutants (e.g. (Psistaki *et al.*, 2022; Font-Ribera *et al.*, 2023; Gariazzo *et al.*, 2023; Parliari, Giannaros, *et al.*, 2023). In order to assess the combined impact of air quality and heat discomfort on health, some studies have either confound the temperature-mortality relationship with pollutants ((Stafoggia et al., 2008; Cheng and Kan, 2012; Meng

et al., 2012; Jhun et al., 2014)), or the pollution-mortality association with temperature ((Turner et al., 2012; Breitner et al., 2014)). Nevertheless, no uniform conclusions have been drawn. For example, some publications have provided evidence on the synergistic effects (Baccini *et al.*, 2008), others report no combined impact (Samet *et al.*, 1998; Hales *et al.*, 2000; Basu, Feng and Ostro, 2008), while in some instances, effect modifications are reported for only one of the pollutants under study (Breitner *et al.*, 2014).

4.3 A unifying modelling approach for distributed lag models

DLNMs have certain restrictions with regards to studying synergy of a combination of different exposures, and over-fitting (over-explaining) the data. Recently, a modelling approach to fitting DLNMs that encompasses all necessary extensions into a unified framework has been developed and implemented by (Economou *et al.*, 2023). More specifically, they proposed to fit DLNMs using the machinery of penalised Generalised Additive Models (GAMs) as implemented in the R package *mgcv* (Wood, 2011, 2017).

The hierarchical nature of the approach is particularly important when considering interactions of an already complex structure (i.e., the 2D exposure-lag function) with other factors, since hierarchical structures are a natural way of dealing with sparse data and the 'curse of dimensionality'. GAMs from the *mgcv* package have been utilised before in fitting DLNMs across a range of application areas and one of the main contribution of this study was to illustrate further the use of GAMs for flexible analyses involving the effect from lagged exposures.

Assuming that $h(l, AT_{t-l})$ in Equation (3) is smooth, then the model is a GAM which (using *mgcv*) can be used to optimally estimate $h(l, AT_{t-l})$ by penalising its flexibility. Penalisation in non-linear regression models is important in order to avoid over-fitting the data, and to also reduce sensitivity to the choice of L (as long as this is large enough).

More generally, equation (3) above can be written as:

$$\log(\mu_t) = \alpha + \sum_{l=0}^{L} h(l, x_{t-l}) + \log(O_t) \quad (4)$$

where x_t is an environmental exposure (AT_t or whichever stressor desired) and O_t is population count, in case we pool data from different regions. Using *mgcv*, the splines are by default constrained to be centered at zero, so that $h(l, x_{t-l})$ is the additive change in the overall log mean count of deaths.

Additionally,

$$RR(l, x_{t-l}) = \exp\{h(l, x_{t-l})\}$$
 (5)

can be interpreted as the relative risk (RR), the multiplicative change with respect to $\exp\{a\}$ (the mean count across the time period of study).

In the work described in the present thesis, we aimed to study the synergistic effect of air pollution (PM10, O_3 , NO_2) on daily mortality. For this, we define interactions of the AT-lag effect with other exposures (e.g. with PM10), via

$$\log(\mu_t) = \alpha + h(0, AT_t, PM_{10,t}) + h(1, AT_{t-1}, PM_{10,t-1}) + \dots + h(L, AT_{t-L}, PM_{10,t-L}).$$
(6)

Function $h(\cdot)$ can be also defined using tensor product interactions of regression splines (Wood, 2017), a robust and flexible approach to defining smooth functions across many dimensions. The models presented in Chapter 7 were implemented in the R package mgcv, which makes use of penalised maximum likelihood to estimate these smooth functions.

Since we are using mgcv, $RR(l, AT_{t-l}, PM_{10,t-l}) = (\exp \{h(l, AT_{t-l}, PM_{10,t-l})\})$ is still the relative risk compared to the mean mortality count $\exp\{a\}$. Therefore, $RR(\cdot) > 1$ implies greater than average mortality risk, whereas $RR(\cdot) < 1$ means lower-than-average risk.

4.3.1 Interpretation of estimated quantities

The conventional way of illustrating the estimated effects from DLNMs is a plot of RR over a grid of finite values for x_{t-l} and l (Figure 4.1). This is a 3D plot showing how estimated risk varies for values of the exposure across different lags. It is a counterfactual "statement" of keeping x_{t-l} fixed for all 1 (e.g. fixing temperature at 40°C for 21 days) and seeing what the associated contributions are to risk for each of lag.



Figure 4.1. Relative risk estimates for the Thessaloniki data based on a Negative Binomial GAM (Adapted from (Economou *et al.*, 2023). Red colour indicates RR>1 and blue shows RR<1.

A useful summary of the effect in the exposure dimension is the cumulative risk (CR) where the additive effects $h(l, x_{t-l})$ are summed across the lags. For instance,

$$CR(AT) = \exp\left\{\sum_{l=0}^{L} h(l, AT)\right\}$$
(6)

is the cumulative risk (CR) for AT_t from model (3).



Figure 4.2. Cumulative relative risk estimates for the Thessaloniki data (Adapted from (Economou *et al.*, 2023)).

GAMs implemented in *mgcv* can be interpreted in a Bayesian way (Wood, 2017) and Monte Carlo simulation can be used to quantify the associated uncertainty in estimating the smooth function $h(\cdot)$. This in turn enables the assessment of "significance" in the estimated RR, which here is determined by checking whether the value 1 lies within the 95% credible interval of $\rho(\cdot)$. If it is not, then we can say that with probability 0.95 the value of the RR is not 1 and that the weight of evidence supporting this is strong.

In terms of interpretation, RR and CR are counterfactual quantities, since the variability of the exposure is not allowed for. As a result, metrics such as the Attributable Fraction (AF) have been proposed (Steenland and Armstrong, 2006) to interpret the risk estimates in terms of the observed data. If we define R_0 as the mortality risk at the optimum temperature value (or OT, the exposure value form which cumulative risk it at its minimum) as the risk of the "non-exposed" population, and R_1 as the risk due to any other exposure value x, then

$$AF = 1 - \frac{R_0}{R_1}$$
 (7)

is one minus the risk ratio which can be interpreted as the fraction of mortality cases attributable to x_t being different to the OT.

For DLNMs we use CR to define risk so for observed exposure x_t on day t:

$$AF(x_t) = 1 - \frac{\exp\{\sum_{l=0}^{L} h(l, OT)\}}{\exp\{\sum_{l=0}^{L} h(l, x_t)\}} = 1 - \exp\{\sum_{l=0}^{L} [h(l, x_t) - h(l, OT)]\}$$
(8)

The exposure in the sum is x_t rather than x_{t-l} so what is quantified here is how the risk of experiencing x_t "today" is distributed over the next L days. This was termed as the forward AF in (Gasparrini and Leone, 2014) who provide a thorough exposition of attributable risk from DLNMs.

The attributable number (AN) then uses the observed number of cases on day t, y_t to quantify the number of cases attributable to x_t . Specifically, the number of deaths attributed to the exposure being x_t on day t, is:

$$AN(x_t) = AF(x_t) * y_t (9)$$

Chapter 5

Short-Term Effects of Apparent Temperature on Cause-Specific Mortality in the Urban Area of Thessaloniki, Greece

The interactions between ambient temperature and human health have emerged as major issues in the global research community during recent years. Not only extreme temperatures (Basu and Samet, 2002b; Anderson and Bell, 2009; McGregor *et al.*, 2015), but also changes in moderate temperatures have been shown to present a direct increase in mortality (Analitis *et al.*, 2008). Meanwhile, climate change is recognized as one of the most imperative global health threats in the 21st century (S. C. Keppas *et al.*, 2021). Several areas around the globe are at high risk not only in present (Iyakaremye *et al.*, 2021) but in future climate as well (Ullah *et al.*, 2022).

The Mediterranean area is particularly susceptible to temperature increases and it is anticipated to experience more intense, severe and frequent heat waves under warmer and drier future conditions (Giannakopoulos *et al.*, 2009; Giannaros, Melas and Giannaros, 2019; Georgoulias *et al.*, 2022). Concerning Greece, located in southeast Mediterranean, the average temperature under a "business as usual" scenario (Representative Concentration Pathway RCP8.5) is projected to increase by 4.5 °C in summer and 2.6 °C in winter, until 2100 (S. C. Keppas *et al.*, 2021). Taken into consideration poor air quality, comfort conditions become even more unfavorable during heat wave episodes (Papanastasiou, Melas and Kambezidis, 2015).

Global warming is generally predicted to lead to both an increase in heat-related mortality and a decrease in cold-related mortality (Watts, Adger and Agnolucci, 2015). Nevertheless, it is well established that the temperature-mortality relationships demonstrate strong dependencies on local climate characteristics and vary significantly among study areas (Armstrong et al., 2011). Therefore, it is of vital importance to conduct local-based environmental epidemiology studies and thorough investigation on the city level, in order to deliver robust heat-health action plans for the improvement and strengthening of public decision making and health care (Matthies et al., 2008). These action plans should also include meteorological variables such as relative humidity and precipitation, when possible, in order to better formulate adaptation strategies and measures (Mie et al., 2022; Sein et al., 2022). Associating ambient temperatures with health outcomes poses significant challenges. To model the temperature-health dependencies, the dose-response association, which is non-linear and heterogeneous between populations with different demographic characteristics, should be taken into consideration (Gasparrini, Guo and Hashizume, 2015). In addition, the impact of the exposure event may not be immediate, but rather delayed in time by a few days or even weeks (defined lags). Sophisticated statistical approaches are necessary to describe such complex patterns.

Distributed lag non-linear models (DLNMs) is a flexible modeling framework capable of simultaneously representing the nonlinear and delayed effects of temperature on mortality. In addition to the usual exposure-response relationship over the space of the predictor, the lag dimension represents a new space over which the association is defined, by describing a lag-response relationship. The dependency is characterized in the bi-dimensional space of predictor and lag, and it is defined as an exposure-lag-response association (Gasparrini, 2011b). DLNMs have been successfully applied in previous temperature-related epidemiological studies: Nordio et al. (Nordio et al., 2015) investigated 211 US cities from 1962 to 2006, accounting for hot and cold temperatures at different lags, to evaluate change of associations over space and time. An extensive analysis between 1998 and 2012 estimated the location-specific temperature-mortality relationships of 147 regions in 16 European countries (Martínez-Solanas et al., 2021), and de'Donato et al. (de' Donato et al., 2015) evaluated the patterns of changes in the temperature-mortality relationship and the number of deaths attributable to heat in nine European cities before and after summer 2003 (1996–2002 and 2004–2010), also controlling for air pollution.

Greece lies within the Mediterranean climate change hotspot, and according toWHO, more than 400 heat-attributable deaths per year are expected in Athens for 2030 under RCP8.5 (World Health Organization, 2022). Previous studies concerning Athens showed an increase in cardiovascular and respiratory mortality among the elderly (65 years and over) by 20% to 35% respectively (Paravantis *et al.*, 2017), and reported a 5.54% increase in natural mortality above the city-specific threshold (Baccini *et al.*, 2008). Last, a recent study of Zafeiratou et al.

investigated the intra-urban differentiation in the temperature–mortality effects of 42 Municipalities within the Greater Athens Area, showing an increase on allcause, cardiovascular and respiratory mortality for increases in temperatures (Zafeiratou *et al.*, 2019).

As it has been made evident, the literature focuses almost entirely on Athens, with the rest of Greek urban agglomerations being under-represented. The recent study by Kouis et al. (Kouis *et al.*, 2019) is the first significant contribution for the greater area of Thessaloniki, on the association between high ambient air temperature and cardio-respiratory mortality from 1999 to 2012. The study shows significant effects of heat above the temperature threshold of 33 °C. Furthermore, cardiovascular and respiratory mortality risk increased by 4.4% and 5.9% above the threshold, respectively.

For such purposes, we investigate the association between daily maximum apparent temperature and daily all-cause (natural, non-accidental), cardiovascular, cerebrovascular and respiratory mortality and mortality concerning the elderly (all-causes, 65+ years) from 2006 to 2016 in the urban area of Thessaloniki, by describing the exposure-lag-response association with the use of a DLNM. We also examine the effect of time lag and non-optimum temperatures using measures such as relative risk and attributable fractions, as very little is known about the relative contribution of both heat and cold from moderate and extreme temperatures to the whole disease burden.

The present Chapter is based on recent developments in statistical modeling to account for the versatile temperature–mortality dependencies. This single-city analysis demonstrates significant advantages, taking into consideration the size of the examined area, the city population lasting exposure to high temperatures and the increasing proportion of elderly people among urban dwellers. The data and methodology used in this work are described in Sections 5.1 and 5.2 and contain detailed presentation of results, and Section 5.3 summarizes the major conclusions.

5.1 Data and Methods

5.1.1 Study area

The study focuses on the urban area of Thessaloniki, consisting of six Municipalities and one Municipal Unit. Thessaloniki, the second largest city of Greece and one of the largest urban agglomerations in the Balkan Peninsula, is located in the northern part of the country accommodating approximately 1,000,000 inhabitants (Poupkou *et al.*, 2011). As shown in Figure 5.1, it is built along the north-east coast of the Thermaikos Gulf and its Mediterranean climate, with hot and dry summers and mild and wet winters, and is directly affected by

the sea. The annual mean temperature is 15.9 °C, the annual mean relative humidity is 62.4%, the annual mean precipitation is 448.7 mm and the annual mean wind speed is 5.6 m/s (Giannaros and Melas, 2012).



Figure 5.1. The study area. Topography is shaded with linear altitude (above sea level) scales. The locations of Thessaloniki centre and Airport weather station are shown.

5.1.2 Meteorological and Mortality Data

Hourly values of temperature and dew-point temperature for the period 2006–2016 were obtained from the Makedonia Airport weather station (Longitude 22.97, Latitude 40.53, Elevation 2m), which is operated by the Hellenic National Meteorological Service. Previous studies, e.g., (Ullah *et al.*, 2021), have shown that uniformity and stability tests are performed to achieve better results when using in-situ data. Nevertheless, the meteorological dataset used in this study is provided by HNMS, responsible for Greece's national weather forecast and certifier of the national weather network. Therefore, the evaluation performed on the data by the authors indicated no further assessment.

The exposure variable considered in the study is represented by Apparent Temperature, a thermal index that expresses thermal stress and comfort perceived by humans. To identify the most effective thermal predictor of heat-related mortality, we investigated daily mean temperature (Tmean, °C) and daily maximum Apparent Temperature (Tapp, °C). Tapp was computed from temperature and dew-point data as follows:

 $Tapp = -2.653 + 0.994*Ta + 0.0153*Td^{2}$

where Ta = air temperature (°C) and Td= dew point temperature (°C). Then, the daily maximum value was extracted (Tapp_{max}), defined as the highest among hourly values.

Tapp is a discomfort index widely used in the literature (D'Ippoliti *et al.*, 2010; de' Donato *et al.*, 2015) and it is used operationally as the main thermal index in the Italian National Heat Health Watch Warning Systems (Michelozzi *et al.*, 2010). It is also found that Tapp is an optimal indicator for predicting all-cause mortality risk and for activating heat alerts and warnings (Lin *et al.*, 2012; Zhang *et al.*, 2014).

The Hellenic Statistical Authority (ELSTAT) provided the daily mortality data, including age and cause of death for all Municipalities in the urban area of Thessaloniki for the study period (2006–2016). The causes of death were classified to All causes (natural, non-accidental), Cardiovascular, Cerebrovascular and Respiratory, according to ICD-10 (International Classification of Diseases, 10th version). A focus on elderly total mortality was carried out, considering all total deaths among people aged 65 years and older.

5.1.3 Data analysis

When assessing the mortality-temperature relationship in a specific area, the most appropriate temperature-based index needs to be investigated (Analitis *et al.*, 2008). Various studies have been conducted on the matter but were unable to draw a decisive conclusion (Barnett, Tong and Clements, 2010; Kim *et al.*, 2011), whereas in order to adopt an appropriate temperature index, regional meteorological characteristics and the disease status of population should be considered.

In the present study, we investigate the effect of temperature-related variables on mortality for the urban area of Thessaloniki during the period 2006–2016, using Distributed Lag Non-linear Models in order to simultaneously account for the non-linear and lagged effect of temperatures on mortality, as previously presented in the literature (Gasparrini, 2014a, 2014b; Guo *et al.*, 2021). The main advantage of this method is that it provides a detailed representation of the time-course of the exposure–response relationship, which then estimates the overall effect in the presence of delayed contributions. This family of models is implemented in the package DLNM (Gasparrini *et al.*, 2017) within the statistical environment R (https://www.r-project.org/foundation/).

According to the Akaike information criterion (AIC) (Gasparrini, Armstrong and Kenward, 2010), we applied a generalized non-linear model with quasi-Poisson family, with the following choices regarding the control of confounders for the effect of temperature index: (i) natural cubic splines to describe the exposure–response function; (ii) natural cubic splines to describe the lag–response function; (iii) knots were placed at quantiles (10th, 75th, 90th) in the range of temperature

variables and at equal intervals in the logarithmic scale of lags; (iv) maximum lag was set to 20 days; (v) indicator variables were chosen for day of the week and 6 degrees of freedom per year were set to describe long-time trends and seasonality.

To evaluate the best model specification concerning the exposure variable, we compared the constructed models with T_{mean} and $Tapp_{max}$ using:

The Quasi Akaike Information Criterion (QAIC), a modified version of AIC to deal with the over-dispersed Poisson model, which can be used to assess the model fit of the quasi-Poisson regression model. It considers both the statistical fitness of the model and the number of parameters fitted (Peng, Dominici and Louis, 2006).

The partial autocorrelation function (PACF) criterion calculated as the absolute value of the sum of the partial autocorrelations of the residuals from lags 1 to 20 (Touloumi et al., 2006). PACF investigates the remaining autocorrelation.

The best model is chosen when/if both QAIC and PACF are minimized at the same time. These two criteria are widely used in time series analysis to decide for the best fit (Peng and Dominici, 2008; Rodopoulou et al., 2015; Yan et al., 2019).

Table 5.1 reports QAIC and PACF values for the two models. Based on these, Tapp_{max} can be uniformly chosen as the most representative exposure variable for Thessaloniki for our dataset.

Table 5.1. Model fit statistical criteria per exposure variable.				
Temperature Index	QAIC	PACF		
T _{mean}	23,125	3.918		
Tapp _{max}	23,088	3.879		

. . .

5.2 Results and Discussion

During the 11-year study period, we analyzed 73,990 deaths from all natural causes— 21,811 from cardiovascular diseases, 10,007 from cerebrovascular causes and 7134 from respiratory causes. Moreover, total mortality concerning the elderly counted 62,482 deaths. Table 5.2 provides the descriptive statistics on exposure variable Tapp_{max} (bottom panel) and daily mortality (top panel) during the study period. Elderly mortality is distinctively noticeable, as Greece is among the countries with the highest proportion of elderly population in Europe (EUROSTAT, 2019). It is also evident that cardiovascular deaths account for an important part of cause-specific daily mortalities. This result is supported by

(European Observatory on Health Systems and Policies, 2019), which reported that stroke and ischemic heart disease is the leading causes of death in Greece.

Tapp _{max}					
Median	Range	Min	5th perc.	95th perc.	Max
18.7	49.1	-5.6	2.5	35	43.5
		Dai	ly Mortality		
		Mean		St. dev.	
All-cause	ll-cause 18.4		3.4	4.7	
Cardiovascular		5.4		2.4	
Cerebrovascular		2.5		1.6	
Respiratory		1.8		1.4	
Elderly		15.5		44	

Table 5.2. Summary statistics of daily mortality (number of deaths, bottom) and
Tappmax (°C, top).

Figure 5.2 displays the bi-dimensional exposure-lag-response surface of the fitted RR in a three-dimensional diagram for maximum apparent temperature and lag values. The association between $Tapp_{max}$ and mortality is non-linear, suggesting an immediate increase in mortality for exposures to high levels of $Tapp_{max}$ at lag days 1–3, whereas for low levels the effect is delayed (day 10–15).



Figure 5.2. Exposure–lag-response risk surface demonstrating the nonlinear association between temperature and mortality, as calculated for the city of Thessaloniki for the period 2006–2016.

Figure 5.3 top panel shows an overall cumulative exposure–response curve (with 95% CI marked as grey areas), interpreted as the risk cumulated over the entire lag period of 20 days, and Tapp_{max} range as a frequency histogram. Dashed lines define extreme temperatures (5th and 95th percentile, 2.5°C and 35°C respectively) and dotted line represents minimum mortality temperature (MMT) at 22°C.



Figure 5.3. Overall cumulative exposure–response association (top panel) and Tapp_{max} distribution (bottom panel), as calculated for the city of Thessaloniki for the period 2006-2016.

Identifying MMT has important practical implications for the investigation of the relationship between temperature and mortality and the definition of non-optimum temperatures, as mortality rates increase at temperatures outside local MMT (Yin *et al.*, 2019). Studies have shown that MMT varies from city to city, according to local climate and latitude (Tobías, Armstrong and Gasparrini, 2017). The values in the present work were anticipated for a Mediterranean city (Kouis *et al.*, 2019; Yin *et al.*, 2019), although for respiratory mortality, MMT was found to be lower. Generally, MMT levels are higher for Mediterranean cities than those in Northern Europe, indicating that residents of south-coastal cities are acclimatized to heat and risk conditions (Baccini *et al.*, 2008).

Risk increases rather slowly for cold temperatures below MMT and shows a more prominent exponential rise below the 5th percentile of $Tapp_{max}$, although it is not statistically significant for Thessaloniki. Similarly, hot temperatures between MMT and 95th percentile depict a slow, insignificant increase in RR, with a significant effect only for extreme values above 35 °C, reaching the highest RR values at 43 °C.

Table 5.3 summarizes the effect of heat and cold on cause-specific mortality and mortality of the elderly. Results show statistically significant effects of high temperatures on total and cause-specific mortality with a greater risk for respiratory deaths (RR 1.47, CI 95% 0.9–2.4), as shown in previous studies (Breitner *et al.*, 2014). In Thessaloniki, exposure to cold (Tapp_{max} between 5th percentile and MMT) did not show a significant effect on mortality.

Cause of Death	MMT (°C)	Relative Risk (95% CI)	Relative Risk (95% CI)	
		Extreme Low, 2.5 °C	Extreme High, 35 °C	
All-cause	22	1.05 (0.95–1.16)	1.17 (1.04–1.32)	
Cardiovascular	30	1.18 (0.93–1.52)	1.20 (1.08–1.31)	
Cerebrovascular	30	1.41 (0.97–2.04)	1.08 (0.93–1.24)	
Elderly (65+)	19	1.09 (0.98–1.20)	1.25 (1.07–1.45)	
Respiratory	14	1.13 (0.9–1.42)	1.47 (0.9–2.4)	

Table 5.3. Relative risks of daily cause-specific mortality and elderly mortality associated with non-optimum ambient temperatures, for changes below and above MMT and the 95th and 5th percentile.

Figure 5.4 illustrates the non-linear effects of Tapp_{max} (lag 0–20) on cause-specific and elderly mortality. The exposure–response curves show the typical J-shape with an increase in the risk of mortality for high temperatures. U, J or V-shaped relationships between temperature (or temperature-based thermal indices) and mortality have been identified in many previous studies (Michelozzi *et al.*, 2006; Anderson and Bell, 2009; D'Ippoliti *et al.*, 2010; Breitner *et al.*, 2014; Ma, Chen and Kan, 2014; Zhang *et al.*, 2016), with optimum temperature corresponding to the lowest point in the curve; the shape of the curve may vary by geographic locations, climatic characteristics and demographic factors (Liu *et al.*, 2011; Breitner *et al.*, 2014; Nordio *et al.*, 2015). It is worth noting that respiratory mortality risk reaches very high values (RR > 10), as found in other studies (Zhang *et al.*, 2016).





Figure 5.4. Cumulative exposure–response curves between daily $Tapp_{max}$ and cause–specific mortality and elderly mortality, over lag days 0–20, as calculated for the city of Thessaloniki for the period 2006–2016. 95% CI marked as grey areas.

As shown in Figure 5.5, the mortality risks of extreme cold $Tapp_{max}$, defined as the 5th percentile at 2.5 °C, reach maximum values on day 3 and subsequently decrease until day 10 (all-cause, cardiovascular, elderly) to 15 (cerebrovascular, respiratory).





Figure 5.5. Non–linear effects of extreme cold Tapp_{max} on daily cause–specific mortality and elderly mortality at lag 0–20, as calculated for the city of Thessaloniki for the period 2006–2016. Effects were defined as the risks at 5th percentile of Tapp_{max} distribution compared with the estimated MMT.

Figure 5.6 shows the estimated lag–response curves at the 95th percentile of Tapp_{max} (35 °C) for various causes of mortality and elderly mortality. The graphs indicate strong risk in the first days, followed by a protective association (RR < 1) at longer lags for All-cause, Cerebrovascular, Respiratory and Elderly mortality. For respiratory and cerebrovascular mortality, there is a suggestive mortality displacement, or a harvesting effect, as the initial increase caused by an event of extreme temperatures is followed by a negative risk in mortality (Baccini *et al.*, 2008; Gasparrini *et al.*, 2012; Kouis *et al.*, 2019). The overall shape of the curves is consistent with Kouis et al. who focused on Thessaloniki (Kouis *et al.*, 2019).





Percent change in mortality per 1°C change above and below MMT (Table 5.4) shows a larger increase for all-cause mortality in heat (1.95%, 95% CI: 1.07–2.84), in contrast to smaller (insignificant) increases in cold (0.54%, 95% CI: 0–1.09). Cerebrovascular mortality is largely affected by cold (1.7%, 95% CI: 0.2–3.3), whereas the highest increase in mortality concerns respiratory causes in heat (5.07%, 95% CI: 2.1–8.1), followed by elderly mortality (2.36%, 95% CI: 1.34–3.34). Bunker et al. (Bunker *et al.*, 2016) reported that with 1 °C increase above

the heat-related threshold, respiratory mortality was the most affected category, depicting the largest risk increase, followed by cardiovascular and cerebrovascular causes. This pattern is also evident in the present study. It is interesting to notice that, although heat risk is often associated with effects on the cardiovascular system, relative risks for cardiovascular mortality are no higher than those of most other causes of death, and remarkably lower than respiratory causes, a result found also in Gasparrini et al. (Gasparrini *et al.*, 2012). Statistically insignificant results were obtained for cerebrovascular heat mortality (0.98%, 95% CI: -1.4-3.41), cardiovascular cold mortality (0.97%, 95% CI: -0.1-1.98), and respiratory cold mortality (1.48%, 95% CI: -0.3-3.24).

Table 5.4. Percentage change in mortality per 1°C change in Tapp_{max} above and
below MMT.

Cause of Death	Mortality Heat (% Change)	Mortality Cold (% Change)
All-cause	1.95 (1.07–2.84)	0.54 (0-1.09)
Cardiovascular	2.15 (0.52–3.81)	0.97 (-0.1-1.98)
Cerebrovascular	0.98 (-1.4-3.41)	1.7 (0.2–3.3)
Elderly (65+)	2.36 (1.34–3.34)	0.77 (0.18–1.37)
Respiratory	5.07 (2.1-8.1)	1.48 (-0.3-3.24)

The estimation of the overall attributable fraction categorized per cause of death is reported in Table 5.5. Regarding All-cause mortality, 3.51% of deaths are attributed to temperature. The total attributable risk is then separated into components due to cold and hot temperatures, defined as those below and above MMT, respectively (Gasparrini and Leone, 2014; R. Chen *et al.*, 2018). The comparison of the two contributions clearly indicates that heat is responsible for most of the all-cause mortality attributable to Tapp_{max}, with an value equal of 2.34% compared to 1.34% for cold.

Cardiovascular and cerebrovascular mortality is mostly attributed to cold temperatures, a result which is expected from the high MMTs of 30°C. Mortality concerning the elderly and mortality due to respiratory causes are mostly attributed to heat (3.75% and 5.6% respectively).

 Table 5.5. Cause-specific mortality and elderly mortality fraction (%) attributable to high and low levels of Tapp_{max}.

Cause of Death	MMT (°C)	Overall (%)	Hot (%)	Cold (%)
All-cause	22	3.51	2.34	1.34
Cardiovascular	30	10.1	2.36	8.1
Cerebrovascular	30	14	1.07	13.12
Elderly (65+)	19	5.4	3.75	1.88
Respiratory	14	6.76	5.6	2.04

The analysis is extended by further separating the attributable components into contributions from mild and extreme $Tapp_{max}$ as follows:

- Extreme cold: minimum to 5th percentile.
- Mild cold: 5th percentile to MMT.
- Mild hot: MMT to 95th percentile.
- Extreme hot: 95th percentile to maximum.

As indicated in Figure 5.7, which illustrates the attributable fractions of causes of mortality and elderly mortality associated with different components of non-optimum Tapp_{max}, All-cause and Respiratory mortality are mainly attributed to extreme high values of Tapp_{max} (1.35% and 3.3%, respectively). Mild cold Tapp_{max} accounts for the majority of attributable mortality due to cardiovascular and cerebrovascular causes by far (7.21% and 11.6%, respectively). Lastly, Mild hot Tapp_{max} is mostly responsible for Elderly mortality (2.22%).



Figure 5.7. Fractions of mortality and elderly mortality attributable to non–optimum Tapp $_{max}$.

5.3 Conclusions

In the present study, the effects of $Tapp_{max}$ on daily mortality from 2006 to 2016 in the urban area of Thessaloniki, Greece, were assessed. DLNMs were used to examine the associations between thermal indicator $Tapp_{max}$ and cause-specific mortality, and to investigate the effect of time lag and non-optimum temperatures.

There is a documented association between high temperatures and mortality for Thessaloniki. The present results come to agreement with recent work (World Health Organization, 2022) and contribute further in increasing the knowledge concerning the effects of temperature on daily mortality in the urban area of Thessaloniki: (a) a wide range of mortality categories was investigated in the present study, covering the majority of deaths related to temperature with the use of a state-of-the-art statistical framework, (b) instead of choosing a single temperature parameter, we calculated Tapp_{max} as a thermal index expressing thermal stress and comfort perceived by humans, and (c) we also examined the attributable fraction on non-optimum temperatures.

In Thessaloniki, extreme high temperatures strongly influence the risk in mortality. Strong exponential rise was evident over 35° C, with further and more prominent increase with Tapp_{max} values over 40 °C. J-shaped relationships were found between temperature and mortalities. Heat was short lived with an immediate effect (mortality increase), whereas cold had a prolonged effect.

Percent change in mortality per 1 °C change above and below MMT showed large increase for all-cause, cardiovascular and respiratory mortality in heat. Similarly, elderly mortality was also affected by heat. Taking into consideration the attributable fraction of mortality assigned to the various causes investigated, it was noted that overall 3.51% deaths were attributable to Tapp_{max}. Deaths attributed to heat outnumbered deaths attributed to cold, with stronger impacts on respiratory mortality and among the elderly.

In conclusion, heat-attributable mortality in Thessaloniki was found to be mainly associated with high temperatures. Given the eminent vulnerability of the Mediterranean area to temperature rise due to climate change and the expected more frequent occurrence of heat wave episodes (Ullah *et al.*, 2022), these results could contribute to the decision-making process regarding extreme weather preparedness. The need to conduct heat health action plans for the implementation and coordination of extreme heat response activities that will reduce the negative impact of extreme heat is undeniable and requires high-resolution information on the association between temperature and mortality on the city-level.

The results of the present study can effectively contribute to this scope, whereas future work should be conducted to include not only major Greek cities but additional environmental stressors such as air pollutants as well, in order to draw decisive conclusions on the national frame. Investigating mortality behavior under the influence of climate change could be a critical next step.

Chapter 6

Short-Term Effects of Air Pollution on Mortality in the Urban Area of Thessaloniki, Greece

In recent years, poor air quality, both ambient and indoor, has become a pressing issue, with more frequent and intense episodes of high pollution levels being prevalent in cities across the globe. Currently, it is considered the biggest environmental risk to human health and the second-greatest environmental concern among Europeans, second only to climate change (European Environmental Agency, 2020).

According to the WHO (WHO, 2016), 3 million deaths were solely attributable to outdoor air pollution globally in 2012, an estimation which Fuller et al. (Fuller *et al.*, 2022) has raised to 4.5 million, particularly for ambient particulate matter (4.14 million) and ambient ozone (0.37 million). This not only impacts mortality but morbidity as well. The most compelling evidence regarding the health consequences of air pollution relates to cardiovascular and respiratory ailments; nevertheless, studies exploring other health impacts are also increasing (WHO, 2022; WHO Regional Office for Europe, 2022). Older adults are more susceptible to the negative health impacts of air pollution due to their decreased ability to adapt to stressors on their physiological, metabolic, and compensatory processes, as well as their higher likelihood of having cardiovascular and respiratory diseases (Geller and Zenick, 2005; Shumake *et al.*, 2013). Elderly mortality has been found to be particularly affected by PM10 and O₃, with higher excess risks than other age groups (Cakmak, Dales and Vidal, 2007; Katsouyanni *et al.*, 2009; Liu *et al.*, 2022; Olstrup, Åström and Orru, 2022).

PM10 and O_3 are considered to represent a major part of the problem (Stanaway et al., 2018). Ozone exposure has significantly increased worldwide, leading to a 46% increase in ozone-attributable mortality from 2000 to 2019 (Malashock et al., 2022). PM10 and O_3 are linked to a rise in all-cause, cardiovascular, and respiratory mortalities (Biggeri et al., 2004; Katsouyanni et al., 2009; Garrett and Casimiro, 2011; Peng et al., 2013; Olstrup et al., 2019; Vicedo-Cabrera et al., 2020). The WHO has also emphasized PM2.5 (World Health Organization, 2006) as they are found to be associated with the premature mortality of several age groups (Garrett and Casimiro, 2011; Pascal et al., 2014; Lelieveld et al., 2015). In 2020, approximately 238,000 premature deaths in the European population were caused by exposure to PM10 concentrations above the WHO's 2021 guideline level of 45 μ g/m³ (EEA, 2022a). Additionally, the European Environment Agency attributed 16,800 premature deaths to acute ozone exposure in 2019 (EEA, 2021). Despite EU and national policies, the pollutant levels in many areas exceed the recommended guidelines (European Council Directive 2008/50/EC), and although significant improvements are evident, the impacts of serious air pollution in Europe still persist. Approximately 11% and 12% of the EU urban population is exposed to PM10 and O₃ concentrations above EU standards, a percentage that rises to 71% and 95%, respectively, when taking into account the WHO guidelines of 2021 (EEA, 2022b).

Threshold violations take place at several locations throughout Europe (EEA, 2012). However, the problem appears to be more pronounced in Southern and Eastern Europe (EEA, 2018), especially with respect to the PM and ozone concentrations in Greece, Spain, and Italy (Pleijel, 2000; Karanasiou *et al.*, 2014; Sicard *et al.*, 2021). These areas are characterized as climate change hotspots; thus, the collective impacts of climate change and air pollution variables should be taken into consideration (Bytnerowicz, Omasa and Paoletti, 2007) under the specific topographical and meteorological conditions of each region (Valjarević *et al.*, 2021). With respect to the latter air pollutant, the ground-level ozone concentrations in Southern Mediterranean countries are often alarmingly high and are comparable to the highest levels of places that are located in the most contaminated parts of Central Europe (Moussiopoulos *et al.*, 2009).

When focusing on Greece, the country has been found to be in violation of the three most commonly exceeded EU air quality standards for PM10, O_3 , and NO_2 , according to (EEA, 2019). Using 2019 data, it was estimated that 75 deaths per 100,000 population in Greece could be attributed to air pollution, where the deaths were primarily caused by ischemic heart disease, stroke, and respiratory infections (WHO Regional Office for Europe, 2022); this corresponds to 1,101 attributable DALYS (Disability-adjusted life years) per 100,000 citizens (WHO, no date a), or 104,000 YLLs (Years of life lost) for the entire Greek population (WHO, no date b). As expected, the two largest cities of the country suffer the most from the acute air quality problems because of the dense population and build-up of air pollutants caused by the topography and adverse meteorological conditions, e.g.,

the urban heat island effect (Poupkou *et al.*, 2011). The EU air quality standards are significantly surpassed by the PM10 concentrations observed in Athens and Thessaloniki (Kalabokas, Adamopoulos and Viras, 2010), and the YLLs are primarily affected by PM10 exposure as well as O_3 to a lesser extent (Kassomenos, Dimitriou and Paschalidou, 2013).

Thessaloniki in particular is one of the most polluted cities in Europe, especially with respect to the PM level (Vlachokostas *et al.*, 2012) but also with respect to the O₃, VOCs, and noise pollution levels (Vlachokostas *et al.*, 2013). O₃ limit values are mostly exceeded during the summer months, while winter is the most favorable season for PM10 violations (Moussiopoulos *et al.*, 2009). Nevertheless, Thessaloniki's major air quality problem consists of PM10 concentration levels. As a result, in December 2020, the European Commission decided to take legal action against Greece by referring the country to the European Court of Justice for the substandard PM₁₀ air quality of Thessaloniki (European Commission, 2020).

To address the issue, effective and enduring air pollution mitigation plans must be identified and put into action (Miranda *et al.*, 2015; Silveira *et al.*, 2016). Such measures and policies to combat particulate air pollution were tested in a recent study (Progiou *et al.*, 2023), which resulted in a more than 20% reduction in the PM10 concentrations in Thessaloniki, Greece. Moreover, it is necessary to assess the health benefits of the abatement measures by quantifying the impact of air pollution on human health.

However, majority of the literature focuses almost entirely on Athens (e.g., (Gryparis *et al.*, 2004; Touloumi *et al.*, 2006; Kassomenos, Dimitriou and Paschalidou, 2013)); only recently has a study by (Psistaki *et al.*, 2022) discovered that brief exposure to PM2.5 and PM10 in Thessaloniki is connected to an amplified risk of all-cause and cardiovascular mortality. In addition to the above, there is a significant lack of studies specifically examining the suitability of mitigation measures in terms of health benefits for the area of Thessaloniki.

In this Chapter, we utilize advanced statistical tools to investigate the associations between short-term exposure to PM10 and O_3 and daily all-cause (natural, non-accidental), cardiorespiratory, and cerebrovascular mortality from 2006 to 2016 in the urban area of Thessaloniki. We also examine the effect of air pollution on the elderly (all-causes, 65+ years) as it is crucial to understand the specified response of frail subgroups to environmental stressors.

Most importantly, to assist air quality planning, we estimate for the first time the impact of the modification of PM10 levels on Thessaloniki's population mortality under two air pollution abatement scenarios: (1) full compliance to EU levels, thus eliminating the exceedances of PM10 daily values; and (2) a 20% horizontal reduction in the PM10 concentration in order to assist air quality planning. These scenarios are based on the most cost-efficient measures identified by the recent study of (Progiou *et al.*, 2023) to combat PM10 pollution in the urban area of Thessaloniki.

Thus, the main goal of the current Chapter is to present evidence on the air pollution-mortality relationship in the Thessaloniki urban area, accounting for the cause-specific deaths, lag structure, elderly mortality, and potential mitigation measures that can be of utmost importance for environmental stakeholders and local policy makers

6.1 Data and Methods

6.1.1 Study area

This research centered on the urban area of Thessaloniki (Figure 6.1), which includes seven municipalities. Thessaloniki, the second largest city in Greece and an important economic and industrial center in the Balkans, is situated in the northern part of the country and has a population of about 1,000,000, representing 20% of the country's industrial activity (Moussiopoulos et al., 2009). The city is located on the northeastern coast of the Thermaikos Gulf and is close to Hortiatis mountain (1200 m) on the eastern side. The western side is characterized by a large flat area, which houses the industrial zone of Sindos. The city's location to the south means that it is greatly affected by the nearby sea, which contributes to its Mediterranean climate (Giannaros and Melas, 2012). Vehicular traffic, residential heating (Progiou et al., 2023), biomass burning (Diapouli et al., 2017), and industrial emissions (Moussiopoulos et al., 2009) are the main origins of air pollutants in Thessaloniki (Melas et al., 2017), resulting in the deteriorated air quality in the area, especially during years of economic crisis (Zyrichidou et al., 2019). Dust storms originating from North Africa also significantly contribute to particle pollution in the area (Achilleos et al., 2020; Psistaki et al., 2022; Rizos et al., 2022).



Figure 6.1. The study area with the locations of air quality stations (1: Agias Sofias, 40.63° N 22.94° E; 2: AUTh, 40.63° N 22.95° E; 3: Panorama, 40.58° N 23.03° E; 4: Kalamaria, 40.57° N 22.96° E; 5: Kordelio, 40.67° N 22.89° E). Dotted lines represent municipalities.

6.1.2 Air Quality and Mortality Data

The hourly values of PM10 and O_3 concentrations ($\mu g/m^3$) for the period of 2006–2016 were acquired by 5 air quality monitoring stations that cover the urban area of Thessaloniki and are operated by the Ministry of the Environment and Energy. The highest PM10 value and maximum 8-hour moving average for O_3 over each station were used in the present study, which represented the daily concentrations for the datasets.

The Hellenic Statistical Authority (ELSTAT) provided the daily mortality data, consisting of age and cause of death, for all municipalities in the urban region of Thessaloniki (2006–2016); the causes of death were categorized into all-cause (natural, non-accidental), cardiorespiratory, and cerebrovascular according to the ICD-10. Emphasis was placed on studying the overall mortality rate among the elderly, specifically for deaths that occurred among individuals aged 65 years and older.

6.1.3 Data analysis

We applied a DLNM to our data in order to show the impact of air pollution on mortality with delay in time, in accordance to previous studies (Gu *et al.*, 2020; Vicedo-Cabrera *et al.*, 2020; Kriit *et al.*, 2022). DLNMs are a powerful modeling tool that are capable of simultaneously capturing both non-linear exposure–
response dependencies and delayed effects. Unlike conventional distributed lag models, which struggle with non-linear relationships, the DLNM methodology utilizes a 'cross-basis', a two-dimensional function space that depicts the connection between predictor variables and the lag dimension of their occurrence. This approach offers a comprehensive portrayal of the exposure–response relationship's time course, making it possible to estimate the overall effect with precision, even in the presence of delayed contributions. In order to describe the air pollution–mortality associations in the present study, we applied generalized non-linear models with a quasi-Poisson family based on the quasi Akaike information criterion. The DLNM package (Gasparrini *et al.*, 2017) in R programming language (R version 4.1.1; R Foundation for Statistical Computing) was used to implement the family of applied models.

There are differences in the literature regarding the lag structure used to best describe the association between air pollution and mortality; in some cases, short lags of 0–1 days (Katsouyanni *et al.*, 2001; Ballester *et al.*, 2002; Analitis *et al.*, 2006; Lelieveld *et al.*, 2015; Khaniabadi *et al.*, 2017; Liu *et al.*, 2019) or up to 3 days (Garrett and Casimiro, 2011) are deemed to be the most appropriate, while in other studies, a week is chosen (Samoli *et al.*, 2009; Stafoggia *et al.*, 2010; Gariazzo *et al.*, 2023). There are also examples in the literature suggesting that the adverse response to pollution persists for more than a month (Zanobetti *et al.*, 2002, 2003; Stojić *et al.*, 2016). To this end, we investigated the correlation between short-term exposure to PM10 and O₃ and specific causes of death at various lags in order to decide the effect estimates for the present analysis.

In order to investigate the efficiency of mitigation measures in terms of health benefits, we not only applied the DLNM analysis for the original PM10 dataset, but also for 2 mitigation scenarios:

- Complete compliance with the EU limits (daily PM10 value $< 50 \,\mu g/m^3$),
- 20% reduction in the PM10 concentration.

Table 6.1 shows the percentage of days in which the daily EU limits were exceeded during the range of 2006–2016. The EU air quality guidelines were surpassed on 1894 (47%) days of the 4018-day study period for PM10 (>50 μ g/m³) and on 1124 (28%) days for O₃ (>120 μ g/m³). Under the 20% PM10 reduction scenario, only 27% of days surpassed EU limits, resulting in 1119 exceedances.

	O ₃ Mean Annual	PM10 Mean Annual	PM10 20% Reduction
Year	Value (%—Days	Value (%—Days Over	Scenario (%—Days Over
	Over 120 μg/m ³)	50 μg/m ³)	$50 \ \mu g/m^{3}$)

2006	85.8 (6.5%)	58.9 (53%)	47.1 (34%)
2007	95.3 (18%)	70.5 (79%)	56.4 (53%)
2008	118.7 (48%)	66.9 (76%)	53.5 (51%)
2009	112.7 (45%)	56.1 (57%)	44.9 (28%)
2010	99.1 (26%)	51.2 (39%)	41 (20%)
2011	114 (50%)	56.8 (45%)	45.5 (25%)
2012	115.2 (47%)	52.6 (43%)	42 (24%)
2013	101 (30%)	48.7 (35%)	39 (21%)
2014	82.2 (2.5%)	46.6 (30%)	37.3 (14%)
2015	99.5 (25%)	49.7 (33%)	39.8 (17%)
2016	92.3 (9%)	47 (31%)	37.6 (15%)

Table 6.1. Mean annual values of pollutants $(\mu g/m^3)$ during the study period and during PM10 reduction scenario 2. Numbers in parentheses denote the percentage of annual violations of the EU daily limits.

6.2 Results

6.2.1 Mortality Data analysis

During the study period, we analyzed 73,990 natural deaths that occurred from all causes, 28,945 from cardiorespiratory diseases, and 10,007 from cerebrovascular causes. The number of deaths among the elderly population amounted to 62,482. The descriptive statistics of the pollution and daily mortality for the reference period are provided in Table 6.2.

Table 6.2. Statistics of the daily mortality (number of deaths, top) and pollution $(\mu g/m^3, bottom).$

	Daily Mortality	
	Mean	St. dev.
All-cause	18.4	4.7
Cardiorespiratory	7.2	2.9
Cerebrovascular	2.5	1.6
Elderly	15.5	4.4

PM10					
Median	Mean	Min	25th perc.	75th perc.	Max
49	55	11.6	38	65	256.6
			O ₃		
Median	Mean	Min	25th perc.	75th perc.	Max
99	101	14	76	123	232

The data on deaths show that cardiorespiratory mortality accounts for over 40% of all natural deaths, making it a crucial group to examine in terms of susceptibility; the authors of (European Observatory on Health Systems and Policies, 2019) have reported that stroke and ischemic heart disease are the primary causes of mortality in Greece, which supports the claim. Elderly mortality reflects 84% of all-cause mortality for all ages, as Greece has one of the highest percentages of individuals aged over 65 years in Europe (EUROSTAT, 2020). The daily mean and median pollutant concentrations are generally higher in Thessaloniki than those reported in other metropolitan areas (Garrett and Casimiro, 2011; Samoli *et al.*, 2014) and resemble the values of large cities with important air quality issues (Samoli *et al.*, 2009; Stafoggia *et al.*, 2010; Wang *et al.*, 2018). Similar values of mean daily mortality and summary statistics of PM10 in Thessaloniki are also verified in (Samoli *et al.*, 2013).

6.2.2 Lag Effect analysis

Table 6.3 displays the correlation between short-term exposure to PM10 and O_3 and specific causes of death at various lags. The lag structure here yields a prolonged effect of PM10 and O_3 on all mortalities from the current day to day 6 in Thessaloniki. As a result, the relative risk per 10 µg/m³ increase in PM10 and O_3 concentrations over lag 0–6 is used hereinafter as the effect estimates.

Table 6.3. Associations between cause-specific mortality and short-term exposure to PM10 and O₃ at various time intervals (0–1, 1–6, and 0–6 days). Results are presented as a percentage increase of risk (RR%) and as 95% confidence intervals (95% CI) per 10 μ g/m³.

PM10						
Mortality	RR%, Lag 0–1	RR%, Lag 1–6	RR%, Lag 0–6			

All-cause	2.2 (0.6–3.3)	0.8 (-1.9-3.2)	2.3 (0.8–3.8)
Cardiorespiratory	1.9 (0.5–3.6)	0.7 (-2-3.5)	2 (0.1–4.5)
Cerebrovascular	1.5 (-1.8-5.2)	1.1 (-1.9-4.9)	1.8 (-2-6.1)
Elderly	2.7 (1-3.5)	1.5 (-0.4-2.9)	3.2 (1.5–5)
		O ₃	
Mortality	RR%, Lag 0–1	RR%, Lag 1–6	RR%, Lag 0–6
All-cause	1.9 (0.9–3)	2.7 (0.2–5.3)	3.9 (2.5–5.3)
Cardiorespiratory	2.8 (1.06-4.5)	3.5 (-0.54-4)	5.3 (3.1–7.7)
Cerebrovascular	-0.7 (-3.5-2.23)	2.8 (-4-9.9)	3 (-7-11)
Elderly	2.2 (1.55-3.4)	3 (0-5.86)	4.4 (2.9–6)

The estimated associations between the PM10, O_3 , and mortality in Thessaloniki are illustrated in Figure 6.2. The diagrams show the relationship among the air pollutants concentrations, excess risk, and lag values as a three-dimensional surface. The associations of PM10, all-cause, and cardiorespiratory mortalities are non-linear. An immediate increase in deaths is evident for exposures to high levels of pollutants at lag days 0–2; however, for cardiorespiratory causes, a secondary maximum is present at lag 6. Concerning O_3 , a lag of up to 3 days depicts a large increase in excess risk, which results in higher values of cardiorespiratory deaths. At days 5–6, a smaller increase is evident for both causes of death, indicating a prolonged impact.





Figure 6.2. Overall effect of PM10 on all-cause mortality (a) and cardiorespiratory mortality (b); overall effect of O₃ on all-cause mortality (c) and cardiorespiratory mortality (d) for Thessaloniki in the years 2006–2016.

The dose–response relationships for the natural and cardiorespiratory mortalities for PM10 and O_3 (not shown here) were found to be linear, as noted in previous studies (Gryparis *et al.*, 2004; Ortiz *et al.*, 2017; Psistaki *et al.*, 2022).

6.2.3 Total effect analysis

We present the evidence of the positive association of natural all-cause and cardiorespiratory deaths with PM10 and O_3 in Table 6.3.

A 10 unit increase in PM10 is associated with a 2.3% (95% CI: 0.8–3.8) increase in natural all-cause mortality and a 2% (95% CI: 0.1–4.5) increase in cardiorespiratory mortality; O₃ causes a 3.9% (95% CI: 2.5–5.3) increase in allcause mortality and a 5.4% (95% CI: 3.1–7.7) increase in cardiorespiratory mortality. Neither of the two air pollutants are associated with cerebrovascular outcomes, as confirmed in similar studies (Stafoggia *et al.*, 2010; Samoli *et al.*, 2014).

Due to the significant differentiation of the lag selection, there is no uniformed way to compare our results with other studies. PM10 levels are generally associated with increases of 0.8–4.3% in all-cause mortality, 0.12–6.6% in cardiovascular mortality, and 0.47–4.2% in respiratory mortality, respectively (Kunzli *et al.*, 2000; Katsouyanni *et al.*, 2001; Ballester *et al.*, 2002; Zanobetti *et al.*, 2002, 2003; Analitis *et al.*, 2006; Pascal *et al.*, 2014; Khaniabadi *et al.*, 2017); the RR estimations in the present study are found to be within the range demonstrated above. Thessaloniki is underrepresented in similar publications; ref. (Psistaki *et al.*, 2022) linked exposure to PM10 to a 1.75% increase in cardiovascular deaths (lag 0–6) but found no link to respiratory mortality.

Many studies (Gryparis et al., 2004; Samoli et al., 2008, 2009; Stafoggia et al., 2010; Garrett and Casimiro, 2011; Khaniabadi et al., 2017; Olstrup et al., 2019)

have reported positive associations between O_3 and increases in all-cause (0.33–2%), cardiovascular (0.45–2.5%), and respiratory mortalities (0.6–2.8%.), and correlations are evident in the present analysis. In particular, ref. (Stafoggia *et al.*, 2010) indicated higher impacts of O_3 on respiratory and cardiac mortality than on all-cause mortality, which is also confirmed by our results. However, the excess risks estimated here are higher compared with those obtained in other studies.

It is worth noticing, however, that the estimates from single-city studies tend to be higher compared with pooled multi-city results as the model specification utilized in the studies focused on individual cities could result in an overestimation of the outcome (Katsouyanni *et al.*, 2001; Samoli *et al.*, 2008).

When comparing the effect of O_3 and PM10 on different causes of mortality, we document more severe impacts from the former than the latter. This consistent behavior is evident in similar studies covering various areas worldwide and various time spans, e.g., South Africa (2006–2015) (Christina Adebayo-Ojo *et al.*, 2006), Russia (2003–2005) (Revich and Shaposhnikov, 2010), and China (Shang *et al.*, 2013).

Susceptible population subgroups are often separately considered in order to account for the specified behavior of these groups to environmental stressors. In the present work, we developed a dedicated DLNM model for assessing the impact of PM10 and O_3 on the elderly.

Elderly mortality is affected by both PM10 and ozone; a 3.2% RR increase (95% CI: 1.5–5) per 10 unit increase of PM10 and a 4.4% raise (95% CI: 2.9–6) per 10 unit increase of O_3 are evident. Similar results are verified in (Katsouyanni *et al.*, 2009; Stafoggia *et al.*, 2010; Garrett and Casimiro, 2011; Shang *et al.*, 2013). The air pollution in Thessaloniki has been found to demonstrate a more intense impact on elderly mortality than on the all-cause mortality for all ages, as found in (Revich and Shaposhnikov, 2010).

Additionally, 5% of elderly deaths are attributed to PM10 and 2.6% are attributed to O_3 (a total of 4,750 deaths out of 62,482). This corresponds to 284 and 146 annual deaths due to PM10 and O_3 , respectively, for people aged 65 years and older.

Table 6.4 presents the attributable mortality and attributable fraction of mortality based on the PM10–mortality and O_3 –mortality relationships. We estimate that 3.6% of total mortalities and 3.2% of cardiorespiratory causes were attributed to PM10, while the respective percentages for O_3 are 2.3% and 3%. These estimates correspond to 242 annual premature all-cause mortalities from PM10 and 170 from O_3 , respectively. On an annual basis, 82 cardiorespiratory deaths are related to elevated PM10 levels, and another 80 cardiorespiratory deaths are related to O_3 levels. Overall, in Thessaloniki, 412 deaths are recorded annually due to PM10 and O_3 pollution, out of which 162 are attributed to cardiorespiratory causes.

	PM10			03		O ₃
Mortality	ortality AM AF Average Annual Deatl		Average Annual Deaths	AM	AF	Average Annual Deaths
All-cause	2664	3.6	242	1865	2.3	170
Cardiorespiratory	914	3.2	82	876	3	80
Elderly	3146	5	284	1604	2.6	146

Table 6.4. Attributable mortality (AM, number of deaths) and attributable mortality fraction (AF, %) for different causes of mortality.

Our results are similar to previous studies, where the attributable fraction of natural mortality fluctuated between 1.35% and 6% and cardiovascular mortality fluctuated between 1.63% and 6.89% due to PM10 pollution (Christina Adebayo-Ojo *et al.*, 2006; Wang *et al.*, 2018). Ref. (Khaniabadi *et al.*, 2017) reported that 1.96% of cardiovascular mortality is attributed to O_3 and 6.6% to PM10, while (Goudarzi *et al.*, 2015) found that 3.2% of cardiovascular and 6.2% of respiratory mortality is attributed to O_3 . According to (Fattore *et al.*, 2011), 2% of cardiovascular mortality, 5.6% of respiratory, and 1.5% of total mortality is attributed to O_3 levels.

6.2.4 Suitability of Studied Scenarios in Terms of Health Benefits

We present an examination of the suitability of two mitigation measures in terms of their health benefits for the urban area of Thessaloniki. The first case study (scenario 1) corresponds to a full abidance to EU limits concerning daily PM10 values ($<50 \ \mu g/m^3$), whereas the second case study (scenario 2) horizontally reduces PM10 concentrations by 20%, a case that is more realistically applicable as shown in (Progiou *et al.*, 2023).

Table 6.5 displays the RR, AF, and AM for scenarios 1 and 2, respectively. Reducing PM10 concentrations by 20% would result in 2368 deaths and a 3.2% AF value with respect to total mortality. Full compliance with EU environmental legislation leads to a 1.8% attributable all-cause mortality, which corresponds to 710 deaths. When comparing the scenarios, the RR increases from 1.7% (scenario 1) to 2.1% (scenario 2). It is obvious that radical measures positively affect human health to a larger degree than moderate ones.

Scenarios	RR	AM	AF	
1—Full EU compliance	1.7	710	1.8	
2—20% reduction	2.1	2368	3.2	

Table 6.5. RR (%), AM (number of deaths), and AF (%) of total mortality for
different PM10 scenarios.

When comparing the results of Table 6.3 and 6.4, the mortality burden decreases when mitigation measures are implemented. The AF is reduced by 0.4% and 1.8% compared with the original PM10 dataset for the 20% reduction and full compliance scenarios, respectively. Thessaloniki would count 27 less deaths on an annual basis if the PM10 concentration were reduced by 20% and 177 less annual deaths if under full EU compliance.

Thus, even with the more moderate abatement scenario, the health impact of PM10 concentration on the local population could be significantly lower.

6.3 Discussion

In the international literature, the interaction between human health and air quality is well-defined (WHO Regional Office for Europe, 2013) with respect to morbidity and mortality (Héroux *et al.*, 2015). The adverse impact of deteriorated air quality has also raised international concern with respect to the natural environment (Bytnerowicz, Omasa and Paoletti, 2007) and economy (Font-Ribera *et al.*, 2023). Cities in the Mediterranean area are frequently experiencing elevated levels of air pollution (Pleijel, 2000) under the additional pressure of the climate crisis. Thessaloniki, Greece, is particularly impaired with respect to the air pollution, especially due to PM10 and O₃ levels (Vlachokostas *et al.*, 2012; Karanasiou *et al.*, 2014). Although some recent studies have quantified the impact of temperature on mortality (Orellano *et al.*, 2020; Parliari *et al.*, 2022), there is insufficient evidence concerning air quality, thus pointing a gap in relevant knowledge.

The present study aimed to address this vacancy by presenting an evaluation of the short-term changes in daily mortality counts as associated with the concentrations of daily air pollutants from 2006 to 2016 in the urban area of Thessaloniki. We analyzed the associations between the daily maximum values of PM10 and O_3 levels and cause-specific mortality, and we investigated this effect on the susceptible elderly subgroup with the use of DLNMs. To quantify the mortality burden, we used relative risk changes for every 10 µg/m³ increase in air

pollution concentrations as the primary effect estimates (Braga, Zanobetti and Schwartz, 2001; Orellano *et al.*, 2020). After conducting a specific analysis using a lag structure, which has great heterogeneity among literature, we determined the most suitable lag for this work to be defined at days 0–6, similar to other studies (Stafoggia *et al.*, 2010; Gariazzo *et al.*, 2023).

Based on our results, a 10 unit increase $(\mu g/m^3)$ in PM10 concentration is associated with a 2.3% (95% CI: 0.8–3.8) increase in natural all-cause mortality and 2% (95% CI: 0.1–4.5) increase in cardiorespiratory mortality. O₃ causes increases of 3.9% (95% CI: 2.5–5.3) in all-cause mortality and increases of 5.4% (95% CI: 3.1–7.7) in cardiorespiratory mortality. Meanwhile, neither of the two air pollutants is associated with cerebrovascular outcomes. Considering the assigned attributable fraction of mortality for the various investigated causes, it is noted that overall, 3.6% of total mortalities are attributable to PM10 and 2.3% are attributable to O₃. PM10 levels are responsible for 3.2% of cardiorespiratory mortality (3% for O₃). These estimations correspond to 242 annual premature allcause casualties due to PM10 and 170 due to O₃.

The direct comparison of our findings with similar studies in this field is particularly challenging due to the differentiation of the lag selection and underrepresentation of the specific area. Nevertheless, both RR estimates and attributable mortalities are in agreement with comparable research (Kunzli *et al.*, 2000; Zanobetti *et al.*, 2003; Stafoggia *et al.*, 2010; Garrett and Casimiro, 2011). It is worth noting that (Psistaki *et al.*, 2022) linked exposure to PM10 to a 1.75% increase in cardiovascular deaths (lag 0–6) but found no link to respiratory mortality in the Thessaloniki area.

Elderly mortality is also affected by the 10 unit increase in the air pollutants to an even larger degree than the mortality accounting for all ages, which was also confirmed in (Revich and Shaposhnikov, 2010). We report that excess risks increase by 4.4% and 3.2% due to O_3 and PM10, respectively, while 284 annual deaths are attributed to PM10 and 146 are attributed to O_3 , corresponding to a 5% and 2.6% attributable mortality, respectively. Studies on elderly people, such as (Katsouyanni *et al.*, 2009; Garrett and Casimiro, 2011), report similar results.

The need to abide by EU environmental legislation is crucial for reducing the negative impact of air pollutants on public health (Progiou *et al.*, 2023); thus, high-resolution, location-specific information on the association of human morbidity and mortality to environmental stressors is of utter importance. Appropriate mitigation actions should be taken to decrease the population's exposure to pollutants and to further explore how location-specific factors contribute to this vulnerability. An innovative aspect of this work is the quantification of the health benefits as a result of two PM10 abatement scenarios, which was conducted for the first time in the study's urban area. The first case study (scenario 1—full abidance to EU limits, 50 μ g/m³) yields 177 less annual

deaths, and the second case study (scenario 2—horizontal reduction by 20%) results in 27 less casualties compared with the baseline.

The above findings of the present study clearly indicate that local residents are at risk from the current levels of PM10 and ozone. O_3 is found to have a more severe impact than PM10, and the elderly are particularly frail to poor air quality in the area. If the two proposed mitigation measures were implemented, the attributed mortality fraction would decrease by 0.4% and 1.8%, respectively.

It should be noted that this study is limited by the fact that no confounding effects (e.g., temperature and humidity) were considered during the modeling process.

Future work should be conducted to include more air pollutants such as PM2.5 and to further study the synergy between thermal stress and air pollution on health so as to draw decisive conclusions. Examining the impact of climate change and projected air quality conditions on mortality patterns could be a crucial next step.

6.4 Conclusions

While there is considerable literature on the impact of air pollution on human health, the case of Thessaloniki, Greece, is considerably under-studied, despite it being a city with significantly deteriorated air quality. By exploring the link between short-term exposure to air pollutants and cause-specific mortality, the current study offers proof of a positive association between daily mortality from natural and cardiorespiratory causes and exposure to PM10 and O₃. However, no connections were identified between these pollutants and cerebrovascular mortality. The study indicates that the elderly population is particularly vulnerable to the effects of PM10 and O₃. To further contribute to policy-making-associated knowledge for a sustainable environment for humans, the study quantified the health benefits that resulted from two air pollution abatement scenarios and found a significant reduction in total excess mortality. The respective results demonstrate significant decreases in air quality-related mortality, highlighting the importance of appropriate civil protection actions based on scientific expertise tailored to local populations for the development of proper health and air quality plans.

Chapter 7

Synergistic impact of air quality and thermal conditions on human mortality

Rapid urbanization rate has caused cities to face increasing pressure from growing populations, limited resources and escalating impacts of climate change (World Health Organization, 2017). The degradation of air quality has become so prominent in cities that the term urban pollution island (UPI) was recently introduced, to describe the spatial and temporal variations in pollution concentrations between urban and rural areas (Ulpiani, 2021). The synergistic interactions of urban heat islands (UHIs) and UPIs lead to increased pollutant concentrations via mechanisms, such as accelerated atmospheric chemistry cycles due to high temperatures and increased emissions of precursor species (Ulpiani, 2021).

The Mediterranean region faces poor air quality conditions, associated with a range of health challenges. In 10 Mediterranean cities, an increase in PM2.5 was linked to higher all-cause and respiratory mortality, particularly among those aged 75 years and above (Samoli *et al.*, 2013). PM10 was associated with increased mortality from various causes such as diabetes, cardiac, cerebrovascular, and chronic obstructive pulmonary disease (Samoli *et al.*, 2014). Rising temperatures in the area contribute to increased pollutant production, mostly ozone (Pyrgou, Hadjinicolaou and Santamouris, 2018). Climate change projections suggest that the Mediterranean will become a prominent "hot spot," with scorching summers becoming more common (Lelieveld *et al.*, 2012). The region will experience a temperature increase of 2° C to 5° C and a 40% decrease in precipitation, especially

in southern Europe (Lelieveld *et al.*, 2012; Zanis *et al.*, 2015). Greece is particularly susceptible to climate change, and is projected to experience a 4.3°C increase in temperature as well as reduced precipitation (Gao, Pal and Giorgi, 2006; Georgoulias *et al.*, 2022).

Thessaloniki, one of Europe's most polluted cities (Ministry of the Environment and Energy, 2023), faces unfavorable meteorological and air quality conditions. For instance, heat was associated with increased cardiovascular and respiratory mortality risk (Kouis *et al.*, 2019). The city also struggles with air pollution, primarily from O₃ in summer and PM10 in winter (Moussiopoulos *et al.*, 2009; Kalabokas, Adamopoulos and Viras, 2010). PM10 has consistently exceeded EU limits, resulting in legal consequences (Case C-70/21). Exposure to such pollution levels has negative health impacts, including increased hospital admissions and mortality rates for cardiac and respiratory diseases (Giannaros *et al.*, 2011; Psistaki *et al.*, 2022; Parliari, Giannaros, *et al.*, 2023). The interaction between air quality and thermal stress is crucial in Thessaloniki (Zoumakis *et al.*, 2011), which also experiences a strong Urban Heat Island Effect (Giannaros and Melas, 2012). The combination of air pollution and heat-discomfort conditions during heat waves leads to high discomfort levels in the city (Papanastasiou, Melas and Kambezidis, 2015).

The available evidence concerning the potential interactive effects of heat and ambient air pollution remains largely inconclusive. Previous studies mostly focused on the individual impact of environmental stressors on human health (Bao *et al.*, 2016; Kriit *et al.*, 2022), with only a few examining the combined influence of temperature and daily air pollution levels (Stafoggia *et al.*, 2008; Cheng and Kan, 2012; Meng *et al.*, 2012; Turner *et al.*, 2012; Breitner *et al.*, 2014; Jhun *et al.*, 2014). Different publications present varying perspectives, with some suggesting synergistic effects (Baccini *et al.*, 2008; Cheng and Kan, 2012), others finding no combined impact (Samet *et al.*, 1998; Hales *et al.*, 2000; Basu, Feng and Ostro, 2008), and some reporting effect modifications for only one of the pollutants under study (Breitner *et al.*, 2014).

The diverse perspectives on this matter continue to be a subject of debate within academic literature. Moreover, there has been relatively limited investigation into the simultaneous interactive effects of temperature and air pollution as predictors of mortality. Recent research investigating synergy has shown increased death numbers at high temperatures and high levels of O_3 or PM10, but changes in cold effects with increasing air pollution levels were not reported (Burkart *et al.*, 2013a; Scortichini *et al.*, 2018; Rai *et al.*, 2023).

The present study investigates the interactive effects between daily maximum Apparent Temperature (Tapp_{max}) and air pollution (NO₂, O₃ and PM10) in cause-specific mortality (non-accidental, cardiovascular, respiratory), by developing a Distributed Lag Non-linear Model using the framework of Generalized Additive

Models. A separate analysis is conducted for the Elderly citizens (> 65 years old), as well as evaluation of the lag structure.

The synergy is estimated by introducing a tensor product between Tapp_{max} and either PM10 (lag 0–6), NO₂ (lag 0–6) or O₃ (lag 0–8). Temperature estimates were extrapolated at low, medium, and high levels of pollutants defined as the 5th, 50^{th} , and 95^{th} percentile of pollutant-specific distribution; heat and cold effects were estimated as the percentage change in mortality between the 75th and 99th, and the 1st and 25th percentiles of Tapp_{max}, respectively. This is the first application of the proposed GAM-based approach and the first analysis to explore this complex association for Thessaloniki, Greece.

7.1 Data and Methods

7.1.1 Study area

The second largest city in Greece and an important economic and industrial center in the Balkans, Thessaloniki is home to more than 1,000,000 inhabitants (Hellenic Statistical Authority, 2023). The city is notable for its aging population, with 21.3% of the population older than 64 (Kouis *et al.*, 2019).

Its Mediterranean climate is significantly affected by the adjacent Thermaikos Gulf in the south; the mean annual temperature of Thessaloniki is ~16 °C and the mean annual relative humidity is ~62%. The city is affected by a prominent UHI effect with an intensity of between 1 °C and 3 °C in winter and 2°C and 4°C in summer (Giannaros and Melas, 2012).

The primary sources of air pollution in the greater Thessaloniki area are road traffic, residential heating (Progiou *et al.*, 2023), biomass burning (Diapouli *et al.*, 2017), and industrial emissions (Moussiopoulos *et al.*, 2009). These activities have generally increased, leading to a decline in air quality (AQ) (Melas *et al.*, 2017), particularly during the economic crisis when unregulated fuel burning for domestic energy use was notorious (Zyrichidou *et al.*, 2019). In addition, particle pollution in the area is also substantially impacted by dust storms that originate in North Africa (Achilleos *et al.*, 2020; Psistaki *et al.*, 2022; Rizos *et al.*, 2022).



Figure 7.1. The study area with the locations of monitoring stations (1: Agias Sofias, 2: AUTh, 3: Panorama, 4: Kalamaria, 5: Kordelio, 6: Makedonia airport). Dashed lines represent municipal borders.

7.1.2 Data description

Hourly values of PM_{10} , O_3 , and NO_2 concentrations ($\mu g/m^3$) for the period 2006–2016 were obtained from five air quality monitoring stations covering the urban area of Thessaloniki, operated by the Ministry of the Environment and Energy (Table 7.1). The highest PM_{10} and NO_2 values and the maximum 8-hour moving average for O_3 over each station were used, representing the daily concentrations for the datasets.

Data regarding temperature and dew-point temperature on an hourly basis from 2006 to 2016 were acquired from the Makedonia Airport weather station, operated and certified by the Hellenic National Meteorological Service (Table 7.1).

The study focused on the Apparent Temperature, a thermal index that gauges the perceived thermal stress and comfort levels experienced by humans. Hourly Tapp was computed as follows:

 $Tapp = -2.653 + 0.994*Ta + 0.0153*Td^2$

where Ta is air temperature (°C) and Td the dew point temperature (°C). Subsequently, the daily maximum value (Tapp_{max}) was computed as the highest among the hourly values. Tapp is an optimal indicator for predicting all-cause mortality risk and for issuing heat alerts and warnings (Michelozzi *et al.*, 2010;

Lin *et al.*, 2012; Zhang *et al.*, 2014) and has been previously used to study mortality in Thessaloniki (Parliari *et al.*, 2022; Parliari, Giannaros, *et al.*, 2023).

Station	Description	Coordinates
1. Agias Sofias	Urban-traffic	40.63°N 22.94°E
2. AUTh	Urban-background	40.63°N 22.95°E
3. Panorama	Exurban-background	40.58°N 23.03°E
4. Kalamaria	Exurban-background	40.57°N 22.96°E
5. Kordelio	Urban-industrial	40.67°N 22.89°E
6. Makedonia Airport	Meteorological	40.52°N 22.97°E

Table 7.1. Description of the monitoring stations

Daily mortality data were obtained from the Hellenic Statistical Authority (ELSTAT), including age and cause of death for all Municipalities in the urban area of Thessaloniki for the study period (2006-2016). The data are stratified by causes of death: All-cause (natural, non-accidental), Cardiovascular and Respiratory, according to ICD-10 (International Classification of Diseases, 10th version). Emphasis was placed on studying the overall mortality rate among the elderly, specifically aged 65 and higher.

7.1.3 Statistical analysis

To quantify the association between the various exposures and attributed mortality we use the general framework of distributed lag models (DLMs). The DLM is a regression modelling framework that can capture the distributed effect of an exposure across temporal lags (Gasparrini, Armstrong and Kenward, 2010). Here, we implement DLMs as GAMs with objective penalization to guard against overfitting, while also allowing for a straightforward quantification of interactions across exposures.

Let M_t and AT_t denote the mortality count and Tappmax for day t. A DLM is defined as

$$M_t \sim NegBin(\mu_t, \theta)$$
(1)
$$\log(\mu_t) = \alpha + \beta_0 A T_t + \beta_1 A T_{t-1} + \dots + \beta_L A T_{t-L}$$
(2)

where L days is the maximum temporal lag. The use of the negative binomial (NegBin) distribution is a conventional choice in epidemiological analyses. It extends the Poisson distribution to allow for over-dispersion (extra variance). The linear "effects" β_l act multiplicatively on the mean mortality count μ_t (and

additively on $log(\mu_t)$. The framework can be extended to DLNMs to allow for non-linear effects by extending the formulation of μ_t via:

$$\log(\mu_t) = \alpha + h(0, AT_t) + h(1, AT_{t-1}) + \dots + h(L, AT_{t-L})$$
(3)

where $h(l, AT_{t-l})$ is an unknown 2D function of Tapp_{max} and lag *l*. Assuming that $h(l, AT_{t-l})$ is smooth, then this is a GAM which can be implemented in the R package mgcv (Wood, 2011, 2017) to optimally estimate $h(l, AT_{t-l})$ by penalising its flexibility. An example of an estimated $h(l, AT_{t-l})$ is shown in Figure 7.2.

Penalization in non-linear regression models is important to avoid over-fitting (over-explaining) the data, and in this case, to also reduce sensitivity to the choice of L (as long as this is large enough). More importantly, it is straightforward to define interactions of the Tappmax-lag effect with other exposures (e.g., with PM10), via

$$\log(\mu_t) = \alpha + h(0, AT_t, PM_{10,t}) + h(1, AT_{t-1}, PM_{10,t-1}) + \dots + h(L, AT_{t-L}, PM_{10,t-L}).$$
(4)

Function $h(\cdot)$ can be defined using tensor product interactions of regression splines (Wood, 2017), a robust and flexible approach to defining smooth functions across many dimensions. The models presented in what follows were implemented in the R package mgcv, which makes use of penalised maximum likelihood to estimate the smooth functions.

In *mgcv*, the "effects" (i.e., the smooth functions) are constructed such that they are centered at zero, i.e., their sum is zero. As such, the term $\exp \{\alpha\}$ is interpreted as the overall (across all time and exposure values) mean mortality count. Function $\rho(\cdot) = (\exp \{h(\cdot)\})$ is therefore the multiplicative increase or decrease of the mean mortality count. $\rho(\cdot) > 1$ implies greater than average mortality risk, whereas $\rho(\cdot) < 1$ means lower-than-average risk. Thus $\rho(\cdot)$ is the relative risk or RR (relative to the overall mean death count). Moreover, a quantity to summarise the "marginal" risk due to the exposure is based on the sum of $h(\cdot)$ over the lags. For instance,

$$CR(AT) = \exp\left\{\sum_{l=0}^{L} h(l, AT)\right\}$$
(5)

is the cumulative risk (CR) for Tappmax from model (3).

Moreover, GAMs implemented in mgcv can be interpreted in a Bayesian way (Wood, 2017) and Monte Carlo simulation can be used to quantify the associated uncertainty in estimating the smooth function $h(\cdot)$. This in turn enables the assessment of "significance" in the RR estimated, which here is determined by checking whether the value 1 lies within the 95% credible interval of $\rho(\cdot)$. If it is not, then we can say that with probability 0.95 the value of the RR is not 1 and that the weight of evidence supporting this is strong.

The surface that graphically explains the concept of the present work is demonstrated in Figure 7.2, where the cumulative risk plot results from summing across the lag dimension and along the pollutant and Tappmax ranges.



Figure 7.2. Representation of the tri-dimensional surface depicting the non-linear association between Tappmax and pollutant on mortality. Effect estimates of heat (75th and 99th percentiles of Tapp_{max} distribution) and cold (1st and 25th percentiles) are shown, along with sections corresponding to low, medium and high pollution values (5th, 50th and 95th percentiles).

7.1.4 Data analysis

Figure S1 shows the violations of EU Air Quality Guidelines for PM10, NO₂ (annual, 40 μ gr/m³) and O₃ (daily, 120 μ gr/m³) during study period 2006-2016. EU annual limits were surpassed every year for PM10 and 6 out of 11 years for NO₂, while 28% of total days (4018) recorded O₃ daily values above threshold.

We calculated the mean monthly values of the environmental stressors for Thessaloniki across the 11-year period, to get a picture of their seasonal variability. PM10 and NO₂ display peaks during winter months and troughs at

summertime. O_3 exhibits reverse seasonality and follows the pattern of Tapp_{max} behavior, with summertime maxima due to enhanced photochemical processes (Figure S2).

During the study period, there were 73,990 natural deaths from all-causes, 21,811 from cardiovascular (CVD) diseases, and 7,134 from respiratory (RD) causes. Mortality among the elderly was 62,482 representing more than 80% of the total number of deaths. Table 7.2 provides the descriptive statistics of pollution, Tapp_{max} and daily mortality for the reference period.

The mortality data indicates that 30% of all natural deaths are attributed to cardiovascular diseases, highlighting the significance of this health condition for investigating susceptibility, noting that stroke and ischaemic heart disease are the leading causes of death in Greece (European Observatory on Health Systems and Policies, 2019). In addition, Greece is among the countries with the highest population rates in the age group of over 65 in Europe (EUROSTAT, 2020); in our dataset elderly mortality reflects 84% of all-cause mortality. Pollutant concentrations are generally higher in Thessaloniki than those reported in Greece (Rai *et al.*, 2023) and other metropolitan areas (Garrett and Casimiro, 2011; Samoli *et al.*, 2014), comparable to values of other large cities with substantial air quality issues (Samoli *et al.*, 2009; Stafoggia *et al.*, 2010; Wang *et al.*, 2018). Similar values of mean daily mortality and summary statistics of PM10 in Thessaloniki are verified also in (Samoli *et al.*, 2013).

Daily Mortality (number of deaths)					
		Mean	St.	dev.	
All-cause		18.4	4.7		
Cardiovaso	cular	5.4	2.4	1	
Respirator	У	1.8	1.4		
Elderly		15.5	4.4		
		Тарр	D _{max} (°C)		
Mean	1 st perc.	25 th perc.	75th perc.	99 th perc.	
19	0.5	10	28	38	
		PM10	$(\mu g/m^3)$		
Mean	5 th pe	rc.	50 th perc.	95 th perc.	

Table 7.2. Descriptive statistics of mortality (number of deaths, top), Tapp_{max} (°C,middle) and pollution ($\mu gr/m^3$, bottom).

55	28	49	104
		O_3 (ug/m ³)	
Mean	5 th perc.	50 th perc.	95 th perc.
101	54	99	160
	1	NO ₂ (μ g/m ³)	
Mean	5 th perc.	50 th perc.	95 th perc.
40.3	17	37	74

7.2 Results

7.2.1. Application of GAM for thermal stress

A model defined by equations (1) and (3) is fitted to quantify the effect of Tapp_{max} along its lags with L= 21 days. The estimated RR surface is given in Figure 7.3 left panel, indicating elevated risk for high Tappmax at short lags (0-3). The surface is comparable to the one shown in (Parliari *et al.*, 2022), who studied the same data set using a DLNM from the *dlnm* package in R. Moreover, Figure 7.3 right panel shows the associated CR (cumulative risk as defined in equation 5) for Tappmax, which shows that risk grows exponentially above 33 °C.



Figure 7.3. Exposure-lag-response risk surface demonstrating the nonlinear association between Tapp_{max} and mortality (left panel); Cumulative exposure-response curve between daily Tapp_{max} and all-cause mortality over lag days 0-20 (right panel).

Using the estimated CR, we can quantify the percentage change in risk when Tapp_{max} changes. Specifically, we define as "heat effect" the percentage change in risk when Tapp_{max} increases from its 75th to its 99th percentile. Similarly, the "cold effect" is the change occurring from the 25th to the 1st percentile. A single model was fitted for each cause of death. The estimates of the heat and cold effects as defined above show that the impact from heat is considerable across all causes, with RD recording the largest increase (54.8%), while the age group of > 65 years exhibits the smallest (28.4%) increase. Lower (but still positive) changes are evident for the cold effects, with RD being the health condition with the most pronounced effect (30.2%), roughly three times larger than the others (Figure S3).

Generally, Thessaloniki's population is more severely impacted by high temperatures than low ones with respect to all types of mortality. For comparison, Rai et al. (2023) reported a 17% increase in CVD and 19% increase in RD mortality between the 75th and the 99th percentile of temperature distribution, pooled from five cities in Greece. Concerning lag effects, there is an immediate increase in RR upon exposure to high temperatures (lag 0-3), whereas for low temperatures the effect is delayed (lag > 3). This finding is in agreement with previous studies concluding that heat has an almost immediate effect on human health, while impacts from cold take longer to manifest (Guo *et al.*, 2011; Burkart *et al.*, 2013a; Bao *et al.*, 2016; Parliari *et al.*, 2022).

7.2.2 Synergy between PM10 and Tapp_{max}



Figure 7.4. Combined effect of PM10 and Tapp_{max} on all-cause mortality stratified by levels of pollutant (defined as "low", "medium", and "high" based on 5th, 50th, and 95th percentiles of PM10 distribution).

A model defined by equations (1) and (4) is implemented to study the joint effects of Tapp_{max} and PM10. The estimated RR surfaces $h(l, AT, PM_{10})$ for the 5th, 50th and 95th percentiles of PM10 (29, 49, 104 µg/m³) are given in Figure 7.4. Clearly,

the most harmful combination of outdoor conditions for human health is high Tapp_{max} and high pollution levels (RR up to 1.28), while at low and medium levels of pollution, risk is lower (RR up to 1.18 and 1.23, respectively).



Figure 7.5. Cumulative exposure-response association across lags 0-6 for different levels of PM10 (blue denotes "low", yellow denotes "medium", and red denotes "high").

Figure 7.5 shows the corresponding CR plots for each PM10 level. For Tapp_{max} < 7 °C, low PM₁₀ levels result in higher mortality risk than medium and high pollutant concentrations. In the region [7 °C, 20 °C], the effects of low and medium pollution levels decrease with temperature, while at high levels of pollution there is a ceiling of impact at 13 °C, above which, risk decreases. This could be related to data availability in the region, rather than physical behavior of the system. In warmer conditions below 33 °C, all RR values are lower than 1, indicating that these conditions are favorable. In extremely hot conditions (Tapp_{max} > 33 °C), there is a sharp increase in mortality risk, which becomes more marked with increasing PM₁₀ levels. The highest RR value of around 2 occurs when air quality is at its poorest under conditions of extreme thermal discomfort of 43°C. These conditions, high Tapp_{max} values and high airborne PM levels, are quite frequent in Thessaloniki.

7.2.2.1 Weight of evidence

As discussed in section 7.1.3, uncertainty in the estimated risk can be quantified in a Bayesian manner and produce probabilistic statements regarding the weight-ofevidence (or statistical significance) behind the estimates. We produce the 3D RR surface plots as 2D raster images of RR in Figure S4. Grid points for which RR=1 is outside the 95% credible interval are marked with grey shading (circles). Grey regions indicate statistically significant areas where the RR is either greater (red) or smaller (blue) than 1 (the mean mortality count).

In general, many of the regions (particularly for heat effects at Tappmax > 20 °C) are statistically significant, and therefore the conclusions drawn are robust. However, the areas below 30 °C for low and medium PM10 levels and the area 5 °C–30 °C for the high level at lags 0–3 define regions where the RR is not different from 1. This could either be because there are very few Tappmax-PM10 values there, or because the association between mortality and stressors is weak at short lags.

To further assess the robustness of the findings, sensitivity analyses were performed to investigate confounding effects with other pollutants. Specifically, we fitted a model where data points corresponding to high (> 90th percentile) NO2 values were excluded (Figure S4). This resulted in extended grey regions suggesting some degree of synergy between high PM10 and high NO2 concentrations. The same procedure is repeated for O3 values and again, the significance increased, implying confounding effects among the three pollutants at high concentrations. Nevertheless, the differences were not pronounced enough to justify using the cropped dataset for the rest of the analysis.

For the corresponding CR plots, uncertainty is presented via 95% intervals (dashed lines) in Figure 7.6. Risk is "significant", where the value of 1 is not included in the intervals. There is little confidence for Tappmax < 0 °C at low and medium PM10 levels and below 8 °C at a high PM10 level. A possible explanation might be the relatively lower exposure of the population to outdoor conditions during winter. The wider breadth of the intervals at extremely low and extremely high Tappmax is due to the very small sample size of temperature values in those regions.



Figure 7.6. Cumulative exposure–response associations across lags 0-6 for Low, Medium and High PM10 levels. Dashed lines represent 95% CI.

7.2.3 Synergy between NO₂ and Tapp_{max}

To study the NO₂ effect, the same model as in section 7.2.2 was implemented, but with NO₂ rather than PM10. Similar to PM₁₀, 3D plots in Figure S5 reveal consistent synergy between Tapp_{max} and NO₂ concentrations in all pollution ranges (low, medium and high defined at 17, 37 and 74 μ g/m³). With increasing temperature, mortality risk increases, exhibiting maxima at short lags (0–2). For all three levels, a second peak is present below 10 °C for longer lags. An increased mortality risk of about 20% is evident at the high NO₂ level for Tapp_{max} > 40 °C, confirming that intense thermal stress and poor air quality pose major threats to human health.

The pattern of cumulative RR for NO₂ over the entire lag period (0-6) for the three levels of pollution is similar to PM₁₀, highlighting the similarity between the two pollutants (Figure S6).

The peaks are statistically significant as indicated by the grey shading areas (Figure S7), and this remained largely unchanged when investigating the possible confounding of NO₂ with high (>90th percentile) values of PM₁₀ and O₃ (results not shown). The top right region (high Tapp_{max} and long lags) remains insignificant, indicating that those values may not be supported by adequate data or a strong relationship between exposure and mortality. Finally, there is high statistical significance below 0 °C (at low and medium NO₂ levels) and below 8 °C (at a high NO₂ level) (Figure S8).

7.2.4 Synergy between O₃ and Tapp_{max}

The analysis for O_3 is analogous to that of PM_{10} and NO_2 . The three pollution levels in this case are defined as 54 µg/m³ (low), 99 µg/m³ (medium), and 160 µg/m³ (high pollution). The highest mortality risk occurs for high Tapp_{max} over short lags (0–2), with a second peak appearing at 10 °C, for all pollution levels over longer lags. O_3 concentrations of 160 µg/m³ are associated with the most adverse human health impacts at Tapp_{max} > 40 °C (Figure S9).

Cumulative risk is slightly elevated for Tapp_{max} between 0 °C and 10 °C in conditions of medium O₃ pollution. At low and high levels of ozone, there are insignificant relationships for specific segments of temperature (Tapp_{max} < 8 °C and Tapp_{max} > 15 °C for low, and Tapp_{max} >17 °C, for high ozone concentrations).

The weight of evidence for high O_3 concentrations and low Tapp_{max} is low (Figure S11). Indeed, no such combination exists in our dataset (i.e., the estimates are statistical extrapolations) given the opposite seasonal behavior of these components. An additional hiatus is found at low O_3 levels for Tapp_{max} between 10 °C and 20 °C. After omitting high PM₁₀ and NO₂ values, the 95% CI did not increase: therefore, we assume that confounding does not occur.

7.2.5 Lag structure analysis

In order to better understand the temporal dimension of the complex exposurelag-response association, an appropriate representation of the temporal pattern of risk is needed. In what follows, we cumulate the Tapp_{max}-Pollutant-Lag surface across the Tapp_{max} space (i.e., sum the RR at the log scale) to "integrate out" Tapp_{max}. We then present "slices" of the resulting Pollutant-Lag surface for different lags in Figure 7.7.

The top left panel of the plot relates to PM10 where it is evident that the highest mortality risk is on the day of exposure (lag 0), and this diminishes as the lag increases. This behavior shows that there is no threshold in the PM10 dose–response relationship, and even a small increase in this pollutant has health impacts, in agreement with previous studies (Psistaki *et al.*, 2022; Parliari, Giannaros, *et al.*, 2023).

In contrast, the impact of NO2 on mortality (right panel) is more pronounced several days after the exposure and reaches a maximum on the last day (lag 6). This indicates that the health impacts of NO2 may be cumulative.

The temporal pattern of O3 (bottom left) is different from PM10 and NO2. Up to a certain threshold (~ 170 µg/m3), the risk increases with lag. On lag 8 the risk is greater than on lag 7 and the smallest RR values occur at lag 0. For very high O3 concentrations (> 170 µg/m3) this pattern is reversed: the effect on mortality is immediate (lag 0). Interestingly, this convergence point for O3 concentrations is partly evident in other studies focusing on the impact of O3 on human health (Parliari, Giannaros, *et al.*, 2023), showing that the lag structure is different depending on the pollutant concentration under study.





Figure 7.7. Dose-response curves of cumulative RR across $Tapp_{max}$ range, demonstrating the lag structure for PM10 (top left, lags 0-6), NO₂ (top right, lags 0-6) and O₃ (bottom left, lags 0-8).

We then looked at the lag structure of RR as a function of Tapp_{max}, by cumulating the risk across the pollutants. Figure 7.8 shows the associated curves relating to PM10 (top left), NO₂ (top right) and O₃ (bottom left). In all cases, the following temporal structure of the exposure-response relationship is evident: heat effects are immediate while the ones from cold become predominant at longer time lags. These results are in line with previous studies (Liu *et al.*, 2011; Lubczyńska, Christophi and Lelieveld, 2015; Schnell and Prather, 2017; Rodrigues, Santana and Rocha, 2019).

For PM10, the RR for Tapp_{max} < 15 °C is higher 7 days after the exposure (lag 6) than present day (lag 0). Conversely, the RR for Tapp_{max} > 15 °C increases exponentially and reaches its peak at lag 0. The same applies for NO₂ beyond the convergence point of 31°C, and O₃ at 27°C.





Figure 7.8. Dose-response curves of cumulative RR across pollutants concentration ranges, demonstrating the lag structure for PM10 (top left), NO_2 (top right) and O_3 (bottom left).

7.2.6 Heat effect by pollutant levels (all- and specific-cause mortality)

Heat effects on all-cause mortality stratified by pollutant levels are shown in Figure S13. The exposure of the population to outdoor conditions is higher in summer than in winter; therefore, the percentage change in mortality for all pollutants is greater for heat compared to cold. Figure 7.9 shows that mortality rises with increasing concentrations of PM_{10} , NO₂, and O₃. Overall, PM_{10} accounts for the largest health burden, expanding to 47.7% mortality increases for high levels, compared with 37.7% for NO₂ and 32% for O₃. Similarly, in the cold part of the range, the highest pollutant concentrations exert the greatest impact on mortality, but the respective increases are significantly lower (Figure S14).

Figure 7. 9 shows the heat effects on mortality across the various causes for the whole population and the elderly (> 65 years) – stratified by levels of PM_{10} , NO_2 , and O_3 . A consistent rise in the impact of heat on mortality due to all causes and CVD diseases is observed. The highest percentages occur at high pollutant levels. All-cause mortality is primarily affected by all levels of PM_{10} (20%, 27.5%, and 47.7%, respectively), secondly by NO_2 and lastly, by O_3 . A very large effect of heat on CVD mortality is found at the high level of PM_{10} (58.9%), followed by O_3 (45.3%). Interestingly, CVD mortality is not affected by varying levels of NO_2 and its levels remained high (37.2%). It is worth noting the respiratory (RD) decrease with increasing PM_{10} and NO_2 levels.

The elderly population is primarily affected by PM_{10} at all levels, followed by NO_2 , and O_3 . For all pollutants, the elderly estimates qualitatively follow the ones



for all-cause mortality for the respective species, but with much higher values compared to the general population.

Figure 7.9. Estimates of heat effects as % change in cause-specific mortality for Tapp_{max} increases between the 75th and 99th percentile by levels of pollutants (defined as "low", "medium", and "high" based on 5th, 50th, and 95th percentiles of pollutants distributions).

7.2.7 Attributable Mortality Fraction (all- and specific-cause mortality) for non-optimum Tapp_{max} ranges

The approach presented in this paper is extended to encompass the mortality burden (Attributable Fraction, AF). In contrast to RR, which is evaluated at potentially counterfactual situations and which does not allow for the likelihood of stressor values, AF is a more rigorous metric which quantifies the effect on the population, taking into consideration the observed number of deaths and also the frequency of extreme Tapp_{max} occurrence (Kim, Kim and Liu, 2014; de' Donato *et al.*, 2015).

Like RR, the AF is computed using the estimated functions (e.g., h(l, AT)) but using the actual Tapp_{max} values in the data to attribute the number of deaths on any given day to the distributed effect of Tapp_{max}.

Figure 7.10 shows the AF for non-optimum $Tapp_{max}$ (extreme cold, mild cold, mild heat, and extreme heat) stratified by pollutant level as follows:

- (a) Extreme cold: minimum to 5th percentile
- (b) Mild cold: 5th percentile to MMT
- (c) Mild heat: MMT to 95th percentile
- (d) Extreme heat: 95th percentile to maximum

It is apparent that in the vast majority of cases, extreme heat has the biggest influence on the AF. Two exceptions are noted, specifically at low levels of PM_{10} and NO_2 , where extreme cold is more prominent. Very frequently, extreme cold is the second most harmful temperature range.

Although in articles studying the impact of temperature on mortality, optimum temperatures (mild cold, mild heat) are associated with most of the attributable deaths (Parliari *et al.*, 2022; Psistaki, Dokas and Paschalidou, 2023), in our synergistic analysis, extreme heat results not only in the highest RR values, but highest AF as well, compared to other optimum and non-optimum ranges. This confirms our previous inference that extremely high Tapp_{max} values are most harmful to health when combined with increased levels of pollutants.

 PM_{10} has the largest mortality burden among the three pollutants at the medium and high levels, while NO₂ is mostly prominent when levels of pollution are low. It is also worth noting that the AF decreases (overall) from medium to high O₃ levels, probably due to the limited exposure of the population to extremely high temperatures in the summer months.





Figure 7.10. Fractions of all-cause mortality attributable to non–optimum Tapp_{max} by levels of pollutants (defined as "low", "medium", and "high" based on 5th, 50th, and 95th percentiles of pollutants distributions).

Stratifying AF by the various diseases confirms that heat impact on cause-specific mortality burden is strongly modified by pollutant levels (Figure 7.11). Mortality fraction from all- and respiratory causes is primarily affected by PM10 levels and then by NO₂. AF related to CVD mortality, although increasing with rising O₃ concentrations, decreases for PM10 and NO₂.

The estimates of the AF for the Elderly follow the all-cause distribution for all pollutants but are exhibit much higher numbers compared to the general population. Effect modification is higher for PM10, followed by NO_2 and O_3 .





Figure 7.11. Fractions of cause-specific mortality attributable to levels of pollutants (defined as "low", "medium", and "high" based on 5th, 50th, and 95th percentiles of pollutants distributions).

7.3. Discussion

To our knowledge, this study is the first to use a GAM-based approach to assess the synergistic impacts of thermal conditions and air quality for the urban area of Thessaloniki.

We used an advanced statistical approach that captures the complex non-linear and lagged dependencies in both the exposure-response and lag-response associations. This approach is flexible and can be adjusted to account for other synergies as well. Our results are statistically significant based on high-quality data spanning a period of 11 years.

The research holds significant potential for applications in various fields such as atmospheric sciences, biometeorology, and environmental epidemiology as our findings can shed light on the impacts of climate and weather on human health. This interdisciplinary significance can inform evidence-based policies, adaptation strategies, and targeted public health interventions in the vulnerable Southern European region facing climate change challenges.

We provide compelling evidence for the compounded effects of air pollution (PM_{10} , NO_2 , and O_3) and the maximum apparent temperature, with particularly pronounced adverse impacts on the elderly. We show that deterioration in air quality renders the population more susceptible to the effects of temperature variability, and *vice versa* that extreme temperatures influence susceptibility to air pollution. We also examine the effects of temporal lag and non-optimum Tapp_{max}, as very little is known about the relative contribution of both heat and cold from moderate and extreme temperatures to the disease burden. In this study, we identified Tapp_{max} as the most appropriate thermal predictor, based on previous

investigations of the dataset (Parliari et al., 2022; Parliari, Giannaros, et al., 2023).

Relationships between deaths and temperature are frequently expressed in the literature with the use of V, U, J, and reverse J-shaped structures (Baccini *et al.*, 2008; Zhang *et al.*, 2016; Psistaki, Dokas and Paschalidou, 2022). Our analysis produced J-shaped Tapp_{max}-mortality curves stratified by air pollution levels, with RR values increasing sharply over 33 °C. This specific threshold is demonstrated in Kouis et al. (2019) for Thessaloniki, indicating that it is typical. We found that low temperatures are associated with increased mortality as well, but not to the same extent.

The exposure–response associations between $Tapp_{max}$ and RR cumulated over all lags shows consistent patterns for the three studied pollutants:

- Winter temperatures (Tapp_{max} < 10 °C), probably affected by limited exposure to outdoor conditions and confounded by the transmission of diseases such as influenza, increase the mortality risk up to 1.18, 1.17, and 1.15 for PM₁₀, NO₂, and O₃, respectively. As temperatures rise (10–20 °C), people spend more time outdoors or increasingly ventilate with ambient air, increasing exposure and the RR.
- In view of excess deaths, a prominent "harvesting effect" or mortality displacement (RR < 1) is evident between 20 °C and 33 °C. This indicates the existence of subgroups of vulnerable individuals, for whom exposure to cold and high pollution levels leads to a subsequent decrease in deaths, as reported also in Breitner et al. (2014). It is worth mentioning that this Tapp_{max} range is in force during months when dust events are more frequent in Thessaloniki, with recommendations to avoid outdoor activity, which in turn may safeguard the local population (Psistaki *et al.*, 2022). Another explanation of the RR < 1 area of the graph (particularly at around 25 °C) could be the "September phenomenon" (Falagas *et al.*, 2009), which proposes reduced numbers of deaths in the late summer to early fall months in the Mediterranean countries due to human behavioral patterns, socio-economic factors, and environmental parameters.
- From 33 °C upwards, cumulative mortality risk rises exponentially for all pollutants. The most adverse impact is reported for the combination of very high Tapp_{max} (> 40 °C) and high pollution levels (RR = 1.95 for PM₁₀, 2.0 for NO₂, and 1.82 for O₃). The same result arises from the three-fold lag–exposure–response surfaces, where the death risk reaches +26%, 20% and +15% for PM₁₀, NO₂ and O₃, respectively, at lag 0. The findings of Chen et al. (2018) support the idea that high air pollution enhances temperature effects on daily mortality, while associations between high temperatures and mortality seem to be generally stronger at high pollution levels. Significant heat impact (expressed by UTCI) was evident for the medium-high and high O₃ categories of Lisbon and Berlin and high PM₁₀

pollution in Lisbon (Burkart *et al.*, 2013b). Breitner et al. (2014) also showed that correlations between high temperatures and mortality were most pronounced at elevated levels of PM_{10} and O_3 .

Generally, heat effects are found to be more harmful than cold in this study. High levels of PM_{10} , NO_2 , and O_3 are associated with increases in mortality of 47.7%, 38.7%, and 32%, respectively, whereas the respective increases for cold are 13.6%, 19.2%, and 2.1%. Similar results are reported in Burkart et al. (2013a), noting that the cold impact was mostly unaffected by air pollution.

We found that in Thessaloniki, EU air pollution limits are regularly exceeded for the studied pollutants. By reducing the concentration of these harmful substances, especially during heat waves when vulnerable populations are at a greater risk, it will be possible to significantly reduce the numbers of fatalities caused by various ailments. Strict compliance with EU limits represents a proactive approach to improving public health and mitigating the adverse effects of extreme heat events.

In view of the mortality burden, we show that extremely high temperatures (Tapp_{max} > 35 °C) are responsible for the majority of deaths at all three levels of PM₁₀, NO₂, and O₃. For all pollutants, extreme heat, extreme cold, mild cold, and mild heat are responsible for 60.7%, 48%, 45% and 11% of attributable mortality, respectively. These findings confirm our previous results regarding mortality risk, indicating that non-optimum Tapp_{max} values in Thessaloniki (hot more so than cold) are responsible for the majority of the detrimental impacts on health when coincident with enhanced concentrations of ambient air pollution.

The assessment of percentage increases in cause-specific deaths across pollution levels shows that all-cause mortality is strongly affected by enhanced PM_{10} levels (20%, 27.5%, 47.7%), whereas both PM_{10} (31.4%, 37.5%, 58.9%) and O₃ (33.5%, 41.2%, 45.3%) affect cardiovascular disease (CVD). Interestingly, the death rate from respiratory disease (RD) marginally decreases as NO_2 and PM_{10} concentrations increase, possibly due to behavioral patterns of the population. Nevertheless, consistently higher increases in RD mortality were found for all pollutants compared to all-cause and CVD, also evident in Rai et al. (2023). Lastly, in our study, CVD has a higher death toll than all causes, also demonstrated by Chen et al. (2018).

Our analysis shows that elderly citizens (65 years and older) are particularly vulnerable to the adverse combination of thermal conditions and poor air quality. Many studies report that older people are susceptible to extreme meteorological conditions and air pollution (Barnett, 2007; Cakmak, Dales and Vidal, 2007; Oudin Åström, Bertil and Joacim, 2011; Gasparrini *et al.*, 2012; Shim, Byun and Lee, 2023), due to the prevalence of pre-existing chronic conditions and because they may be subject to physiological changes in thermoregulation and homeostasis. In addition, the limited care and social support for this age category may also render them more susceptible (Breitner *et al.*, 2014).

The lag structure analysis across the Tapp_{max} range revealed varying influence of the three pollutants: PM_{10} showed the highest mortality risk at lag 0 with no threshold in PM_{10} dose-response relationship, as in previous studies (Psistaki *et al.*, 2022; Parliari, Giannaros, *et al.*, 2023). However, for NO₂ the health impacts were delayed by several days. O₃ had an immediate effect at high concentrations, even at day 0, while below 170 µg/m³ the situation was reversed. This "threshold" in O₃ emerges in other studies as well (Parliari, Giannaros, *et al.*, 2023). Further, the dose–response relationships of cumulative RR across pollutant concentrations showed that the impacts of high temperatures are immediate (days 0–3), whereas the cold effects became more pronounced with longer time lags (lag > 3). In general, heat effects tend to occur on a short-term, typically a few days, whereas cold effects evolve and persist over a longer duration, ranging from several days to weeks (Burkart *et al.*, 2013b; Breitner *et al.*, 2014; K. Chen *et al.*, 2018).

While the biological mechanisms responsible for the combined impact of air pollution and thermal conditions on mortality are not well known, several hypotheses have been put forward. Hot days induce physiological stress which may modify the response to air pollution, potentially increasing the susceptibility to adverse effects. Similarly, air pollution can render individuals more susceptible to thermal stress, as exposure impacts heart rate variability, the autonomic nervous system, and inflammatory parameters, for example (Gordon, 2003).

The limitations of this study include the following. Given the location-specific character of our and similar studies (Samoli *et al.*, 2014; Ma, Zhou and Chen, 2020), generalizing the findings and directly comparing with assessments of health impacts in other regions with possibly distinct basic health and air pollution conditions may be problematic. Furthermore, using air quality data from fixed stations instead of exposure data may result in bias from exposure misclassification, as individual exposure does not necessarily coincide with measured atmospheric parameters.

Our findings emphasize the need to mitigate the health burdens associated with ambient air pollution and thermal conditions. Additional research is necessary to explore air pollution and air temperature inter-relationships using morbidity data, such as hospitalization and emergency room visits, in order to gain a more comprehensive understanding of the interaction between thermal stress and air pollution.

7.4. Conclusions

We investigated the interactive effects of meteorological conditions (i.e., thermal conditions) and air pollution, using PM_{10} , NO_2 , and O_3 as predictors. We found that air pollution strongly modifies thermal effects, especially heat, and *vice versa*, in varying degrees across different lags. Our results underscore that low

temperatures have delayed effects on mortality, whereas high temperatures have immediate effects. We studied the compounded effects of air pollutants and the maximum apparent temperature and found that mortality strongly increases at high temperatures and high levels of air pollution. The elderly (65 years and older) are particularly vulnerable to heat stress and impaired air quality. These findings have important implications for the development of public health interventions aimed at controlling and preventing health consequences of exposure to extreme temperatures and poor air.



7.5. Supplementary material







Figure S2. Seasonal distribution of PM_{10} (top left), NO_2 (top right), O_3 (bottom left) and $Tapp_{max}$ (bottom right).



Figure S3. Estimates of heat and cold effects on cause-specific mortality and elderly mortality.



Figure S4. Contour plot of the joint association of PM_{10} and $Tapp_{max}$. Statistical significance (95% CI) is indicated by grey points.



Figure S5. Combined effect of NO_2 and $Tapp_{max}$ on all-cause mortality stratified by levels of pollutant (defined as "low", "medium", and "high" based on 5th, 50th, and 95th percentiles of NO_2 distribution).



Figure S6. Cumulative exposure–response association across lags 0-6, the temperature range and different levels of NO₂ (blue denotes "low", yellow denotes "medium", and red denotes "high" levels).


Figure S7. Contour plot of the joint association of NO₂ and Tapp_{max}. Statistical significance (95% CI) is indicated by grey points.



Figure S8. Cumulative exposure–response associations across lags 0-6 for Low, Medium and High NO₂ levels. Dashed lines represent 95% CI.







Figure S10. Cumulative exposure–response association across lags 0-8, the temperature range and different levels of O₃ (blue denotes "low", yellow denotes "medium", and red denotes "high" levels).



Figure S11. Contour plot of the joint association of O_3 and Tapp_{max}. Statistical significance (95% CI) is indicated by grey points.



Figure S12. Cumulative exposure–response associations across lags 0-8 for Low, Medium and High O₃ levels. Dashed lines represent 95% CI.



Figure S13. Estimates of heat effects as % change in all-cause mortality for Tapp_{max} increases between the 75th and 99th percentile by levels of pollutants (defined as "low", "medium", and "high" based on 5th, 50th, and 95th percentiles of pollutants distributions).



Figure S14. Estimates of cold effects as % change in all-cause mortality for Tapp_{max} increases between the 1st and 25th percentile by levels of pollutants (defined as "low", "medium", and "high" based on 5th, 50th, and 95th percentiles of pollutants distributions).

Chapter 8

Conclusions

Although the health effects of air pollution and heat stress pose a growing concern in today's world, there has been relatively limited investigation into the simultaneous interactive effects of temperature and air pollution as predictors of mortality. Thessaloniki, the second largest city in Greece, faces unfavorable air quality and meteorological conditions, exacerbating the relating health risks for the local citizens. The impact of climate change in the area is expected to further compound these challenges, increasing the frequency and intensity of heatwaves and pollution episodes. Unfortunately, the understanding of Thessaloniki's specific vulnerability to these issues remains understudied, hindering effective mitigation strategies. The current dissertation has introduced a Distributed Lag Non-linear Model using the framework of Generalized Additive Models to investigate the interactive effects between daily maximum Apparent Temperature and air pollution in cause-specific mortality from 2006 to 2016, in the urban area of Thessaloniki. This is the first application of the proposed GAM-based approach and the first analysis to explore this complex association for Thessaloniki, Greece.

The first study (Chapter 5) investigated the association between daily maximum Apparent Temperature and daily all-cause (natural, non-accidental), cardiovascular, cerebrovascular and respiratory mortality, by describing the exposure-lag-response association with the use of a DLNM. The effect of time lag and non-optimum temperatures, as well as specific analysis of the Elderly (+65 years), were also examined.

This study concluded that heat-attributable mortality in Thessaloniki was mainly associated with high temperatures. Significant implications that derived from this work include:

- ✓ Extreme high temperatures strongly influenced the risk in mortality with strong exponential rise over 35°C, with further and more prominent increase with Tappmax values over 40 °C.
- ✓ J-shaped relationships were found between temperature and cause-specific mortalities. Heat was short lived with an immediate effect, whereas cold was prolonged.
- \checkmark The Elderly were more vulnerable to heat than general population.
- ✓ Deaths attributed to heat outnumbered deaths attributed to cold, with stronger impacts on respiratory mortality.

The second study (Chapter 6) investigated the associations between short-term exposure to PM10 and O_3 and daily all-cause (natural, non-accidental), cardiorespiratory, and cerebrovascular mortality on the general population and the Elderly. To assist air quality planning, the impact of the modification of PM10 levels on Thessaloniki's population mortality under two air pollution abatement scenarios was estimated: (1) full compliance to EU levels, thus eliminating the exceedances of PM10 daily values; and (2) a 20% horizontal reduction in the PM10 concentration.

The key conclusions of this study are summarized as follows:

- ✓ Local population was at risk from the current levels of PM10 and ozone. O₃ was found to have an even more severe impact than PM10, and the elderly were particularly vulnerable to poor air quality in the area.
- ✓ Mortality risk and mortality burden of O₃ were more acute for cardiorespiratory mortality than all-cause; the opposite result was evident concerning PM10.
- \checkmark Neither of the two air pollutants was associated with cerebrovascular outcomes.
- ✓ If the two proposed mitigation measures were implemented, the attributed mortality fraction would decrease significantly in both cases.

The objective of the final study (Chapter 7) was to analyze the interactive effects between daily maximum Apparent Temperature and air pollution (NO₂, O₃, PM10) in cause-specific mortality (non-accidental, cardiovascular, respiratory), by developing a Distributed Lag Non-linear Model using the framework of Generalized Additive Models. Separate analysis was conducted for the Elderly citizens, as well as evaluation of the lag structure. The synergy was estimated by introducing a tensor product between Tappmax and either PM10 (lag 0–6), NO₂ (lag 0–6) or O₃ (lag 0–8). Temperature estimates were extrapolated at low, medium, and high levels of pollutants defined as the 5th, 50th, and 95th percentile of pollutant-specific distribution; heat and cold effects were estimated as the percentage change in mortality between the 75th and 99th, and the 25th and 1st percentiles of Tappmax, respectively. This study found that deteriorated air quality rendered the local population more susceptible to the effects of temperature variability, and vice versa.

The following conclusions were drawn:

- \checkmark Heat effect on mortality was more harmful than cold, in this study.
- ✓ Tappmax-mortality relationships, stratified by air pollution levels, demonstrated J-shaped curves with RR values increasing sharply over 33°C. Low temperatures were associated with increased mortality as well, but not to the same extent.
- ✓ High air pollution enhanced temperature effects on daily mortality, while associations between high temperatures and mortality were generally stronger at high pollution levels. Moreover, impacts of high temperatures were immediate, whereas the prevalence of cold effects became more pronounced as longer time lags were considered.
- ✓ Non-optimum Tappmax values (Extreme High Tappmax > 35 °C, in particular) had the greatest mortality burden on health when coupled with increased concentrations of pollutants.
- ✓ All-cause and respiratory mortalities were heavily affected by PM10 levels, whereas PM10, NO₂ and O₃ affected cardiovascular causes of death.
- ✓ Based on this analysis, the elderly citizens were particularly vulnerable to the adverse combination of thermal conditions and poor air quality.
- ✓ PM10 depicted immediate mortality risk increase, while NO₂ needed more days to exhibit the negative impact on human health. O₃ had prompt effect for large concentrations, and prolonged for decreased values.

To our knowledge, this study was the first application of the proposed GAMbased approach to assess the synergistic impact of thermal conditions and air quality for the urban area of Thessaloniki, by using an advanced statistical approach that captured the complex non-linear and lagged dependencies in both the exposure-response and lag-response associations, and flexible enough to be modified for other synergies. Given the vulnerability of the Southern European region, our work contributed significantly to various research fields such as atmospheric sciences, biometeorology and environmental epidemiology. 11 years of good quality data were available, resulting in satisfactory statistical significance of the produced results.

To sum up, the findings of this research made a substantial impact on the scientific efforts of better understanding the synergies with which environmental factors effect human health. They also concluded to a robust statistical technique that can be implemented to other epidemiological studies. Thus, the principal objective of this dissertation has been met with a great deal of success providing evidence of the combined impact of air quality and biometeorological factors on

human mortality. The present results carry important implications for the development of public health interventions aimed at controlling and preventing the health consequences of exposure to adverse thermal and air quality conditions.

References

Achilleos, S. *et al.* (2020) 'Spatio-temporal variability of desert dust storms in Eastern Mediterranean (Crete, Cyprus, Israel) between 2006 and 2017 using a uniform methodology', *Science of the Total Environment*, 714(October 2019), p. 136693. doi: 10.1016/j.scitotenv.2020.136693.

Almon, S. (1965) 'The Distributed Lag Between Capital Appropriations and Expenditures', *Econometrica*, 33(1), pp. 178–196.

Analitis, A. *et al.* (2006) 'Short-term effects of ambient particles on cardiovascular and respiratory mortality', *Epidemiology*, 17(2), pp. 230–233. doi: 10.1097/01.ede.0000199439.57655.6b.

Analitis, A. *et al.* (2008) 'Effects of cold weather on mortality: Results from 15 European cities within the PHEWE project', *American Journal of Epidemiology*, 168(12), pp. 1397–1408. doi: 10.1093/aje/kwn266.

Anderson, B. G. and Bell, M. L. (2009) 'Weather-Related Mortality', *Epidemiology*, 20(2), pp. 205–213. doi: 10.1097/ede.0b013e318190ee08.

Armstrong, B. (2006) 'Models for the relationship between ambient temperature and daily mortality', *Epidemiology*, 17(6), pp. 624–631. doi: 10.1097/01.ede.0000239732.50999.8f.

Armstrong, B. G. *et al.* (2011) 'Association of mortality with high temperatures in a temperate climate: England and Wales', *Journal of Epidemiology and Community Health*, 65(4), pp. 340–345. doi: 10.1136/jech.2009.093161.

Baccini, M. *et al.* (2008) 'Heat effects on mortality in 15 European cities', *Epidemiology*, 19(5), pp. 711–719. doi: 10.1097/EDE.0b013e318176bfcd.

Balbus, J. M. and Malina, C. (2009) 'Identifying vulnerable subpopulations for climate change health effects in the United States', *Journal of Occupational and Environmental Medicine*, 51(1), pp. 33–37. doi: 10.1097/JOM.0b013e318193e12e.

Ballester, F. *et al.* (2002) 'The EMECAM project: A multicentre study on air pollution and mortality in Spain: Combined results for particulates and for sulfur dioxide', *Occupational and Environmental Medicine*, 59(5), pp. 300–308. doi: 10.1136/oem.59.5.300.

Bao, J. *et al.* (2016) 'The influence of temperature on mortality and its Lag effect: A study in four Chinese cities with different latitudes', *BMC Public Health*, 16(1), pp. 1–8. doi: 10.1186/s12889-016-3031-z.

Barnett, A. G. (2007) 'Temperature and cardiovascular deaths in the US elderly: Changes over time', *Epidemiology*, 18(3), pp. 369–372. doi: 10.1097/01.ede.0000257515.34445.a0.

Barnett, A. G., Tong, S. and Clements, A. C. A. (2010) 'What measure of temperature is the best predictor of mortality?', *Environmental Research*, 110(6), pp. 604–611. doi: 10.1016/j.envres.2010.05.006.

Basu, R. (2009) 'High ambient temperature and mortality: A review of epidemiologic studies from 2001 to 2008', *Environmental Health: A Global Access Science Source*, 8(1). doi: 10.1186/1476-069X-8-40.

Basu, R., Dominici, F. and Samet, J. M. (2005) 'Temperature and Mortality Among the Elderly in the United States', *Epidemiology*, 16(1), pp. 58–66. doi: 10.1097/01.ede.0000147117.88386.fe.

Basu, R., Feng, W. Y. and Ostro, B. D. (2008) 'Characterizing temperature and mortality in nine California counties', *Epidemiology*, 19(1), pp. 138–145. doi: 10.1097/EDE.0b013e31815c1da7.

Basu, R., Malig, B. and Ostro, B. (2010) 'High ambient temperature and the risk of preterm delivery', *American Journal of Epidemiology*, 172(10), pp. 1108–1117. doi: 10.1093/aje/kwq170.

Basu, R. and Samet, J. M. (2002a) 'An exposure assessment study of ambient heat exposure in an elderly population in Baltimore, Maryland', *Environmental Health Perspectives*, 110(12), pp. 1219–1224. doi: 10.1289/ehp.021101219.

Basu, R. and Samet, J. M. (2002b) 'Relation between elevated ambient temperature and mortality: A review of the epidemiologic evidence', *Epidemiologic Reviews*, 24(2), pp. 190–202. doi: 10.1093/epirev/mxf007.

Biggeri, A. *et al.* (2004) 'Meta-analysis of the Italian studies of short-term effects of air pollution (MISA), 1990-1999', *International Journal of Occupational and Environmental Health*, 11(1), pp. 107–122. doi: 10.1179/oeh.2005.11.1.107.

Blazejczyk, K. *et al.* (2012) 'Comparison of UTCI to selected thermal indices', *International Journal of Biometeorology*, 56(3), pp. 515–535. doi: 10.1007/s00484-011-0453-2.

Braga, A., Zanobetti, A. and Schwartz, J. (2001) 'The Time Course of

Weather-Related Deaths', *Epidemiology*, 12(6), pp. 662–667. doi: 10.1097/00001648-200111000-00014.

Brake, R. and Bates, G. (2002) 'A valid method for comparing rational and empirical heat stress indices', *Annals of Occupational Hygiene*, 46(2), pp. 165–174. doi: 10.1093/annhyg/mef030.

Breitner, S. *et al.* (2014) 'Short-term effects of air temperature on mortality and effect modification by air pollution in three cities of Bavaria, Germany: A time-series analysis', *Science of the Total Environment*, 485–486(1), pp. 49–61. doi: 10.1016/j.scitotenv.2014.03.048.

Brooke Anderson, G. and Bell, M. L. (2011) 'Heat waves in the United States: Mortality risk during heat waves and effect modification by heat wave characteristics in 43 U.S. communities', *Environmental Health Perspectives*, 119(2), pp. 210–218. doi: 10.1289/ehp.1002313.

Bunker, A. *et al.* (2016) 'Effects of Air Temperature on Climate-Sensitive Mortality and Morbidity Outcomes in the Elderly; a Systematic Review and Metaanalysis of Epidemiological Evidence', *EBioMedicine*, 6, pp. 258–268. doi: 10.1016/j.ebiom.2016.02.034.

Burkart, K. *et al.* (2013a) 'Interactive short-term effects of equivalent temperature and air pollution on human mortality in Berlin and Lisbon.', *Environmental pollution*, 183, pp. 54–63. doi: 10.1016/j.envpol.2013.06.002.

Burkart, K. *et al.* (2013b) 'Interactive short-term effects of equivalent temperature and air pollution on human mortality in Berlin and Lisbon', *Environmental pollution*, 183, pp. 54–63. doi: 10.1016/j.envpol.2013.06.002.

Bytnerowicz, A., Omasa, K. and Paoletti, E. (2007) 'Integrated effects of air pollution and climate change on forests: A northern hemisphere perspective', *Environmental Pollution*, 147(3), pp. 438–445. doi: 10.1016/j.envpol.2006.08.028.

Cai, M. *et al.* (2023) 'Short-term air pollution exposure associated with death from kidney diseases: a nationwide time-stratified case-crossover study in China from 2015 to 2019', *BMC Medicine*, 21(1), pp. 1–12. doi: 10.1186/s12916-023-02734-9.

Cakmak, S., Dales, R. E. and Vidal, C. B. (2007) 'Air pollution and mortality in Chile: Susceptibility among the elderly', *Environmental Health Perspectives*, 115(4), pp. 524–527. doi: 10.1289/ehp.9567.

Callahan, C. W. and Mankin, J. S. (2022) 'Globally unequal effect of extreme

heat on economic growth', *Science Advances*, 8(43), pp. 43–45. doi: 10.1126/sciadv.add3726.

Campbell, S. *et al.* (2018) 'Heatwave and health impact research: A global review', *Health and Place*, 53(August), pp. 210–218. doi: 10.1016/j.healthplace.2018.08.017.

Carolan-Olah, M. and Frankowska, D. (2014) 'High environmental temperature and preterm birth: A review of the evidence', *Midwifery*, 30(1), pp. 50–59. doi: 10.1016/j.midw.2013.01.011.

Chen, K. *et al.* (2018) 'Two-way effect modifications of air pollution and air temperature on total natural and cardiovascular mortality in eight European urban areas', *Environment International*, 116(December 2017), pp. 186–196. doi: 10.1016/j.envint.2018.04.021.

Chen, R. *et al.* (2018) 'Association between ambient temperature and mortality risk and burden: Time series study in 272 main Chinese cities', *BMJ* (*Online*), 363. doi: 10.1136/bmj.k4306.

Chen, R. *et al.* (2019) 'Associations between coarse particulate matter air pollution and cause-specific mortality: A nationwide analysis in 272 Chinese cities', *Environmental Health Perspectives*, 127(1), pp. 5–10. doi: 10.1289/EHP2711.

Cheng, Y. and Kan, H. (2012) 'Effect of the interaction between outdoor air pollution and extreme temperature on daily mortality in Shanghai, China', *Journal of Epidemiology*, 22(1), pp. 28–36. doi: 10.2188/jea.JE20110049.

Chersich, M. F. *et al.* (2020) 'Associations between high temperatures in pregnancy and risk of preterm birth, low birth weight, and stillbirths: Systematic review and meta-analysis', *The BMJ*, 371, pp. 1–13. doi: 10.1136/bmj.m3811.

Christina Adebayo-Ojo, T. *et al.* (2006) 'Short-Term Effects of PM10, NO2, SO2 and O3 on Cardio-Respiratory Mortality in Cape Town, South Africa, 2006–2015', *Int. J. Environ. Res. Public Health*, 19(2).

Corso, M. *et al.* (2020) 'Short-Term Associations of Nitrogen Dioxide (NO2) on Mortality in 18 French Cities, 2010–2014', *Atmosphere*, 11(2). doi: doi:10.3390/atmos11111198.

D. Jarvis, G. Adamkiewicz, M. Heroux, R. Rapp, F. K. (2010) WHO guidelines for indoor air quality: selected pollutants, World Health Organization. Geneva, Switzerland. Available at: https://www.ncbi.nlm.nih.gov/books/NBK138707/.

D'Ippoliti, D. *et al.* (2010) 'The impact of heat waves on mortality in 9 European cities: Results from the EuroHEAT project', *Environmental Health: A Global Access Science Source*, 9(1), pp. 1–9. doi: 10.1186/1476-069X-9-37.

de' Donato, F. K. *et al.* (2015) 'Changes in the effect of heat on mortality in the last 20 years in nine European cities. Results from the PHASE project', *International Journal of Environmental Research and Public Health*, 12(12), pp. 15567–15583. doi: 10.3390/ijerph121215006.

Deb, C. and Alur, R. (2010) 'The significance of Physiological Equivalent Temperature (PET) in outdoor thermal comfort studies Energy Simulation of Singapore Zero Energy Building View project The significance of Physiological Equivalent Temperature (PET) in outdoor thermal comfort st', *International Journal of Engineering Science and Technology*, 2(7), pp. 2825–2828. Available at: https://www.researchgate.net/publication/50315274.

Diapouli, E. *et al.* (2017) 'Evolution of air pollution source contributions over one decade, derived by PM10 and PM2.5 source apportionment in two metropolitan urban areas in Greece', *Atmospheric Environment*, 164, pp. 416– 430. doi: 10.1016/j.atmosenv.2017.06.016.

Diffenbaugh, N. S. *et al.* (2007) 'Heat stress intensification in the Mediterranean climate change hotspot', *Geophysical Research Letters*, 34(11). doi: 10.1029/2007GL030000.

Economou, T. et al. (2023) A unifying modelling approach for hierarchical distributed lag models.

EEA (2012) Air quality in Europe — 2012 report, EEA Report No 4/2012.

EEA (2018) Air quality in Europe, European Environment Agency. doi: 10.2800/777411.

EEA (2019) Air quality in Europe — 2019 report, EEA Report No 10/2019. Available at: http://www.eea.europa.eu/publications/air-quality-in-europe-2012.

EEA (2021) Air quality in Europe 2021: Health impacts of air pollution in Europe, 2021, 2021. Available at: https://www.eea.europa.eu/publications/air-quality-in-europe-2021/health-impacts-of-air-pollution (Accessed: 14 February 2023).

EEA (2022a) *Air quality in Europe 2022*, 2022. Available at: https://www.eea.europa.eu/publications/air-quality-in-europe-2022/air-quality-in-europe-2022 (Accessed: 23 February 2023).

EEA (2022b) Europe's air quality status 2022, 2022. Available at:

https://www.eea.europa.eu/publications/status-of-air-quality-in-Europe-2022/europes-air-quality-status-2022 (Accessed: 28 November 2022).

Epstein, Y. and Moran, D. S. (2006) 'Thermal Comfort and the Heat Stress Indices', *Industrial Health*, 44(3), pp. 388–398. doi: 10.2486/indhealth.44.388.

European Commission (2020) *EC Press Release*, 2020. Available at: https://ec.europa.eu/commission/presscorner/detail/en/ip_20_2151 (Accessed: 15 October 2022).

European Environment Agency (2019) Healthy environment, healthy lives.

European Environment Agency (no date) *How air pollution affects our health*. Available at: https://www.eea.europa.eu/en/topics/in-depth/air-pollution/eow-it-affects-our-health#:~:text=Both short- and long-term,asthma and lower respiratory infections. (Accessed: 16 June 2023).

European Environment Agency (EEA) (2022) European Air Quality Index. Available at: https://airindex.eea.europa.eu/Map/AQI/Viewer/ (Accessed: 2 June 2023).

European Environment Agency (EEA) (2023) *Economic losses from climaterelated extremes in Europe*. Available at: https://www.eea.europa.eu/ims/economic-losses-from-climate-related (Accessed: 19 June 2023).

European Environmental Agency (2020) *Air quality in Europe - 2020 report, EEA Report No 09/2020.* Available at: https://www.eea.europa.eu//publications/air-quality-in-europe-2020-report.

European Observatory on Health Systems and Policies (2019) *Greece: Country Health Profile 2019*. Brussels. Available at: https://read.oecdilibrary.org/social-issues-migration-health/greece-country-health-profile-2019_d87da56a-en#page1%0Ahttps://www.oecd-ilibrary.org/social-issuesmigration-health/greece-country-health-profile-2019_d87da56aen%0Ahttps://read.oecd-ilibrary.o.

European Observatory on Health Systems and Policies (2021) *Greece: Country Health Profile 2021, State of Health in the EU.*

EUROSTAT (2019) Ageing Europe-Looking at the lives of older people in the EU, Publications Office of the European Union. Edited by H. Strandell and P. Wolf. EUROSTAT. doi: 10.1136/bmj.315.7115.1103.

EUROSTAT (2020) Ageing Europe — Looking at the lives of older people in the EU. doi: 10.2785/628105.

Falagas, M. E. *et al.* (2009) 'Seasonality of mortality: The September phenomenon in Mediterranean countries', *Cmaj*, 181(8), pp. 484–486. doi: 10.1503/cmaj.090694.

Fattore, E. *et al.* (2011) 'Human health risk in relation to air quality in two municipalities in an industrialized area of Northern Italy', *Environmental Research*, 111(8), pp. 1321–1327. doi: 10.1016/j.envres.2011.06.012.

Font-Ribera, L. *et al.* (2023) 'Estimating ambient air pollution mortality and disease burden and its economic cost in Barcelona', *Environmental Research*, 216(September 2022). doi: 10.1016/j.envres.2022.114485.

Fuller, R. *et al.* (2022) 'Pollution and health: a progress update', *The Lancet Planetary Health*, 6(6), pp. e535–e547. doi: 10.1016/S2542-5196(22)00090-0.

Galanaki, E. *et al.* (2023) 'Spatio-Temporal Analysis of Heatwaves Characteristics in Greece from 1950 to 2020', *Climate*, 11(1), pp. 14–16. doi: 10.3390/cli11010005.

Gao, X., Pal, J. S. and Giorgi, F. (2006) 'Projected changes in mean and extreme precipitation over the Mediterranean region from a high resolution double nested RCM simulation', *Geophysical Research Letters*, 33(3), pp. 2–5. doi: 10.1029/2005GL024954.

Gariazzo, C. *et al.* (2023) 'Association between short-term exposure to air pollutants and cause-specific daily mortality in Italy. A nationwide analysis', *Environmental Research*, 216(March 2022). doi: 10.1016/j.envres.2022.114676.

Garrett, P. and Casimiro, E. (2011) 'Short-term effect of fine particulate matter (PM2.5) and ozone on daily mortality in Lisbon, Portugal', *Environmental Science and Pollution Research*, 18(9), pp. 1585–1592. doi: 10.1007/s11356-011-0519-z.

Gasparrini, A. (2011a) 'Distributed lag linear and non-linear models in R: The package dlnm', *Journal of Statistical Software*, 43(8), pp. 2–20. doi: 10.18637/jss.v043.i08.

Gasparrini, A. (2011b) 'Distributed Lag Linear and Non-Linear Models in R: The Package dlnm', *Journal of Statistical Software*, 43(8). Available at: http://www.lshtm.ac.uk/people/gasparrini.antonio/.

Gasparrini, A. *et al.* (2012) 'The effect of high temperatures on cause-specific mortality in England and Wales', *Occupational and Environmental Medicine*, 69(1), pp. 56–61. doi: 10.1136/oem.2010.059782.

Gasparrini, A. (2014a) 'Distributed lag linear and non-linear models for time

series data', *143.107.212.50*, pp. 1–12. Available at: http://143.107.212.50/web/packages/dlnm/vignettes/dlnmTS.pdf.

Gasparrini, A. (2014b) 'Modeling exposure-lag-response associations with distributed lag non-linear models', *Statistics in Medicine*, 33(5), pp. 881–899. doi: 10.1002/sim.5963.

Gasparrini, A. *et al.* (2017) 'A penalized framework for distributed lag nonlinear models', *Biometrics*, 73(3), pp. 938–948. doi: 10.1111/biom.12645.

Gasparrini, A. (2021) 'Distributed lag linear and non-linear models for time series data', pp. 1–12. Available at: http://143.107.212.50/web/packages/dlnm/vignettes/dlnmTS.pdf.

Gasparrini, A. and Armstrong, B. (2010) 'Time series analysis on the health effects of temperature: Advancements and limitations', *Environmental Research*, 110(6), pp. 633–638. doi: 10.1016/j.envres.2010.06.005.

Gasparrini, A., Armstrong, B. and Kenward, M. G. (2010) 'Distributed lag non-linear models', *Statistics in Medicine*, 29(21), pp. 2224–2234. doi: 10.1002/sim.3940.

Gasparrini, A., Guo, Y. and Hashizume, M. (2015) 'Mortality risk attributable to high and low ambient temperature: A multicounty observational study', *The Lancet*, 14(6), pp. 464–465. doi: 10.1016/S0140-6736(14)62114-0.

Gasparrini, A. and Leone, M. (2014) 'Attributable risk from distributed lag models', *BMC Medical Research Methodology*, 14(1), pp. 1–8. doi: 10.1186/1471-2288-14-55.

Geller, A. M. and Zenick, H. (2005) 'Aging and the environment: A research framework', *Environmental Health Perspectives*, 113(9), pp. 1257–1262. doi: 10.1289/ehp.7569.

Georgoulias, A. K. *et al.* (2022) 'Climate change projections for Greece in the 21st century from high-resolution EURO-CORDEX RCM simulations', *Atmospheric Research*, p. 106049. doi: 10.1016/j.atmosres.2022.106049.

Giannakopoulos, C. *et al.* (2009) 'Climatic changes and associated impacts in the Mediterranean resulting from a 2 °C global warming', *Global and Planetary Change*, 68(3), pp. 209–224. doi: 10.1016/j.gloplacha.2009.06.001.

Giannaros, C., Melas, D. and Giannaros, T. M. (2019) 'On the short-term simulation of heat waves in the Southeast Mediterranean: Sensitivity of the WRF model to various physics schemes', *Atmospheric Research*, 218(November 2018), pp. 99–116. doi: 10.1016/J.ATMOSRES.2018.11.015.

Giannaros, T. M. *et al.* (2011) 'On the impact of air quality on human health: application of an air quality index for Thessaloniki, Greece', (1), p. 241557.

Giannaros, T. M. and Melas, D. (2012) 'Study of the urban heat island in a coastal Mediterranean City: The case study of Thessaloniki, Greece', *Atmospheric Research*, 118, pp. 103–120. doi: 10.1016/j.atmosres.2012.06.006.

Giorgi, F. and Lionello, P. (2008) 'Climate change projections for the Mediterranean region', *Global and Planetary Change*, 63(2–3), pp. 90–104. doi: 10.1016/j.gloplacha.2007.09.005.

Global Health Impacts of Air Pollution / State of Global Air (no date). Available at: https://www.stateofglobalair.org/health/global (Accessed: 24 May 2023).

González Ortiz, A. *et al.* (2021) *Health Risk Assessment of air pollution, European Environment Agency.* Available at: https://www.eionet.europa.eu/etcs/etc-atni/products/etc-atni-reports/etc-atnireport-10-2021-health-risk-assessments-of-air-pollution-estimations-of-the-2019hra-benefit-analysis-of-reaching-specific-air-quality-standards-andmore/@@download/file/2021-10.

Gordon, C. J. (2003) 'Role of environmental stress in the physiological response to chemical toxicants', *Environmental Research*, 92(1), pp. 1–7. doi: 10.1016/S0013-9351(02)00008-7.

Goudarzi, G. *et al.* (2015) 'Cardiovascular and respiratory mortality attributed to ground-level ozone in Ahvaz, Iran', *Environmental Monitoring and Assessment*, 187(8), pp. 1–9. doi: 10.1007/s10661-015-4674-4.

Gryparis, A. *et al.* (2004) 'Acute effects of ozone on mortality from the "Air Pollution and Health: A European Approach" project', *American Journal of Respiratory and Critical Care Medicine*, 170(10), pp. 1080–1087. doi: 10.1164/rccm.200403-333OC.

Gu, S. *et al.* (2020) 'Associations between ambient air pollution and daily incidence of pediatric hand, foot and mouth disease in Ningbo, 2014-2016: A distributed lag nonlinear model', *Epidemiology and Infection*, pp. 2014–2016. doi: 10.1017/S0950268820000321.

Guo, C.-Y. *et al.* (2021) 'Extensions of the distributed lag non-linear model (DLNM) to account for cumulative mortality', *Environmental Science and Pollution Research*, 28, pp. 38679–38688. doi: 10.1007/s11356-021-13124-0/Published.

Guo, M. *et al.* (2017) 'CO2 emissions from the 2010 Russian wildfires using GOSAT data', *Environmental Pollution*, 226, pp. 60–68. doi: 10.1016/j.envpol.2017.04.014.

Guo, Y. *et al.* (2011) 'The impact of temperature on mortality in Tianjin, china: A case-crossover design with a distributed lag nonlinear model', *Environmental Health Perspectives*, 119(12), pp. 1719–1725. doi: 10.1289/ehp.1103598.

Hajat, S. *et al.* (2005) 'Mortality displacement of heat-related deaths: A comparison of Delhi, São Paulo, and London', *Epidemiology*, 16(5), pp. 613–620. doi: 10.1097/01.ede.0000164559.41092.2a.

Hales, S. *et al.* (2000) 'Daily mortality in relation to weather and air pollution in Christchurch, New Zealand', *Australian and New Zealand Journal of Public Health*, 24(1), pp. 89–91. doi: 10.1111/j.1467-842X.2000.tb00731.x.

Hauptmann, M. *et al.* (2000) 'Analysis of exposure-time-response relationships using a spline weight function', *Biometrics*, 56(4), pp. 1105–1108. doi: 10.1111/j.0006-341X.2000.01105.x.

Havenith, G. et al. (2012) 'The UTCI-clothing model', International Journal of Biometeorology, 56(3), pp. 461–470. doi: 10.1007/s00484-011-0451-4.

Health Canada (2010) Evaluation of the National Air Quality Health Index Program. Available at: https://www.ec.gc.ca/aeve/default.asp?lang=En&n=9918CDC7-1#sdfootnote9sym (Accessed: 15 May 2023).

Health effects of PM10 (2023) *Marlborough District Council*. Available at: https://www.marlborough.govt.nz/environment/air-quality/smoke-and-smog/health-effects-of-pm10 (Accessed: 11 May 2023).

Health Impacts of Ozone | State of Global Air (no date). Available at: https://www.stateofglobalair.org/health/ozone (Accessed: 24 May 2023).

Hellenic Statistical Authority (2023) Population and Housing Census, ELSTAT.

Héroux, M. E. *et al.* (2015) 'Quantifying the health impacts of ambient air pollutants: recommendations of a WHO/Europe project', *International Journal of Public Health*, 60(5), pp. 619–627. doi: 10.1007/s00038-015-0690-y.

Honda, Y. *et al.* (2014) 'Heat-related mortality risk model for climate change impact projection', *Environ Health Prev Med*, (19), pp. 56–63. doi: 10.1007/s12199-013-0354-6.

Hoppe, P. (1999) 'The physiological equivalent temperature - a universal index for the biometeorological assessment of the thermal environment.', *International journal of biometeorology*, 43(2), pp. 71–75. doi: 10.1007/s004840050118.

Iyakaremye, V. et al. (2021) 'Recent Observed Changes in Extreme High-Temperature Events and Associated Meteorological Conditions over Africa', International Journal of Climatology, (November 2021), pp. 1–16. doi: 10.1002/joc.7485.

Jendritzky, G., De Dear, R. and Havenith, G. (2012) 'UTCI - why another thermal index ?', *International Journal of Biometeorology*, 56(3), pp. 421–428. doi: 10.1007/s00484-011-0513-7.

Jhun, I. *et al.* (2014) 'Effect modification of ozone-related mortality risks by temperature in 97 US cities', *Environment International*, 73, pp. 128–134. doi: 10.1016/j.envint.2014.07.009.

Kalabokas, P. D., Adamopoulos, A. D. and Viras, L. G. (2010) 'Atmospheric PM10 particle concentration measurements at central and peripheral urban sites in Athens and Thessaloniki, Greece', *Global Nest Journal*, 12(1), pp. 71–83. doi: 10.30955/gnj.000691.

Kanchan, Gorai, A. K. and Goyal, P. (2015) 'A review on air quality indexing system', *Asian Journal of Atmospheric Environment*, 9(2), pp. 101–113. doi: 10.5572/ajae.2015.9.2.101.

Karanasiou, A. *et al.* (2014) 'Particulate matter and gaseous pollutants in the Mediterranean Basin: Results from the MED-PARTICLES project', *Science of the Total Environment*, 488–489(1), pp. 297–315. doi: 10.1016/j.scitotenv.2014.04.096.

Kassomenos, P. A., Dimitriou, K. and Paschalidou, A. K. (2013) 'Human health damage caused by particulate matter PM10 and ozone in urban environments: The case of Athens, Greece', *Environmental Monitoring and Assessment*, 185(8), pp. 6933–6942. doi: 10.1007/s10661-013-3076-8.

Katsouyanni, K. *et al.* (2001) 'Confounding and effect modification in the short-term effects of ambient particles on total mortality: Results from 29 European cities within the APHEA2 project', *Epidemiology*, 12(5), pp. 521–531. doi: 10.1097/00001648-200109000-00011.

Katsouyanni, K. *et al.* (2009) 'Air pollution and health: a European and North American approach (APHENA).', *Research report (Health Effects Institute)*, (142), pp. 5–90.

Kendrovski, V. *et al.* (2017) 'Quantifying projected heat mortality impacts under 21st-Centurywarming conditions for selected European countries', *International Journal of Environmental Research and Public Health*, 14(7). doi: 10.3390/ijerph14070729.

Kenney, W. L., Craighead, D. H. and Alexander, L. M. (2014) 'Heat waves aging and human cardiovascular health', *Medicine and Science in Sports and Exercise*, 46(10), pp. 1891–1899. doi: 10.1249/MSS.00000000000325.

Keppas, S. *et al.* (2021) 'Urban Heat Island and Future Projections: A Study in Thessaloniki, Greece', in Dobrinkova, N. and Gadzhev, G. (eds) *International conference on Environmental protection and disaster RISKs, ENVIRORISK 2020*, pp. 201–212. doi: 10.1007/978-3-030-70190-1_14.

Keppas, S. C. *et al.* (2021) 'Future climate change impact on urban heat island in two mediterranean cities based on high-resolution regional climate simulations', *Atmosphere*, 12(7). doi: 10.3390/atmos12070884.

Khaniabadi, Y. O. *et al.* (2017) 'Exposure to PM10, NO2, and O3 and impacts on human health', *Environmental Science and Pollution Research*, 24(3), pp. 2781–2789. doi: 10.1007/s11356-016-8038-6.

Kim, Y.-M. *et al.* (2011) 'Comparison of Temperature Indexes for the Impact Assessment of Heat Stress on Heat-Related Mortality', *Environmental Health and Toxicology*, 26. doi: 10.5620/eht.2011.26.e2011009.

Kim, Y. M., Kim, S. and Liu, Y. (2014) 'The impact of climate change on heat-related mortality in six major cities, South Korea, under representative concentration pathways (RCPs)', *Frontiers in Environmental Science*, 2(FEB), pp. 1–11. doi: 10.3389/fenvs.2014.00003.

Koppe, C.; Kovats, S.; Jendritzky, G.; Menne, B. (2004) *Heat-Waves: Risks* and Responses, Health and Global Environmental Change Series, No. 2. Geneva, Switzerland: World Health Organization. Available at: http://www.springerlink.com/content/r8xpnug2028v2503.

Kouis, P. *et al.* (2019) 'The effect of ambient air temperature on cardiovascular and respiratory mortality in Thessaloniki, Greece', *Science of the Total Environment*, 647, pp. 1351–1358. doi: 10.1016/j.scitotenv.2018.08.106.

Kouis, P. *et al.* (2021) 'Heat-related mortality under climate change and the impact of adaptation through air conditioning: A case study from Thessaloniki, Greece', *Environmental Research*, 199(May), p. 111285. doi: 10.1016/j.envres.2021.111285.

Kriit, H. K. *et al.* (2022) 'Using Distributed Lag Non-Linear Models to Estimate Exposure Lag-Response Associations between Long-Term Air Pollution Exposure and Incidence of Cardiovascular Disease', *International Journal of Environmental Research and Public Health*, 19(5). doi: 10.3390/ijerph19052630.

Kunzli, N. *et al.* (2000) 'Public-health impact of outdoor and traffic-related air pollution: a European assessment', *The Lancet*, 356(9232), pp. 795–801. doi: 10.1016/S0140-6736(00)02653-2.

Larrieu, S. *et al.* (2007) 'Short term effects of air pollution on hospitalizations for cardiovascular diseases in eight French cities: The PSAS program', *Science of the Total Environment*, 387(1–3), pp. 105–112. doi: 10.1016/j.scitotenv.2007.07.025.

Lelieveld, J. *et al.* (2002) 'Global air pollution crossroads over the Mediterranean', *Science*, 298(5594), pp. 794–799. doi: 10.1126/science.1075457.

Lelieveld, J. *et al.* (2012) 'Climate change and impacts in the Eastern Mediterranean and the Middle East', *Climatic Change*, 114(3–4), pp. 667–687. doi: 10.1007/s10584-012-0418-4.

Lelieveld, J. *et al.* (2015) 'The contribution of outdoor air pollution sources to premature mortality on a global scale', *Nature*, 525(7569), pp. 367–371. doi: 10.1038/nature15371.

Lin, Y. K. *et al.* (2012) 'High-temperature indices associated with mortality and outpatient visits: Characterizing the association with elevated temperature', *Science of the Total Environment*, 427–428, pp. 41–49. doi: 10.1016/j.scitotenv.2012.04.039.

Liu, C. *et al.* (2019) 'Ambient Particulate Air Pollution and Daily Mortality in 652 Cities', *New England Journal of Medicine*, 381(8), pp. 705–715. doi: 10.1056/nejmoa1817364.

Liu, J. *et al.* (2021) 'Is there an association between hot weather and poor mental health outcomes? A systematic review and meta-analysis', *Environment International*, 153, p. 106533. doi: 10.1016/j.envint.2021.106533.

Liu, L. *et al.* (2011) 'Associations between air temperature and cardiorespiratory mortality in the urban area of Beijing, China: A time-series analysis', *Environmental Health: A Global Access Science Source*, 10(1), pp. 1–11. doi: 10.1186/1476-069X-10-51.

Liu, T. et al. (2022) 'Association between short-term exposure to ambient air pollution and dementia mortality in Chinese adults', Science of the Total

Environment, 849(August), p. 157860. doi: 10.1016/j.scitotenv.2022.157860.

Lubczyńska, J. M., Christophi, C. A. and Lelieveld, J. (2015) 'Heat-related cardiovascular mortality risk in Cyprus : a case-crossover study using a distributed lag non-linear model', *Environmental Health*, 14(39), pp. 1–11. doi: 10.1186/s12940-015-0025-8.

Ma, W., Chen, R. and Kan, H. (2014) 'Temperature-related mortality in 17 large Chinese cities: How heat and cold affect mortality in China', *Environmental Research*, 134, pp. 127–133. doi: 10.1016/j.envres.2014.07.007.

Ma, Y., Zhou, L. and Chen, K. (2020) 'Burden of cause-specific mortality attributable to heat and cold: A multicity time-series study in Jiangsu Province, China', *Environment International*, 144(172), p. 105994. doi: 10.1016/j.envint.2020.105994.

Makri, A. and Stilianakis, N. I. (2008) 'Vulnerability to air pollution health effects', *International Journal of Hygiene and Environmental Health*, 211(3–4), pp. 326–336. doi: 10.1016/j.ijheh.2007.06.005.

Malashock, D. *et al.* (2022) 'Trends in Ozone Concentration and Attributable Mortality for Urban, Peri-Urban and Rural Areas Worldwide between 2000 and 2019: Estimates from Global Datasets', *SSRN Electronic Journal*, 6(12), pp. e958–e967. doi: 10.2139/ssrn.4155322.

Martínez-Solanas, È. *et al.* (2018) 'Evaluation of the impact of ambient temperatures on occupational injuries in Spain', *Environmental Health Perspectives*, 126(6), pp. 1–10. doi: 10.1289/EHP2590.

Martínez-Solanas, È. *et al.* (2021) 'Projections of temperature-attributable mortality in Europe: a time series analysis of 147 contiguous regions in 16 countries', *The Lancet Planetary Health*, 5(7), pp. e446–e454. doi: 10.1016/S2542-5196(21)00150-9.

Matthies, F. *et al.* (2008) *Heat–health action plans-Guidance*, *WHO Europe*. Available at: http://www.euro.who.int/___data/assets/pdf_file/0006/95919/E91347.pdf.

Matzarakis, A., Mayer, H. and Iziomon, M. (1999) 'Applications of a universal thermal index: PET', *International Journal of Biometeorology*, 43, pp. 76–84.

McGregor, G. R. et al. (2015) Heatwaves and Health: Guidance on Warning-System Development. World Meteorological Organization and World Health Organization. Available at: http://www.who.int/globalchange/publications/WMO_WHO_Heat_Health_Guida nce_2015.pdf.

Meehl, G. A. and Tebaldi, C. (2004) 'More intense, more frequent, and longer lasting heat waves in the 21st century', *Science*, 305(5686), pp. 994–997. doi: 10.1126/science.1098704.

Melas, D. *et al.* (2017) 'An Up-to-Date Assessment of the Air Quality in Greece with the Use of Modeling Tools', *Perspectives on Atmospheric Sciences*, pp. 1123–1128. doi: 10.1007/978-3-319-35095-0_161.

Meng, X. *et al.* (2012) 'Temperature modifies the acute effect of particulate air pollution on mortality in eight Chinese cities', *Science of the Total Environment*, 435–436, pp. 215–221. doi: 10.1016/j.scitotenv.2012.07.008.

Meng, X. *et al.* (2021) 'Short term associations of ambient nitrogen dioxide with daily total, cardiovascular, and respiratory mortality: multilocation analysis in 398 cities', *BMJ*, 372(2). doi: 10.1136/bmj.n534.

Michelozzi, P. *et al.* (2006) 'Temperature and summer mortality: Geographical and temporal variations in four Italian cities', *Journal of Epidemiology and Community Health*, 60(5), pp. 417–423. doi: 10.1136/jech.2005.040857.

Michelozzi, P. *et al.* (2010) 'Surveillance of summer mortality and preparedness to reduce the health impact of heat waves in Italy', *International Journal of Environmental Research and Public Health*, 7(5), pp. 2256–2273. doi: 10.3390/ijerph7052256.

Mie, Z. *et al.* (2022) 'Recent variability of sub-seasonal monsoon precipitation and its potential drivers in Myanmar using in-situ observation during 1981-2020', *Int J Climatol*, 42, pp. 3341–3359. doi: 10.1002/joc.7419.

Miles-Novelo, A. and Anderson, C. A. (2019) 'Climate Change and Psychology: Effects of Rapid Global Warming on Violence and Aggression', *Current Climate Change Reports*, 5(1), pp. 36–46. doi: 10.1007/s40641-019-00121-2.

Ministry of Environment and Energy (2020) Annual Report on Air Quality 2019. Athens, Greece.

Ministry of the Environment and Energy (2023) Annual report on air quality 2022.

Miranda, A. *et al.* (2015) 'Current air quality plans in Europe designed to support air quality management policies', *Atmospheric Pollution Research*, 6(3),

pp. 434–443. doi: 10.5094/APR.2015.048.

Mo, S. *et al.* (2023) 'Short-term effects of fine particulate matter constituents on myocardial infarction death', *Journal of Environmental Sciences (China)*, 133, pp. 60–69. doi: 10.1016/j.jes.2022.07.019.

Moussiopoulos *et al.* (2009) 'Air quality status in Greater Thessaloniki Area and the emission reductions needed for attaining the EU air quality legislation', *Science of the Total Environment*, 407(4), pp. 1268–1285. doi: 10.1016/j.scitotenv.2008.10.034.

Muggeo, V. M. R. (2010) 'Analyzing temperature effects on mortality within the R environment: The constrained segmented distributed lag parameterization', *Journal of Statistical Software*, 32(12), pp. 1–17. doi: 10.18637/jss.v032.i12.

Nitrogen Dioxide | American Lung Association (no date). Available at: https://www.lung.org/clean-air/outdoors/what-makes-air-unhealthy/nitrogen-dioxide (Accessed: 16 May 2023).

Nordio, F. *et al.* (2015) 'Changing patterns of the temperature-mortality association by time and location in the US, and implications for climate change', *Environment International*, 81, pp. 80–86. doi: 10.1016/j.envint.2015.04.009.

OECD/European Union (2022) *Health at a Glance: Europe 2022: State of Health in the EU, OECD Publishing.* Available at: https://doi.org/10.1787/507433b0-en.

Olstrup, H. *et al.* (2019) 'Association between mortality and short-term exposure to particles, ozone and nitrogen dioxide in Stockholm, Sweden', *International Journal of Environmental Research and Public Health*, 16(6). doi: 10.3390/ijerph16061028.

Olstrup, H., Åström, C. and Orru, H. (2022) 'Daily Mortality in Different Age Groups Associated with Exposure to Particles, Nitrogen Dioxide and Ozone in Two Northern European Capitals: Stockholm and Tallinn', *Environments - MDPI*, 9(7). doi: 10.3390/environments9070083.

Orellano, P. *et al.* (2020) 'Short-term exposure to particulate matter (PM10 and PM2.5), nitrogen dioxide (NO2), and ozone (O3) and all-cause and cause-specific mortality: Systematic review and meta-analysis', *Environment International*, 142(June), p. 105876. doi: 10.1016/j.envint.2020.105876.

Ortiz, C. *et al.* (2017) 'Evaluation of short-term mortality attributable to particulate matter pollution in Spain', *Environmental Pollution*, 224, pp. 541–551. doi: 10.1016/j.envpol.2017.02.037.

Osilla, E. V, Marsidi, J. L. and Sharma, S. (2023) 'Physiology, Temperature Regulation', *StatPearls*, pp. 1–6. Available at: https://www.ncbi.nlm.nih.gov/books/NBK507838/.

Oudin Åström, D., Bertil, F. and Joacim, R. (2011) 'Heat wave impact on morbidity and mortality in the elderly population: A review of recent studies', *Maturitas*, 69(2), pp. 99–105. doi: 10.1016/j.maturitas.2011.03.008.

Papanastasiou, D. K., Melas, D. and Kambezidis, H. D. (2015) 'Air quality and thermal comfort levels under extreme hot weather', *Atmospheric Research*, 152, pp. 4–13. doi: 10.1016/j.atmosres.2014.06.002.

Paravantis, J. *et al.* (2017) 'Mortality associated with high ambient temperatures, heatwaves, and the urban heat island in Athens, Greece', *Sustainability (Switzerland)*, 9(4). doi: 10.3390/su9040606.

Parliari, D. et al. (2022) Short-Term Effects of Apparent Temperature on Cause-Specific Mortality in the Urban Area of Thessaloniki, Greece, Atmosphere. doi: 10.3390/atmos13060852.

Parliari, D., Keppas, S., *et al.* (2023) 'Projections of Heat-Related Mortality under the Impact of Climate Change in Thessaloniki , Greece', in *Environmental Sciences Proceedings*. doi: https://doi.org/10.3390/environsciproc2023026072.

Parliari, D., Giannaros, C., *et al.* (2023) 'Short-Term Effects of Air Pollution on Mortality in the Urban Area of Thessaloniki, Greece', *Sustainability*, 15(6), p. 548. doi: https://doi.org/ 10.3390/su15065305.

Parsons, K. (2003) *Human Thermal Environments*. 2nd edn, *Taylor and Francis*. 2nd edn. Taylor & Francis. doi: 10.7202/1022009ar.

Pascal, M. *et al.* (2014) 'Short-term impacts of particulate matter (PM10, PM10-2.5, PM2.5) on mortality in nine French cities', *Atmospheric Environment*, 95, pp. 175–184. doi: 10.1016/j.atmosenv.2014.06.030.

Peng, R. D. *et al.* (2013) 'Acute effects of ambient ozone on mortality in Europe and North America: Results from the APHENA study', *Air Quality, Atmosphere and Health*, 6(2), pp. 445–453. doi: 10.1007/s11869-012-0180-9.

Peng, R. D., Dominici, F. and Louis, T. A. (2006) 'Model choice in time series studies of air pollution and mortality', *Journal of the Royal Statistical Society. Series A: Statistics in Society*, 169(2), pp. 179–203. doi: 10.1111/j.1467-985X.2006.00410.x.

Peng, R. and Dominici, F. (2008) Statistical Methods for Environmental Epidemiology with R: A Case Study in Air Pollution and Health, Springer. New

York: Springer. doi: 10.1111/j.1541-0420.2009.01315_10.x.

Plaia, A. and Ruggieri, M. (2011) 'Air quality indices: A review', *Reviews in Environmental Science and Biotechnology*, 10(2), pp. 165–179. doi: 10.1007/s11157-010-9227-2.

Pleijel, H. (2000) 'Ground-level ozone. A problem largely ignored in southern Europe', in *AIR POLLUTION AND CLIMATE SERIES*. Swedish NGO Secretariat on Acid Rain, p. 28. Available at: http://www.airclim.org/sites/default/files/documents/APC12.pdf.

Population-Housing Census 2021 - ELSTAT (no date) Hellenic Statistical Authority. Available at: https://www.statistics.gr/2021-census-pop-hous (Accessed: 24 May 2023).

Poupkou, A. *et al.* (2011) 'Climatology of discomfort index and air quality index in a large urban mediterranean agglomeration', *Water, Air, and Soil Pollution*, 222(1–4), pp. 163–183. doi: 10.1007/s11270-011-0814-9.

Pozzer, A. *et al.* (2023) 'Mortality Attributable to Ambient Air Pollution : A Review of Global Estimates GeoHealth', *HeoHealth*, 7(1). doi: 10.1029/2022GH000711.

Progiou, A. *et al.* (2023) 'Measures and Policies for Reducing PM Exceedances through the Use of Air Quality Modeling : The Case of Thessaloniki, Greece', *Sustainability*, (15). doi: https://doi.org/ 10.3390/su15020930.

Psistaki, K. *et al.* (2022) 'Exploring the Impact of Particulate Matter on Mortality in Coastal Mediterranean Environments', *SSRN Electronic Journal*, 865(September 2022), p. 161147. doi: 10.2139/ssrn.4238462.

Psistaki, K., Dokas, I. M. and Paschalidou, A. K. (2022) 'Analysis of the heatand cold-related cardiovascular mortality in an urban mediterranean environment through various thermal indices', *Environmental Research*, 216(P4), p. 114831. doi: 10.1016/j.envres.2022.114831.

Psistaki, K., Dokas, I. M. and Paschalidou, A. K. (2023) 'The Impact of Ambient Temperature on Cardiorespiratory Mortality in Northern Greece', *International Journal of Environmental Research and Public Health*, 20(1). doi: 10.3390/ijerph20010555.

Pyrgou, A., Hadjinicolaou, P. and Santamouris, M. (2018) 'Enhanced nearsurface ozone under heatwave conditions in a Mediterranean island', *Scientific Reports*, 8(1), pp. 1–10. doi: 10.1038/s41598-018-27590-z.

Qian, Z. et al. (2010) 'Seasonal pattern of the acute mortality effects of air

pollution', Journal of the Air and Waste Management Association, 60(4), pp. 481–488. doi: 10.3155/1047-3289.60.4.481.

Rai, M. *et al.* (2023) 'Heat-related cardiorespiratory mortality: effect modification by air pollution across 482 cities from 24 countries', *Environment International*, 174. doi: 10.1016/j.envint.2023.107825.

Ren, C. and Tong, S. (2006) 'Temperature modifies the health effects of particulate matter in Brisbane, Australia', *International Journal of Biometeorology*, 51(2), pp. 87–96. doi: 10.1007/s00484-006-0054-7.

Revich, B. and Shaposhnikov, D. (2010) 'The effects of particulate and ozone pollution on mortality in Moscow, Russia', *Air Quality, Atmosphere and Health*, 3(2), pp. 117–123. doi: 10.1007/s11869-009-0058-7.

Rizos, K. *et al.* (2022) 'Determination of the background pollution in the Eastern Mediterranean applying a statistical clustering technique', *Atmospheric Environment*, 276(March), p. 119067. doi: 10.1016/j.atmosenv.2022.119067.

Roberts, S. and Martin, M. A. (2007) 'A distributed lag approach to fitting non-linear dose-response models in particulate matter air pollution time series investigations', *Environmental Research*, 104(2), pp. 193–200. doi: 10.1016/j.envres.2007.01.009.

Rodopoulou, S. *et al.* (2015) 'Searching for the best modeling specification for assessing the effects of temperature and humidity on health: a time series analysis in three European cities', *International Journal of Biometeorology*, 59(11), pp. 1585–1596. doi: 10.1007/s00484-015-0965-2.

Rodrigues, M., Santana, P. and Rocha, A. (2019) 'Projections of temperatureattributable deaths in portuguese metropolitan areas: A time-series modelling approach', *Atmosphere*, 10(12). doi: 10.3390/ATMOS10120735.

Rotton, J. and Cohn, E. G. (2000) 'Weather, disorderly conduct, and assaults: From social contact to social avoidance', *Environment and Behavior*, 32(5), pp. 651–673. doi: 10.1177/0013916500325004.

Sacks, J. D. *et al.* (2011) 'Particulate matter-induced health effects: Who is susceptible?', *Environmental Health Perspectives*, 119(4), pp. 446–454. doi: 10.1289/ehp.1002255.

Samet, J. *et al.* (1998) 'Does weather confound or modify the association of particulate air pollution with mortality?: An analysis of the Philadelphia data, 1973-1980', *Environmental Research*, 77(1), pp. 9–19. doi: 10.1006/enrs.1997.3821.

Samoli, E. *et al.* (2008) 'Acute effects of ambient particulate matter on mortality in Europe and North America: Results from the APHENA study', *Environmental Health Perspectives*, 116(11), pp. 1480–1486. doi: 10.1289/ehp.11345.

Samoli, E. *et al.* (2009) 'The temporal pattern of mortality responses to ambient ozone in the APHEA project', *Journal of Epidemiology and Community Health*, 63(12), pp. 960–966. doi: 10.1136/jech.2008.084012.

Samoli, E. *et al.* (2013) 'Associations between fine and coarse particles and mortality in Mediterranean cities: Results from the MED-PARTICLES project', *Environmental Health Perspectives*, 121(8), pp. 932–938. doi: 10.1289/ehp.1206124.

Samoli, E. *et al.* (2014) 'Which specific causes of death are associated with short term exposure to fine and coarse particles in Southern Europe? Results from the MED-PARTICLES project', *Environment International*, 67, pp. 54–61. doi: 10.1016/j.envint.2014.02.013.

Schmit, C. *et al.* (2017) 'Cognitive Functioning and Heat Strain: Performance Responses and Protective Strategies', *Sports Medicine*, 2(7), pp. 1289–1302. doi: 10.1007/s40279-016-0657-z.

Schnell, J. L. and Prather, M. J. (2017) 'Co-occurrence of extremes in surface ozone, particulate matter, and temperature over eastern North America.', *Proceedings of the National Academy of Sciences of the United States of America*, 114(11), pp. 2854–2859. doi: 10.1073/pnas.1614453114.

Schwartz, J. (2000) 'Harvesting and long term exposure effects in the relation between air pollution and mortality', *American Journal of Epidemiology*, 151(5), pp. 440–448. doi: 10.1093/oxfordjournals.aje.a010228.

Scortichini, M. *et al.* (2018) 'Short-term effects of heat on mortality and effect modification by air pollution in 25 italian cities', *International Journal of Environmental Research and Public Health*, 15(8). doi: 10.3390/ijerph15081771.

Sein, Z. M. M. *et al.* (2022) 'Observed spatiotemporal changes in air temperature, dew point temperature and relative humidity over Myanmar during 2001–2019', *Meteorology and Atmospheric Physics*, 134(1), pp. 1–17. doi: 10.1007/s00703-021-00837-7.

Shang, Y. *et al.* (2013) 'Systematic review of Chinese studies of short-term exposure to air pollution and daily mortality', *Environment International*, 54, pp. 100–111. doi: 10.1016/j.envint.2013.01.010.

Shi, L. *et al.* (2019) 'Climate Penalty: Climate-driven increases in ozone and PM2.5 levels and mortality', in *ABSTRACTS OF THE 2019 ANNUAL CONFERENCE OF THE INTERNATIONAL SOCIETY FOR ENVIRONMENTAL EPIDEMIOLOGY.* UTRECHT, THE NETHERLANDS, p. 2010. doi: 10.1097/01.EE9.0000610052.11501.34.

Shim, J. I., Byun, G. and Lee, J. T. T. (2023) 'Long-term exposure to particulate matter and risk of Alzheimer's disease and vascular dementia in Korea: a national population-based Cohort Study', *Environmental health : a global access science source*, 22(1), p. 35. doi: 10.1186/s12940-023-00986-9.

Shooter, D. and Brimblecombe, P. (2009) 'Air quality indexing', *International Journal of Environment and Pollution*, 36(1–3), pp. 305–323. doi: 10.1504/ijep.2009.021834.

Shumake, K. L. *et al.* (2013) 'Susceptibility of older adults to health effects induced by ambient air pollutants regulated by the European Union and the United States', *Aging Clinical and Experimental Research*, 25(1), pp. 3–8. doi: 10.1007/s40520-013-0001-5.

Sicard, P. *et al.* (2021) 'Urban population exposure to air pollution in Europe over the last decades', *Environmental Sciences Europe*, 33(1). doi: 10.1186/s12302-020-00450-2.

Silveira, C. *et al.* (2016) 'Assessment of health benefits related to air quality improvement strategies in urban areas: An Impact Pathway Approach', *Journal of Environmental Management*, 183, pp. 694–702. doi: 10.1016/j.jenvman.2016.08.079.

Song, J. *et al.* (2023) 'Premature mortality attributable to NO2 exposure in cities and the role of built environment: A global analysis', *Science of the Total Environment*, 866(2), p. 161395. doi: 10.1016/j.scitotenv.2023.161395.

Song, X. *et al.* (2017) 'Impact of ambient temperature on morbidity and mortality: An overview of reviews', *Science of the Total Environment*, 586(222), pp. 241–254. doi: 10.1016/j.scitotenv.2017.01.212.

Stafoggia, M. *et al.* (2008) 'Does temperature modify the association between air pollution and mortality? A multicity case-crossover analysis in Italy.', *American journal of epidemiology*, 167(12), pp. 1476–1485. doi: 10.1093/aje/kwn074.

Stafoggia, M. et al. (2010) 'Susceptibility factors to ozone-related mortality: A population-based case-crossover analysis', American Journal of Respiratory and Critical Care Medicine, 182(3), pp. 376–384. doi: 10.1164/rccm.200908-

1269OC.

Stanaway, J. D. *et al.* (2018) 'Global, regional, and national comparative risk assessment of 84 behavioural, environmental and occupational, and metabolic risks or clusters of risks for 195 countries and territories, 1990-2017: A systematic analysis for the Global Burden of Disease Stu', *The Lancet*, 392(10159), pp. 1923–1994. doi: 10.1016/S0140-6736(18)32225-6.

Steenland, K. and Armstrong, B. (2006) 'An overview of methods for calculating the burden of disease due to specific risk factors', *Epidemiology*, 17(5), pp. 512–519. doi: 10.1097/01.ede.0000229155.05644.43.

Stieb, D. M. *et al.* (2008) 'A new multipollutant, no-threshold air quality health index based on short-term associations observed in daily time-series analyses', *Journal of the Air and Waste Management Association*, 58(3), pp. 435–450. doi: 10.3155/1047-3289.58.3.435.

Stojić, S. S. *et al.* (2016) 'Single and combined effects of air pollutants on circulatory and respiratory system-related mortality in Belgrade, Serbia', *Journal of Toxicology and Environmental Health - Part A: Current Issues*, 79(1), pp. 17–27. doi: 10.1080/15287394.2015.1101407.

Strand, L. B., Barnett, A. G. and Tong, S. (2011) 'The influence of season and ambient temperature on birth outcomes: A review of the epidemiological literature', *Environmental Research*, 111(3), pp. 451–462. doi: 10.1016/j.envres.2011.01.023.

Sylvestre, M.-P. and Abrahamowicz, M. (2009) 'Flexible modeling of the cumulative effects of time-dependent exposures on the hazard', *Statistics in Medicine*, 28, pp. 3437–3453. doi: 10.1002/sim.

Thom, E. C. (1959) 'The Discomfort Index', *Weatherwise*, 12(2), pp. 57–61. doi: 10.1080/00431672.1959.9926960.

Thomas, D. C. (1988) 'Models for exposure-time-response relationships with applications to cancer epidemiology', *Ann. Rev. Public Health*, 9, pp. 451–482.

Tobías, A., Armstrong, B. and Gasparrini, A. (2017) 'Brief report: Investigating uncertainty in the minimum mortality temperature', *Epidemiology*, 28(1), pp. 72–76. doi: 10.1097/EDE.00000000000567.

Touloumi, G. *et al.* (2006) 'Seasonal confounding in air pollution and health time-series studies: Effect on air pollution effect estimates', *Statistics in Medicine*, 25(April), pp. 4164–4178. doi: 10.1002/sim.2681.

Tsoutsoubi, L., Ioannou, L. G. and Flouris, A. D. (2021) 'Mortality due to

circulatory causes in hot and cold environments in Greece', *Scandinavian Cardiovascular Journal*, 55(6), pp. 333–335. doi: 10.1080/14017431.2021.1970801.

Turner, L. R. *et al.* (2012) 'Ambient temperature and cardiorespiratory morbidity: A systematic review and meta-analysis', *Epidemiology*, 23(4), pp. 594–606. doi: 10.1097/EDE.0b013e3182572795.

Ulbrich, U. *et al.* (2006) 'Chapter 8 The Mediterranean climate change under global warming', *Developments in Earth and Environmental Sciences*, 4(C), pp. 399–415. doi: 10.1016/S1571-9197(06)80011-X.

Ullah, I. *et al.* (2021) 'Evaluating the meteorological drought characteristics over Pakistan using in situ observations and reanalysis products', *International Journal of Climatology*, 41(9), pp. 4437–4459. doi: 10.1002/joc.7063.

Ullah, I. *et al.* (2022) 'Projected Changes in Socioeconomic Exposure to Heatwaves in South Asia Under Changing Climate', *Earth's Future*, 10(2), pp. 1–19. doi: 10.1029/2021EF002240.

Ulpiani, G. (2021) 'On the linkage between urban heat island and urban pollution island: Three-decade literature review towards a conceptual framework', *Science of the Total Environment*, 751, p. 141727. doi: 10.1016/j.scitotenv.2020.141727.

UN Department of Economic and Social Affairs (2018) 2018 Revision of the World Urbanization Prospects, United Nations News. Available at: https://www.un.org/development/desa/en/news/population/2018-revision-of-world-urbanization-prospects.html.

Union of Concerned Scientists (2019) *Killer Heat in the United States*. Available at: https://ucsusa.maps.arcgis.com/apps/MapSeries/index.html?appid=e4e9082a1ec3 43c794d27f3e12dd006d.

United States Environmental Protection Agency (2014) *Air Quality Index: a guide to air quality and your health, Epa.* doi: 10.1023/A:1020370119096.

Vacek, P. M. (1997) 'Assessing the effect of intensity when exposure varies over time', *Statistics in Medicine*, 16(5), pp. 505–513. doi: 10.1002/(SICI)1097-0258(19970315)16:5<505::AID-SIM424>3.0.CO;2-Z.

Valjarević, A. *et al.* (2021) 'Long term monitoring and connection between topography and cloud cover distribution in serbia', *Atmosphere*, 12(8), pp. 1–22. doi: 10.3390/atmos12080964.

Vicedo-Cabrera, A. M. *et al.* (2020) 'Short term association between ozone and mortality: global two stage time series study in 406 locations in 20 countries', *The BMJ*, 368, pp. 1–10. doi: 10.1136/bmj.m108.

Vicedo-Cabrera, A. M. *et al.* (2021) 'The burden of heat-related mortality attributable to recent human-induced climate change', *Nature Climate Change*, 11(6), pp. 492–500. doi: 10.1038/s41558-021-01058-x.

Vlachokostas, C. *et al.* (2012) 'Health effects and social costs of particulate and photochemical urban air pollution: A case study for Thessaloniki, Greece', *Air Quality, Atmosphere and Health*, 5(3), pp. 325–334. doi: 10.1007/s11869-010-0096-1.

Vlachokostas, C. *et al.* (2013) 'Synergies between environmental pressures in the urban climate: Combined air quality and noise exposure assessment in Thessaloniki, Greece', *Global Nest Journal*, 15(2), pp. 209–217. doi: 10.30955/gnj.000962.

Wang, S. *et al.* (2009) 'A study on variations of concentrations of particulate matter with different sizes in Lanzhou, China', *Atmospheric Environment*, 43(17), pp. 2823–2828. doi: 10.1016/j.atmosenv.2009.02.021.

Wang, X. *et al.* (2018) 'Ambient coarse particulate pollution and mortality in three Chinese cities: Association and attributable mortality burden', *Science of the Total Environment*, 628–629, pp. 1037–1042. doi: 10.1016/j.scitotenv.2018.02.100.

Watts, N., Adger, W. N. and Agnolucci, P. (2015) 'Health and climate change: Policy responses to protect public health', *Lancet*, 386(6), pp. 1861–1914. doi: 10.1016/S0140-6736(15)60854-6.

WHO (2016) Ambient air pollution: A global assessment of exposure and burden of disease.

WHO (2021) 'WHO global air quality guidelines', *Coastal And Estuarine Processes*, pp. 1–360.

WHO (2022) World health statistics 2022 (Monitoring health of the SDGs). Available at: http://apps.who.int/bookorders.

WHO (no date a) *THE GLOBAL HEALTH OBSERVATORY*, 2023. Available at: https://www.who.int/data/gho/data/indicators/indicator-details/GHO/ambient-air-pollution-attributable-dalys-(per-100-000-population) (Accessed: 2 December 2022).

WHO (no date b) THE GLOBAL HEALTH OBSERVATORY, 2023. Available

at: https://www.who.int/data/gho/data/indicators/indicator-details/GHO/ambientair-pollution-attributable-ylls (Accessed: 2 January 2023).

WHO Regional Office for Europe (2013) 'Health risks of air pollution in Europe – HRAPIE project', *World Heath Organization (WHO)*, p. 60. doi: 10.1021/acs.est.5b05833.

WHO Regional Office for Europe (2022) The European health report 2021.

Wood, S. (2017) *Generalized Additive Models: An Introduction with R*. 2nd Editio, *Chapman and Hall/CRC*. 2nd Editio. New York: Chapman and Hall/CRC. doi: https://doi.org/10.1201/9781315370279.

Wood, S. N. (2011) 'Fast stable restricted maximum likelihood and marginal likelihood estimation of semiparametric generalized linear models', *Journal of the Royal Statistical Society. Series B: Statistical Methodology*, 73(1), pp. 3–36. doi: 10.1111/j.1467-9868.2010.00749.x.

World Health Organization (2006) WHO Air Quality Guidelines. Global update 2005., World Health Organization. doi: 10.1007/BF02986808.

World Health Organization (2017) 'Urban green spaces: A brief for action', *Regional Office For Europe*, p. 24. doi: 10.1590/S1516-89132004000200018.

World Health Organization (2022) *Heat threatens health: key figures for Europe*. Available at: https://www.euro.who.int/en/health-topics/environment-and-health/Climate-change/archive/heat-threatens-health-key-figures-for-europe (Accessed: 29 April 2022).

Xu, Z. *et al.* (2019) 'Heatwaves, hospitalizations for Alzheimer's disease, and postdischarge deaths: A population-based cohort study', *Environmental Research*, 178(August), p. 108714. doi: 10.1016/j.envres.2019.108714.

Yan, S. *et al.* (2019) 'Short-term effects of meteorological factors and air pollutants on hand, foot and mouth disease among children in Shenzhen, China, 2009-2017', *International Journal of Environmental Research and Public Health*, 16(19). doi: 10.3390/ijerph16193639.

Yari, A. R. *et al.* (2016) 'Study of ground-level ozone and its health risk assessment in residents in Ahvaz City, Iran during 2013', *Toxin Reviews*, 35(3–4), pp. 201–206. doi: 10.1080/15569543.2016.1225769.

Yin, Q. *et al.* (2019) 'Mapping the increased minimum mortality temperatures in the context of global climate change', *Nature Communications*, 10(1), pp. 1–8. doi: 10.1038/s41467-019-12663-y.

Zafeiratou, S. *et al.* (2019) 'Spatial variability in the effect of high ambient temperature on mortality: An analysis at municipality level within the greater athens area', *International Journal of Environmental Research and Public Health*, 16(19), pp. 1–20. doi: 10.3390/ijerph16193689.

Zanis, P. *et al.* (2015) 'Transient high-resolution regional climate simulation for Greece over the period 1960-2100: Evaluation and future projections', *Climate Research*, 64(2), pp. 123–140. doi: 10.3354/cr01304.

Zanobetti, A. *et al.* (2002) 'The Temporal Pattern of Mortality Responses to Air Pollution: A Multicity Assessment of Mortality Displacement', *EPIDEMIOLOGY*, 13(1), pp. 87–93. doi: https://doi.org/10.1097/00001648-200201000-00014.

Zanobetti, A. *et al.* (2003) 'The temporal pattern of respiratory and heart disease mortality in response to air pollution', *Environmental Health Perspectives*, 111(9), pp. 1188–1193. doi: 10.1289/ehp.5712.

Zare, S. *et al.* (2018) 'Comparing Universal Thermal Climate Index (UTCI) with selected thermal indices/environmental parameters during 12 months of the year', *Weather and Climate Extremes*, 19, pp. 49–57. doi: 10.1016/j.wace.2018.01.004.

Zhang, K. *et al.* (2014) 'What weather variables are important in predicting heat-related mortality? A new application of statistical learning methods', *Environmental Research*, 132, pp. 350–359. doi: 10.1016/j.envres.2014.04.004.

Zhang, Y. *et al.* (2016) 'The short-term effect of ambient temperature on mortality in Wuhan, China: A time-series study using a distributed lag non-linear model', *International Journal of Environmental Research and Public Health*, 13(7). doi: 10.3390/ijerph13070722.

Zhang, Yunquan *et al.* (2022) 'Long-term exposure to ambient NO2 and adult mortality : A nationwide cohort study in China', *Journal of Advanced Research*, 41, pp. 13–22. doi: 10.1016/j.jare.2022.02.007.

Zhou, Y. *et al.* (2023) 'Association of biomass fuel use with the risk of vision impairment among Chinese older adults: a cohort study', *Environmental Science and Pollution Research*, (13), pp. 56273–56283. doi: 10.1007/s11356-023-26325-6.

Zinzi, M. and Santamouris, M. (2019) 'Introducing urban overheatingprogress on mitigation science and engineering applications', *Climate*, 7(1), pp. 1–6. doi: 10.3390/cli7010015. Zoumakis, M. et al. (2011) 'Human discomfort due to environmental conditions in Urban Thessaloniki, Greece. Part I. Air quality stress levels.', *Journal of Environmental Protection and Ecology*, 12(2), pp. 407–415.

Zyrichidou, I. *et al.* (2019) 'Adverse results of the economic crisis: A study on the emergence of enhanced formaldehyde (HCHO) levels seen from satellites over Greek urban sites', *Atmospheric Research*, 224(March), pp. 42–51. doi: 10.1016/j.atmosres.2019.03.017.

Appendix A: List of Figures

Figure 1.1	Health impacts of air pollution (Adapted from (European Environment Agency, 2019))	Page 2
Figure 1.2	Health impacts of heat (Adapted from (Union of Concerned Scientists, 2019))	Page 3
Figure 2.1	AQI scale and health warnings (Adapted from (United States Environmental Protection Agency, 2014))	Page 15
Figure 2.2	EAQI health messages (Adapted from (European Environment Agency (EEA), 2022))	Page 16
Figure 2.3	AQHI scale and health messages (Adapted from (Health Canada, 2010))	Page 18
Figure 2.4	Hypothetical lag structure corresponding to the mortality displacement effect (Adapted from (Zanobetti et al., 2002))	Page 20
Figure 3.1	Premature deaths due to PM2.5, 2019 (Adapted from (OECD/European Union, 2022))	Page 25
Figure 3.2	The urban area of Thessaloniki. Dashed lines represent municipal borders.	Page 26
Figure 4.1	Relative risk estimates for the Thessaloniki data (Adapted from (Economou, Parliari, Tobias, Dawkins and Stoner, 2023)).	Page 34
Figure 4.2	Cumulative relative risk estimates for the Thessaloniki data (Adapted from (Economou, Parliari, Tobias, Dawkins and Stoner, 2023)	Page 34
Figure 5.1	The study area. Topography is shaded with linear altitude (above sea level) scales. The locations of Thessaloniki centre and Airport weather station are shown.	Page 39
Figure 5.2	Exposure-lag-response risk surface demonstrating the nonlinear association between temperature and mortality, as	Page 43
calculated for the city of Thessaloniki for the period 2006–2016.

- **Figure 5.3** Overall cumulative exposure–response association (top panel) Page 44 and Tappmax distribution (bottom panel), as calculated for the city of Thessaloniki for the period 2006–2016.
- **Figure 5.4** Cumulative exposure-response curves between daily Tappmax Page 46 and cause-specific mortality and elderly mortality, over lag days 0–20, as calculated for the city of Thessaloniki for the period 2006–2016. 95% CI marked as grey areas.
- **Figure 5.5** Non-linear effects of extreme cold Tappmax on daily Page 47 cause-specific mortality and elderly mortality at lag 0–20, as calculated for the city of Thessaloniki for the period 2006–2016. Effects were defined as the risks at 5th percentile of Tappmax distribution compared with the estimated MMT.
- **Figure 5.6** Non-linear effects of extreme hot Tappmax on daily Page 48 cause-specific mortality and elderly mortality at lag 0–20, as calculated for the city of Thessaloniki for the period 2006–2016. Effects were defined as the risks at 95th percentile of Tappmax distribution compared with the estimated MMT.
- **Figure 5.7** Fractions of mortality and elderly mortality attributable to Page 51 non–optimum Tapp_{max}.
- Figure 6.1 . The study area with the locations of air quality stations (1: Page 57 Agias Sofias, 40.63° N 22.94° E; 2: AUTh, 40.63° N 22.95° E; 3: Panorama, 40.58° N 23.03° E; 4: Kalamaria, 40.57° N 22.96° E; 5: Kordelio, 40.67° N 22.89° E). Dotted lines represent municipalities.
- **Figure 6.2** Overall effect of PM10 on all-cause mortality (a) and Page 62 cardiorespiratory mortality (b); overall effect of O3 on all-cause mortality (c) and cardiorespiratory mortality (d) for Thessaloniki in the years 2006–2016.
- Figure 7.1 The study area with the locations of monitoring stations (1: Page 71 Agias Sofias, 2: AUTh, 3: Panorama, 4: Kalamaria, 5: Kordelio, 6: Makedonia airport). Dashed lines represent municipal borders.

along with sections corresponding to low, r pollution values (5th, 50th and 95th percentile
--

- **Figure 7.3** Exposure-lag-response risk surface demonstrating the nonlinear Page 76 association between Tappmax and mortality (left panel); Cumulative exposure-response curve between daily Tappmax and all-cause mortality over lag days 0-20 (right panel).
- **Figure 7.4** Combined effect of PM10 and Tappmax on all-cause mortality Page 76 stratified by levels of pollutant (defined as "low", "medium", and "high" based on 5th, 50th, and 95th percentiles of PM10 distribution).
- **Figure 7.5** Cumulative exposure-response association across lags 0-6 for Page 78 different levels of PM10 (blue denotes "low", yellow denotes "medium", and red denotes "high").
- Figure 7.6 Cumulative exposure-response associations across lags 0-6 for Page 79 Low, Medium and High PM10 levels. Dashed lines represent 95% CI.
- **Figure 7.7** Dose-response curves of cumulative RR across Tappmax range, Page 82 demonstrating the lag structure for PM10 (top left, lags 0-6), NO2 (top right, lags 0-6) and O3 (bottom left, lags 0-8).
- **Figure 7.8** Dose-response curves of cumulative RR across pollutants Page 83 concentration ranges, demonstrating the lag structure for PM10 (top left), NO2 (top right) and O3 (bottom left).
- **Figure 7.9** Estimates of heat effects as % change in cause-specific Page 84 mortality for Tappmax increases between the 75th and 99th percentile by levels of pollutants (defined as "low", "medium", and "high" based on 5th, 50th, and 95th percentiles of pollutants distributions).
- **Figure 7.10** Fractions of all-cause mortality attributable to non-optimum Page 86 Tappmax by levels of pollutants (defined as "low", "medium", and "high" based on 5th, 50th, and 95th percentiles of pollutants distributions).

Figure 7.11 Fractions of cause-specific mortality attributable to levels of Page 87 pollutants (defined as "low", "medium", and "high" based on 5th, 50th, and 95th percentiles of pollutants distributions).

Appendix B: List of Tables

Table 5.1	Model fit statistical criteria per exposure variable	Page 41
Table 5.2	Summary statistics of daily mortality (number of deaths, top) and Tappmax (°C, bottom).	Page 42
Table 5.3	Relative risks of daily cause-specific mortality and elderly mortality associated with non-optimum ambient temperatures, for changes below and above MMT and the 95th and 5th percentile.	Page 45
Table 5.4	Percentage change in mortality per 1°C change in Tappmax above and below MMT.	Page 49
Table 5.5	Cause-specific mortality and elderly mortality fraction (%) attributable to high and low levels of Tappmax.	Page 50
Table 6.1	Mean annual values of pollutants (μ g/m3) during the study period and during PM10 reduction scenario 2. Numbers in parentheses denote the percentage of annual violations of the EU daily limits	Page 59
Table 6.2	Statistics of the daily mortality (number of deaths, top) and pollution (μ g/m3, bottom).	Page 60
Table 6.3	Associations between cause-specific mortality and short- term exposure to PM10 and O3 at various time intervals (0– 1, 1–6, and 0–6 days). Results are presented as a percentage increase of risk (RR%) and as 95% confidence intervals (95% CI) per 10 µg/m3.	Page 61
Table 6.4	Attributable mortality (AM, number of deaths) and attributable mortality fraction (AF, %) for different causes of mortality.	Page 64
Table 6.5	RR (%), AM (number of deaths), and AF (%) of total mortality for different PM10 scenarios.	Page 65
Table 7.1	Description of the monitoring stations	Page 72
Table 7.2	Descriptive statistics of mortality (number of deaths, top), Tappmax (°C, middle) and pollution (µgr/m3, bottom).	Page 76